Duke Power Company Oconer Naclear Generation Department P.O. Box 1438 Seneca, SC 29679

PONILPOR

3 W. HAMPTON Vice President (803)885-3499 Office (704)373-5222 Fax



DUKE POWER

January 14, 1992

U. S. Nuclear Regulatory Comm. Office of Nuclear Materials Safety and Safeguards Div. of Fuel Cycle and Material Saf. Washington, D.C. 20555

Subject: Oconee Nuclear Station Docket Nos. 72-04, 50-269, 50-270, 50-287 Independent Spent Fuel Ftorage Installation (ISFSI) Final Safety Analysis Peport 1992 Update

Pursuant to 10CFR 72.70, please find attached 8 copies of the 1991 Update to the Oconee ISFSI FSAR. This is a complete reissue of the FSAR in the Bookmaster Format and should replace the entire contents of the existing FSAR manual which should either be discarded or clearly marked as superseded.

Revisions effective with this update are marked by a "1" in the left hand margin. The effective date June 30, 1991, is indicated at the bottom of the page.

Change: F BROWN LAT. Encl NFOG

Very truly yours,

J. W. Hampton

2500 83



U. S. Nuclear Regulatory Commission January 14, 1992 Page 2

J. W. Hampton, being duly sworn, states that he is Vice President of Duke Power Company, Oconee Nuclear Site; that he is authorized on the part of said Company to sign and file with the Nuclear Regulatory Commission this revision to the ISFSI FSAR and that all statements and matter set forth therein are true and correct to the best of his knowledge.

J. W. Hampton, Vice President

Oconee Nuclear Site

Subscribed and sworn to before me this 14th day of January, 1992.

Conice M. Dreascale Notary Public

My Commission Expires:

March 3, 1993

Duke Power Company Wachocia Center P.O. Box 1007 Charlotte: N.C. 28201-1007



DUKE POWER

January 16, 1992

Subject: Oconee Nuclear Station Docket Nos. 72-4, 50-269, 50-270, and 50-287 1991 FSAR Revision Independent Spent Fuel Storage Installation (ISFSI)

The ISFSI FSAR was converted into an electronic publishing format just prior to initiating the 1991 revision cycle. This "converted" FSAR is thus a complete reissue, containing revisions effective as of June 30, 1991. Please discard your copy of the old FSAR, or ensure that it is clearly marked as a superseded document.

12-4 Por port

The revision line indicator is a "1" in the left hand margin corresponding to the 1991 update. These indicators will remain until they are superseded by subsequent indicators. Next year's revision line indicator will be the number "2".

Our goal is to reduce the costs associated with the maintenance of numerous copies of the ISFSI FSAR by providing it in an electronic format.

R. L. Gill, Jr., Technical Syste Manager Regulatory Compliance

-Helen Frache

7201270071 11. HU Wby TORIND

strell

By: Helen Froebe Regulatory Compliance

HAF/onsisfsi

Attachment

Table of Contents

-1

Oconee ISFSI Safety Analysis Report

TABLE OF CONTENTS

2201276077 100FP.

CH	APTER 1. INTRODUCTION AND GENERAL DESCRIPTION OF STORAGE SYSTEM	1-1
1.1	INTRODUCTION	1-3
	1.1.1 FIGURES	1-4
1.2	GENERAL DESCRIPTION OF INSTALLATION	1-5
	1.2.1 GENERAL DESCRIPTION	1.5
	1.2.2 PRINCIPAL SITE CHAPACTERISTICS	1-5
	1.2.3 PRINCIPAL DESIGN CRITERIA	1-5
		1.0
	1.2.3.2 Decay Heat Dissipation	1.0
	1.2.4 OPERATING AND FUEL HANDLING SYSTEMS	1+0
	1.2.5 SAFETY FEATURES	1-0
	1.2.6 RADIOACTIVE WASTE AND AUXILIARY SYSTEMS	1-0
	1.2.7 TABLES	1.11
1.3	GENERAL SYSTEMS DESCRIPTIONS	1-11
	1.3.1 SYSTEM: DESCRIPTIONS	1.11
	1.3.1.1 DSC Design	1.11
	1.3.1.2 Horizontal Storage Module	1.15
	1.5.1.4 Transfer T	1.15
	1.3.1.4 Transfer Pack Child	1.13
	1.2.1.2 ITADNET AASS SKO 1.2.1.6 Unexpected Hodenalis Ram	1.13
	1.2.1.0 Horizonia Hydraulic Nam	1-13
	112 TARLIS	1-15
1.4	IDENTIFICATION OF AGENTS AND CONTRACTORS	1.17
1.8	MATERIAL INCORPORATED BY REFERENCE	1-19
16	REFERENCES	1-21
CB	LAPTER 2. STITE CHARACTERISTICS	2.1
2.1	GEOGRAPHY AND DEMOGRAPHY	2.3
	21.1 SITE LOCATION	2-3
	2.1.2 SITE DESCRIPTION	2-3
	2.1.2.1 Legal Responsibilities for Site	2-3
	2.1.2.2 Other Activities Within the Site Boundary	2.3
	2.1.2.3 Arrangements for Traffic Control	2-4
	2.1.3 POPULATION DISTRIBUTION AND TRENDS	2.4
	2.1.3.1 Transient Population	2.5
	2.1.4 USES OF NEARBY LAND AND WATERS	2.5
	21.5 TABLES	2.7
	2.1.6 FIGURES	2.9
2.2	NEARBY INDUSTRIAL TRANSPORTATION, AND MILLIARY FACILITIES	2-13
	2.2.1 INDU3TRIAL AND MULITAKY FACILITIES	2-13
	2.2.2 TRASSPORTATION ROUTES	2.13
	2.2.2.1 Description of Products and Materials	2-13
2.3	METPOROLOGY	2.13
	23.1 KRAIONAL CLIMATOLOGY	2.17
	232 OR AL METEOROLOGY	2-12
	23.21 IJala Sources	5 10
	2.3.2.2. TOPO202005 2.3.2. ONETTE MUTCODOLOGICAL MEASUBENIENT DROGRAM	610
	223 UNMER METEURUHANICAL MEAN KEMENT PROUBAN	6.1-



2.3.4 DIFFUSION ESTIMATES	2.17
2.3.4.1 Basis	2.17
2.3.4.2 Calculations	2.17
2.3.5 TABLES	. 2.18
2.3.6 FIGURES	2-24
2.4 HYDROLOGIC ENGINEERING	. 2.27
2.4.1 HYDROLOGIC DESCRIPTION	2.27
2.4.1.1 Site and Facilities	. 2.27
2.4.1.2 Hydrosphere	. 2.27
242 FLOODS	. 2.28
2.4.2.1 Flood History	2.28
2.4.2.2 Flood Design Consideration	. 2.28
2.4.3 PROBABLE MAXIMUM FLOOD ON STREAMS AND RIVERS	2.30
2.4.3.1 Probable Maximum Precipitation	. 2.30
2.4.3.2 Runoff and Stream Coulse Models	. 2.30
2.4.3.3 Probable Maximum Flood Flow	2.30
2.4.3.4 Coincident Wind Wave Activity	2-30
244 POTENTIAL DAM FAILURES SEISMICALLY INDUCED	2.30
24.5 FLOODING PROTECTION REQUIREMENTS	2.31
2.4.5.1 Flood Protection Measures for Oconee Station Seismic Class 1 Structures	2.31
2.4 s.2. Flood Protection Measures for ISESI Site	2.31
246 ENVIRONMENTAL ACCEPTANCE OF EFFULENTS	2.31
24.7 SUBSURFACE HYDROLOGY	2.31
2471 Groundwater L'save	2.31
2.4 " 2 Regional Groundwater Conditions	2.32
24.2.3 Groundwater Onality	2.33
2.4.7.4 Promam of Investigation	2.34
2.4.7.4 Program of investigation 2.4.7.5 Groundwater Conditions Due to Feature Reservair	2.34
SAC TADIES	2.36
SAG EIGIDES	2.38
24.3 FIGURES	2.43
2.5 GEOLOGY AND SEISMOLOGY 2.5 L BASIC GEOLOGIC AND SEISMIC INFORMATION	2.43
2.5.1 BASIC GEOLOGIC AND SEISMIC INFORMATION	2.43
2.5.1.1 Keptonal Geology	2.44
ALL DIE WORE	2.46
2.5.2 VIDRATORT OROUND NOTION	2.46
25.2.1 Cannouake rustory 25.2.2 Canalana Structures and Testania Activity	3.47
2.5.2.2 Geologic Structures and Tectorial Activity with Geologic Structures or Tectoric Province	5.40
2.5.2.5 Correlation of Earthquake Activity with Geologic Subsciences of Festival	2.49
2.5.2.4 Maximum Lannquake Potential	2.50
2.5.2.5 Scisific wave Franklinsson Characteristics of the Sile	3.50
2.5.2 6 x78XIIIIum Hypothetical Farinquake (MITH)	3.50
2.5.2 / Design Dase Languase	5.50
2.5.2.5 Design Response Spectra	3.50
2.5.3 SURFACE FACTIONS	2.50
2.5.4 SUBSERFACE MATERIALS	3.61
2.5.4.1 Exploration	3.61
2.3.4.2 Groundwater Conditions	0.61
2.5.5 ISEST FOUNDATION	0.00
2.5.6 LIQUEFACTION	5.24
2.5.7 SLOPI STABILITY	2 - 2-32
2.5.8 TABLES and a second seco	2. 2. 34
2.5.9 FIGURES	2-60



Table of Contents

Oconee ISFSI Safety Analysis Report

2.6 REFERENCES	2-87
CHAPTER 3. PRINCIPAL DESIGN CRITERIA	3-1
3.1 PURPOSE OF THE OCONEE ISFS1	3.3
3.1.1 MATERIAL TO BE STORED	3-3
3.1.1.1 Physical Characteristics	3-3
3.1.1.2 Reactivity Characteristics	3.3
3.1.1.3 Thermal Characteristics	3-3
3.1.1.4 Radiological Characteristics	3.4
3.1.2 GENERAL OPERATING FUS: HUNS	3.4
3.1.2.1 Overall Functions of the Facility	3.4
3.1.2.2 Handung and Transier Equipment	3.6
ALS TABLES AND NATURAL FOR ENVIRONMENTAL CONDITIONS AND NATURA!	
BUENOMENA	3.7
121 TOPNADO AND WIND LOADINGS	3.7
1211 Applicable Design Parameters	3.7
1.2.1.2 Determination of Lorges on Structures	3-7
1.2.1.1. Tornado, Generated Missiles	3-8
3.2.1.2 Ability of Structures to Perform	3-8
122 WATER LEVEL (FLOOD) DESIGN	3-10
3.3.3 SEISMIC DESIGN	3-10
3231 Input Criena	3-10
3.2.3.2 Seismic System Analysis	3-11
324 SNOW AND ICE LOADS	3-11
3.2.5 COMBINED LOAD CRITERIA	3-11
3.26 TABLES	. 3-12
3.3 SAFETY PROTECTION S' STEM	3-13
3.3.1 GENERAL	. 3-13
3.3.2 PROTECTION BY MULTIPLE CONFINEMENT BARRIERS AND SYSTEMS	3-13
3 ° ° 1 Confinement Barriers and Scitems	. 3-13
3	3-14
3.3.3 PROTECTION BY EQUIPMENT AND INSTRUMENTATION SELECTION	. 3-14
3.3.3.1 Equipment	. 3-14
3.3.3.2 Laster Trate on	3-15
3.3.4 NUCLE IN RY CALITY SAFETY	3-15
3.3.4.1 Course' Manual Prevention of Criticality	3-15
3.3.4.2 Error Continguacy Criteria	3-16
3.3.4.3 Verification Analysis	3-16
3.3.5 RADIOLOGICAL PROTECTION	3-17
3.3.6 FITE AND EXPLOSION PROTECTION	. 3-18
3.3.7 MATERIALS HANDLING AND STORAGE	3-18
3.3.8 TABLES	3-19
3.4 SUMMARY OF STORAGE SYSTEM DESIGN CRITERIA	3-21
3.5 REFERENCES	. 3-23
CHAPTER 4. STORAGE SYSTEM	. 4-1
4.1 LOCATION AND LAYOUT	. 4-3
4.1.1 FIGURES	4-5
4.2 STORAGE SITE	4-7
4.2.1 STR* TURES	4-7
4.2.2 STORAGE SITE LAYOUT	4-7

3

.

	4.7
4.2.3 HSM DESCRIPTION	1.6
4.2.4 INSTRUMENTATION SYSTEM DESCRIPTION	4.0
4.2.5 FIGURES	A 11
43 TRANSFER SYSTEM	2 × 1 11-11
431 FUNCTION	A 11
432 COMPONENTS	
43.2.1 Transfer Cask	4-11
43.2.2 Dry Storage Camster (DGC)	4-11
4323 Transfer Cask Skid	4-12
A 3 3 A Transfer Trailer	4-12
4.3.2.4 Trainer Trainer	4-13
4.3.2.5 Shid Fostioning System (HRS) Description	4-13
4 ALD HYDRAULE RAIL SYSTEM FETTY ASSURANCE	4-14
4.3.3 DESIC & BASES AND SALETT AND THE CO	4.14
4.3.3.1 Transfer Cass	4-14
4.3.3.7 ansier Cask Sho	4-14
4.3 2.3 L'arister Trailer	4-14
4.3.5.4 Skid Positioning Syr 5 (Sr 5)	4-16
4.3.3.5 Hydraulic Ram	4-17
4.3.3.6 Other Equipment	
4.3.3.7 Qualification of Components	4-18
4.3.3.8 Maintenance of HKS and SPS	4-19
4.3.4 TABLES	4-21
4.4 OPERATING SYSTEMS	4-21
4.4.1 LOADING AND UNLOADING SYSTEM	4-21
4.4.1.1 Preparation for Fuel Loading	4.22
4.4.1.2 Spent Fuel Selection	4-22
4.4.1.3 Spent Fuel Loading	4.23
4.4.1.4 DSC Drying Backfilling and Sealing	4.23
4.4.1.5 DSC Unloading	4.23
4.4.2 DECC STAMINATION SYSTEM	4.24
4.4.3 DSC K , AIR AND MAINTENANCE	4.04
444 TRANSFER CASK REPAIR AND MAINTENANCE	4.74
445 UTILITY SUPPLIES AND SYST MS	4.24
446 OTHER SYSTEMS	4.24
4.4.6.1 Communications and Alarm System	4-24
4462 Fire Protection System	4.24
4463 Maintenance System.	4-23
4* FIGURES	4-20
ASSIFICATION OF STRUCTURES, SYSTEMS, AND COMPONENTS	4-31
ST TRANSFER CASK	4-31
152 DRY STORAGE CANISTER	4-32
A 53 HORIZONTAL STORAGE MODU'E	
A CALEOUNDATION	
454 FOC ADATION COMPONENTS	4-32
ASA INSTRUMENTATION	4-33
4.5.0 TADIES	4-34
4.5. TABLES	4-35
4.0 DECOMMISSIONING FLORE	4-37
4./ KEFEKEAUES	
CHARTER & STORAGE SYSTEM OPERATIONS	
CHAPTER 5. STORAGE STATE A CHERTICAL	5-3
M OPERATION DESCRIPTION	5-3
MII NAKKAINE DESCRUTING	

Table of Contents

·¥

	5.1.1.1 Loading of the DSC into the HSM 5.1.1.2 Monitoring Operations	5-3
	5.1.1.3 Fuel Identification and Accountability	5.4
	51.2 ELOW SHEET	5-5
	51.3 IDENTIFICATION OF SUBJECTS FOR SAFETY ANALYSIS	5.5
	5131 Criticality Prevention	5-5
	5.1.3.2 Instrumentation	5.5
	5.1.3.3 Maintenance Techniques	
	5.1.3.4 Administrative Controls to Limit DBT Effects	
	5.1.4 FIGURES	
	5.2 CONTROL ROOM AND CONTROL AREAS	5-13
	5.3 REFERENCES	
	CHAPTER 6. WASTE MANAGEMENT	6-1
	CHAPTER 7. RADIATION PROTECTION	7-1
	7.1 ENSURING THAT OCCUPATIONAL RADIATION EXPOSURES ARE ALARA	7-3
	7.1.1 POLICY & ORGANIZATIONAL CONSIDERATIONS	7-3
	7.1.2 DESIGN CONSIDERATIONS	7-5
	7.1.3 ALARA OPERATIONAL CONSIDERATIONS	7-7
	7.2 RADIATION SOURCES	7-9
	7.2.1 CHARACTERIZATION OF SOURCES	7.9
	7.2.2 AIRBORNE RADIOACTIVE MATERIAL SOURCES	7.10
	7.2.3 TABLES	7.11
	7.3 RADIATION PROTECTION DESIGN FEATURES	7-14
	3.1 INSTALLATION DESIGN FEATURES	7.11
	22.2.1 Dediation Shielding Design Festures	7.11
	2.3.2.1 Kadiation Anclusis	7-12
	133 TABLES	. 7-13
	734 FIGURES	7-15
	7.4 ESTIMATED ON-SITE COLLECTIVE DOSE ASSESSMENT	
	7.4.1 OPERATIONAL DOSE ASSESSMENT	
	7.4.2 STORAGE TERM DOSE ASSESSMENT	7-18
	7.4.3 TABLES	7-19
	7.4.4 FIGURES	7-24
	7.5 PADIATION PROTECTION PROGRAM	7-29
	7.6 ENVIRONMENTAL MONITORING PROGRAM	7-31
	7.7 ESTIMATED OFF-SITE COLLECTIVE DOSES	7+33
	7.8 REFERENCES	1.30
	CHAPTER 8. ACCIDENT ANALYSES	8-1
	8.1 OFF-NORMAL OPERATIONS	8-3
	8.1.1 JAMMED DSC DURING LOADING OR UNLOADING	8-3
	8.1.1.1 Postulated Cause of Jammed DSC	
	8.1.1.2 Detection of Jammed DSC	6.3
	8.1.1.3 Analysis of Effects and Consequences	6.3
	8.1.2 PADIOLOGICAL IMPACT OF OFF NORMAL OPERATIONS	8.4
	8.2 ACCIDENTS	8-5
	8.21 LOSS OF AIR OUTLET SHIFLDING	8-5

Table of Contents

8.2.1.1 Cause of Accident	8-5
	10.00
8.2.1.2 Accident Analysis	8-5
8.2.1.3 Accident Dose Consequences	8-0
8.2.2 TORNADO TORNADO MISSILE	8-0
8.2.2.1 Cause of Accident	8-0
8.2.2.2 Accident Analysis	. 8-0
8.2.2.3 Accident Dose Consequences	. B-D
8.2.3 EARTHQUAKE	0.7
8.2.3.1 Cause of Accident	6.7
8.2.3.2 Accident Analysis	- · · · · · · ·
8.2.3.3 Accident Dose Consequences	8-7
8.2.4 CASK DROP	8-7
8.2.4.1 Cause of Accident	817
8.2.4.2 Accident Analysis	. 8-1
8.2.4.3 Accident Dose Consequences	
8.2.5 TRANSFER CASK LOSS OF NEUTRON SHIELD	- B-/
8.2.5.1 Postulated Cause of Solid Shield Loss	5-5
8.2.5.2 Detection of Shield Material Loss	8-8
8.2.5.3 Analysis c? Effects and Consequences	8-8
8.2.6 LIGHTNING	8-8
8.2.6.1 Cause of Accident	8-8
8.2.6.2 Accident Analysis	8-8
8.2.6.3 Accident Dose Consequences	8-9
8.2.7 BLOCKAGE OF AIR INLETS AND OUTLETS	
8. I Cause of Accident	8-9
8.2.7.2 Accident Analysis	8-10
8.2.7.3 Accident Dose Consequences	. 8-10
8.2.8 DRY STORAGE CANISTER LEAKAGE	· · · · · · · · · · · · · · · · · · ·
8.2.8.1 Cause of Accident	8-10
8.2.8.2 Accident Analysis	8-10
8.2.8.3 Accident Dose Consequences	8-11
8.2.9 ACCIDENTAL PRESSURIZATION OF DSC	- B-11
3.2.9.1 Cause of Accident	
8.2.9.2 Accident Analysis	8-11
8.2.9.3 Accident Dose Calculations	· · 8-11
8.2.10 LOAD COMBINATIONS	8-11
8.2.10.1 Cause of Accident	0-14
8.2.10.2 Accident Analysis	0.15
8.2.10.3 Accident Dose Consequences	6.10
8.2.11 FLOODING	0-14
8.2.12 EXPLOSIONS	6.17
8.2.12.1 Cause of Accident	8.13
8.2.12.2 Accident Analysis	9.10
8.2.12.3 Accident Dose Consequences	8.12
8.2.13 TABLES	8.15
A STIL CHARACTERISTICS AFFECTING SALETT ANALTSIS	8.17
A REFERENCES	0.11
THERE & CONDUCT OF OPEN TIONS	0,1
HAPTER 7. CONDUCT OF OPERATIONS	0.1
I UKOANIZATIUNAL STRUCTURE	0.3
9.1.1 CORPORATE ORGANIZATION	0.3
9.1.1.1 Corporate Functions, Responsibilities and Authornies	2.2.2



Table of Contents

9.1.1.2 Applicant's In-House Organization	8-3
9.1.1.3 Interrelationship with Contractors and Suppliers	9.3
9114 Applicant's Technical Staff	9.3
912 OPERATING ORGANIZATION, MANAGEMENT, AND ADMINISTRATIVE	
CONTROL SYSTEM	9-4
9121 Onsite Organization	9-4
9.1.2.2 Personnel Functions Responsibilities and Authorities	9-4
0.1.3. PERSONNEL OF ALTERCATION REOUREMENTS	9-4
9131 Minimum Dualification Requirements	9-4
0.1.2.2 Auditiestions of Personnel	9-4
and LIAISON WITH OTHER ORGANIZATIONS	9-1
DIS LIGURES	9.5
0.2 DELODERATIONAL TESTING AND OPERATION	9-7
6.5.1 ADMINISTRATICE PROCEDURES FOR CONDUCTING TEST PROGRAM	9.7
655 TEST DECORAM DESCRIPTION	9.7
0.221 Operations	9.7
0.1.1 TEST DISCUSSION	9.8
A TRAINING PROGRAM	9.9
8.2.1. TRAINING FOR OPERATIONS PERSONNEL	9.9
0.1.5. TRAINING FOR MAINTENANCE PERSONNEL	9.9
9.5.2 TRAINING FOR MAINTENANCE TERSONNEL	9-9
1 933 TRAINING FOR SECURITY PERSONNEL	9.9
A VODMAL OPEDATIONS	9-11
AND BROCEDIPES	9-11
9.4.1 PROCEDE	9-11
6 CENEDGENCY PLANNING	9-13
A DUNCICAL SECIENT PLAN	9-15
0.2 DIELDINGTS	9-17
3.1 KLITEKLALIA	
CHARTER 10. OPERATING CONTROLS AND LIMITS	10-1
10.1 PROPOSED OPERATING CONTROLS AND LIMITS	10-3
10.1 DEVELOPMENT OF OPERATING CONTROLS AND LIMITS	10-5
10.2.1. FUNCTION &: AND OPERATING LIMITS MONITORING INSTRUMENTS	
AND LIMITING CONTROL SETTINGS	10-5
10.3.2 LIMITING CONDITIONS FOR OPERATION	10-5
10.2.2.1 Engineent	10-5
20.2.2.2. Technical Conditions And Characteristics	10-5
10.2.1 SURVEN LANCE REOLTRI MENTE	10-6
16" A DEC 3X FEATLQFS	10-6
NOS ADMINISTRATIVE CONTROLS	10-6
10.3.5.1. Qualification of Spent Fuel	10-6
10.2.5.2 Spent Fuel Identification	10-7
10.1 OPERATIONAL CONTROL AND LIMIT SPECIFICATION	10-9
10.3 UTLEATIONAL CONTROL AND LIAIT STECH RATIO	10-10
10.4 REFERENCES	10-11
IN A INDERICATED	
CHAPTER 11 OF ALITY ASSURANCE	. 11-1
CHATTER II. QUALITI ASSOCIATED	



-vii

Table of Contents

Oconee ISFSI Safety Analysis Report

LIST OF TABLES

đ

.8

...

4

1-1.	Design Parameters for the Oconee ISFS1	1.7
1-2.	Summary of ISFSI Fuel Handling Operations	1-9
1-3.	Primary Design Parameters for the ISFSI Transport Systems	1-10
1-4	Major Systems. Subsystems and Components of the Oconee ISFS1	1-15
2-1.	1970 Population Distribution 0-10 Miles	2-7
2.2	Population Projection for Oconee Nuclear Station for the Year 2020	2-8
2.3	Joint Frequencies of Wind Direction and Speed by Stability Class	2-18
2.4	Soil Permeability Test Results	2-37
2.5	Significant Earthquakes in the Southeast United States (Intensity V or Greater)	2-55
3.1.	Physical Characteristics of PWR Fuel Assemblies Based on Nominal Design	3-6
3-2.	Transfer Cask Stress Analysis for Tornado Effects	3-12
3-3	Oconee ISFSI Major Components and Functions	3-19
3-4	Oconee ISFSI Radioactive Material Confinement Barriers	3-20
3.5.	Oconce ISFSI Major Components and Design Requirements	3-20
4.1	ONS ISFSI Project Transfer Trailer Design Parameters	4-19
4-2.	Oconee ISFSI Major Components and Classification	4-34
7.1.	Gamma Energy Spectrum	7~10
7-2.	Shielding Analysis Results	7-14
7.3.	Summary of Estimated On-Site Doses Resulting from ISESI Operations(1)	7+19
7-4.	Dose Estimate for Construction of Additional Horizontal Storage Modules Based on	
	Labor Estimates for 2 X 10 Array	7-22
7-5.	Neutron and Gamma Energy Spectra	7-23
8-1	Comparison of Total Dose Rates for HSM With and Without Air Outlet Shielding Blocks	8-1.4
8.3	Cask Drop Target Parameters	8-13

List of Tables

Oconee ISFSI Safety Analysis Report

List of Figures

Oconee ISFSI Safety Analysis Report

LIST OF FIGURES

1

Location of ISFSI								1-4
General Location								2-9
Site Plan								2-10
ISFSI Layout								2-11
Topography Within 5 Miles								2-24
Relative Positions of Meteorological Instruments								2-25
Relative Elevations of Meteorological Instruments								2-26
Areal Groundwater Survey								2-38
Groundwater Survey at Station Site								2-39
Well Permeameter Test Apparatus								. 2-40
Formulae For Determining Permeability								. 2-41
General Site Area								. 2-60
Site Boring Plan								2-61
Core Boring Record								2-62
Core Boring Record								2-64
Core Boring Record								2-66
Core Boring Record								2-67
Core Boring Record								. 2-68
Core Boring Record								. 2-70
Core Boring Record								. 2-72
Core Boring Record								2.74
Core Boring Record								, 2-76
Core Boring Record								2-78
Core Boring Record								2-79
Core Boring Record								2-81
Core Boring Record								. 2-83
Core Boring Record								. 2-85
ISFSI Foundation Profile								2-86
Site Layout and Route								4-5
Site Plan								4-9
Transier Cask Lift Beam								. 4-26
Crane Hook Lift Adaptor								4-27
Spent Fuel Pool Area								. 4-28
Spent Fuel Pool Area								4-29
Spent Fuel Pool Area								. 4-39
NUHOMS System Loading Operations Flowchart								5-7
NUHOMS System Loading Operations Flowchart								. 5-8
NUHOMS System Loading Operations Flowchart								. 5-9
NUHOMS System Loading Operations Flowchart								. 5-10
NUHOMS System Loading Operations Flowchart								. 5-11
NUHOMS System Loading Operations Flowchart								. 5-12
Location of Dose Plates								7-15
Dose Rate Versus Distance From Surface of HSM							1	. 7-24
Dose From Filled HSM Array								. 7-25
Dose From Filled HSM Array								7-26
Radiation Zone Map of Module Surface Dose Rate-								. 7-27
Deleted								. 9-5
Fuel Assembly Acceptance Criteria Cooling Period >	10 Yes	trs :						10-10
	General Location Site Plan ISFSI Layout Topography Within 5 Miles Relative Positions of Meteorological Instruments Relative Elevations of Meteorological Instruments Areal Groundwater Survey Groundwater Survey at Station Site Well Permeameter Test Apparatus Formulae For Determining Permeability General Site Area Site Boring Record Core Boring Record C	General Location Site Plan ISFSI Layout Topography Within 5 Miles Relative Positions of Meteorological Instruments Areal Groundwater Survey Groundwater Survey at Station Site Well Permeameter Test Apparatus Formulae For Determining Permeability General Site Area Site Boring Record Core Boring Record Site Layout and Route Site Layout and Route Site Plan Transier Cask Lift Beam Crane Hook Lift Adaptor Spent Fuel Pool Area Spent Fuel P	Location of ISESI General Location Site Plan ISFSI Layout Topography Within 5 Miles Relative Positions of Meteorological Instruments Areal Groundwater Survey Groundwater Survey at Station Site Well Permeameter Test Apparatus Formulae For Determining Permeability General Site Area Site Boring Record Core Boring Record Site Ian Transiet Cask Lin Beam Crane Hook Lint Adaptor Spent Fuel Pool Area Spent Fuel Pool Area Spent Fuel Pool Area Spent Fuel Pool Area NUHOMS System Loading Operations Flowchart NUHOMS System Loading Operations Flowchart NUH	Location of ISESI General Location Site Plan ISESI Layou Topography Within 5 Miles Relative Positions of Meteorological Instruments Relative Elevations of Meteorological Instruments Areal Groundwater Survey Groundwater Survey at Station Site Well Permeameter Test Apparatus Formulae Fot Determining Permeability General Site Area Site Boring Plan Core Boring Record Core Boring Record Site Layout and Route Site Plan Transet Cask Lin Beam Crane Hook Lift Adaptor Spent Fuel Pool Area Spent Fuel Poo	Location of ISFSI General Location Site Plan ISFSI Layout Topography Within 5 Miles Relative Positions of Meteorological Instruments Relative Positions of Meteorological Instruments Areal Groundwater Survey Groundwater Survey at Station Site Well Permeameter Test Apparatus Formulae For Determining Permeability General Site Area Site Boring Record Core Boring Record SIFSI Foundation Profile Site Layout and Route Site Plan Transier Cask Lift Beam Crane Hook Lift Adaptor Spent Fuel Pool Area Spent Fuel Pool Area System Loading Operations Flowchart NUHOMS System Loading	Location of ISTS1 General Location Site Plan ISTS1 Layout Topography Within 5 Miles Relative Positions of Meteorological Instruments Relative Positions of Meteorological Instruments Areal Groundwater Survey Groundwater Survey at Station Site Well Permeameter Test Apparatus Formulae Foot Determining Permeability General Site Area Site Boring Plan Core Boring Record Core	Location of ISEST General Location Site Plan ISESI Layout Topography Within 5 Miles Relative Positions of Meteorological Instruments Relative Elevations of Meteorological Instruments Areal Groundwater Survey at Station Site Well Permeameter Test Apparatus Formulae For Determining Permeability General Site Area Site Boring Plan Core Boring Record Core Boring Record Site Layout and Route Site Layout Coding Operations Flowchart NUHOMS System Loading Oper	Location of ISEST General Location Site Plan ISESI Layou Topography Within 5 Miles Relative Positions of Meteorological Instruments Relative Elevations of Meteorological Instruments Areal Groundwater Survey at Station Site Well Permeameter Test Apparatus Formulae For Determining Permeability General Site Area Site Boring Plan Core Boring Record Core B



Т

List of Figures

Oconee ISFSI Safety Analysis Report

List of Abbreviations

Oconee ISFSI Safety Analysis Report

LIST OF ABBREVIATIONS

ACI	AMERICAN CONCRETE INSTITUTE
APM	ADMINISTRATIVE POLICY MANUAL
AFR	AWAY-FROM-REACTOR
AISC	AMERICAN INSTITUTE OF STEEL CONSTRUCTION
ALARA	AS LOW AS REASONABLY ACHIEVABLE
ANSI	AMERICAN NATIONAL STANDARDS INSTITUTE
AWS	AMERICAN WELDING STANDARDS
CFR	CODE OF FEDERAL REGULATIONS
DBT	DESIGN BASIS TORNADO
DOF	DEPARTMENT OF ENERGY
DPC	DUKE POWEP COMPANY
DSC*	DRY STORAGE CANISTER
EPR1	FLECTRIC POWER RESEARCH INSTITUTE
EPZ	EMERGENCY PLANNING ZONE
EST	ENGINEERED SAFETY FEATURE
ETQS	EMPLOYEE TRAINING AND QUALIFICATION SYSTEM
FEMA	FEDERAL EMERGENCY MANAGEMENT ADMINISTRATION
FSAR	FINAL SAFFTY ANALYSIS REPORT
HSM	HORIZONTAL STORAGE MODULI
HRS	HYDRAULIC RAM SYSTEM
IFA	IRRADIATED FUEL ASSEMBLY
ISFSI	INDEPENDENT SPENT FUEL STORAGE INSTALLATION
I W M	LIQUID WASTE MANAGEMENT
MHE	MAXIMUM HYPOTHETICAL FARTHQUAKE
MRS	MONITORED RETRIEVABLE STORAGE
NDE	NONDESTRUCTIVE EXAMINATION
NEPA	NATIONAL ENVIRONMENTAL POLICY ACT
NRC	NUCLEAR REGULATORY COMMISSION
NUHOMS-24P	NUTECH ENGINEERS, INC. HORIZONTAL MODULAR STORAGE
NUREG	NUCLEAR REGULATORY GUIDE
POR	PRUDENT OPERATING RESERVE
PWR	PRESSURIZED WATER REACTOR
ONS	OCONEE NUCLEAR STATION
SPS	SKID POSITIONING SYSTEM
VA	VENTILATION AIR SYSTEM
VR	STATION VOLUME REDUCTION SYSTEM
NWPA	WASTE POLICY ACT OF 1982, AS AMENDED

* The term Dry Storage Canister (DSC) in this report refers to the same item termed dry shielded canister in the Nutech Topical Report referenced in this SAR.



List of Abbreviations

Oconee ISFSI Safety Analysis Report

CHAPTER 1. INTRODUCTION AND GENERAL DESCRIPTION OF STORAGE SYSTEM



à

1.1 INTRODUCTION

Duke Power Co began commercial operation of the Oconee Nuclear Station. Units 1, 2, and 3 on July 15, 1973, September 9, 1974 and December 16, 1974 respectively. Since then these three 2568 MWt units have generated millions of KWH in a safe and reliable manner. In so doing, these units have discharged a total of approximately 2100 spent fuel assemblies. These assemblies are currently being stored in two onsite pools and in the McGuire Nuclear Station spent fuel pools. The need to provide additional onsite storage facilities to permit continued operation is discussed in 9, 10, and 11 of the Environmental Report (Reference 1 on page 1-21) submitted as part of the Oconee Independent Spent Fuel Storage Installation (ISFSI) heense application.

To support this need and provide storage until the Department of Energy (DOE) begins to accept title to spent fuel under the requirements of the Nuclear Waste Policy Act of 1982, as amended in 1987, Duke Power is requesting permission to build and operate an ISFSI in compliance with 10CFR 72. Duke Power has chosen the NUHOMS-24P dry storage system designed by Nutech Engineers, Inc. to be used on the Oconee site. The NUHOMS-24P system is more fully described in Revision 1A of the Topical Report for the NUHOMS-24P system submitted by Nutech Engineers, Inc. in July 1988 and accepted by the NRC on April 21, 1989. The location of the ISFSI on the Oconee site is shown on Figure 1-1 on page 1-4.

The NUHOMS-24P system provides long-term interim storage for irradiated fuel assemblies. The fuel assemblies are confined in a helium atmosphere by a stainless steel carister. The canister is protected and shielded by a massive concrete module. Decay heat is removed by thermal radiation, conduction and convection from the canister to an air plenum inside the concrete module. Air flows through this internal plenum by natural draft convection.

The canister containing twenty-four irradiated fuel assemblies is transferred from the spent fuel pool to the concrete module in a transfer cask. The cask is precisely aligned and the canister is inserted into the module by means of a hydraulic ram.

The NUHOMS-24P system is a totally passive installation that is designed to provide shielding and safe confinement of uradiated fuel. The dry storage canister and horizontal storage module have been designed to withstand certain accidents as described in this SAR.

The fuel assemblies to be stored in the ISFSI are currently located in the Occnee spent fuel pools and were irradiated in the Occnee reactors. Twenty-four fuel assemblies are stored in each dry storage canister, and one dry storage canister is stored in each concrete module. The license application requests a license to construct and operate a total of eighty-eight modules (2112 assemblies). These modules will be built incrementally, as needed to match t¹ e requirements for additional storage. Operation of the facility will continue past the first year for up to 20 years under the initial license and continue under license extension as necessary until the fuel can be shipped to a permanent repository. As defined in Table 1.2-2 of Reference 2 on page 1-21 the minimum service life of the facility is 50 years. During this service life, while any given HSM could be unloaded and later reloaded with a new DSC, reloading a given DSC following removal of the original fuel assemblies is not anticipated due to the potential destructive nature of the top end shield plug removal. Eventual reuse of the HSMs will depend upon the schedule and restrictions for spent fuel delivenes to DOE under the NWPA.

1.1.1 FIGURES



Figure 1-1. Location of ISFSI



1.2 GENERAL DESCRIPTION OF INSTALLATION

1.2.1 GENERAL DESCRIPTION

The ISFSI provides for the horizontal, dry storage of irradiated fuel assemblies (IFAs) in a concrete module. The principal components are a concrete horizontal storage module (HSM) and a stainless steel dry storage canister (DSC) with an internal basket which holds the IFAs. Each HSM contains one DSC and each DSC contains twenty-four fuel assemblies.

Despite DOE's obligations under the original Nuclear Waste Policy Act of 1982, Duke Power's current best estimate for receiving spent fuel storage relief under the Nuclear Waste Policy Amendments Act of 1987 is the year 2010 at the earliest. Oconee's prudent storage requirements through 2003 will necessitate the construction of 72 HSMs. This application, however, is for construction and use of up to eighty-eight HSMs. This additional margin of 16 HSMs will provide the minimum storage capacity needed to carry the Oconee Nuclear Station to its end of operating life if necessary.

The initial phase of construction including twenty modules was completed in May 1990. Additional modules will be added as required on separate foundations without impact to the preceeding or succeeding modules. Analyses for structural and foundation requirements provide for constructing modules in any array size between a single HSM and a 10x2 array (2 rows of 10 modules back to back).

In addition to these primary components, the Oconee ISFSI also requires transfer equipment to move the DSCs from the spent fuel pool (where they are loaded with the IFAs) to the HSMs where they are stored. This transfer system consists of a transfer cask, a hydraulic ram, a truck, a trailer and a cask skid. This transfer system will interface with the existing Oconee spent fuel pool, the cask crane, the site layout (i.e., roads and topography) and other procedural requirements.

1.2.2 PRINCIPAL SITE CHARACTERISTICS

The ISFSI is located on the Oconee Nuclear Station site near Seneca, South Carolina, Duke Power Company owns and operates three 2568 MW1 nuclear generating units on the Oconee site. The ISFSI is located outside the protected area be: within the owner controlled area approximately 100 ft, west of the Station's condensor cooling water intake structure (Figure 1-1 on page 1-4).

1.2.3 PRINCIPAL DESIGN CRITERIA

The principal design criteria and parameters for the Oconee ISFSI are shown in Table 1-1 on page 1-7. The radiation sources are for the reference fuel assembly. For the majority of the fuel to be stored, the radiation sources will be less than or equal to the sources described in the NUHOMS-24P Topical Report (Reference 2 on page 1-21). For radiation sources larger than the sources described in Reference 2 on page 1-21 restrictive measures will be used to ensure surface dose rates that are ALARA and below design basis limits.

1.2.3.1 Structural Features

The HSM is a low profile reinforced concrete structure designed to withstand normal operating loads, the abnormal loads created by seismic activity, tornados and other natural events and the postulated accidental loads which may occur during operation.

1.2 General Description of Installation

The structural features of the DSC are defined, to a large extent, by the cask drop accident. The DSC body, the double seal welds on each end, and the DSC internals are designed to provide for fuel retrieval after a postulated maximum credible drop.

1.2.3.2 Decay Heat Dissipation

The decay heat of the IFAs is removed from the DSC by natural uraft convection. Air enters the lower part of the HSM, rises around the DSC and exits through the top shielding slab. The flow cross-sectional area is designed to provide adequate air flow from the draft height of the HSM and the inlet and outlet air temperature differences for the hottest day conditions (i.e., 46.7°C or 116°F).

1.2.4 OPERATING AND FUEL HANDLING SYSTEMS

The major operating systems of the ISFSI are those required for fuel handling and transport of the fuel from the spent fuel pool to the ISFSI. General operations are outlined in Table 1-2 on page 1-9 and the primary design parameters of the required systems are listed in Table 1-3 on page 1-10. The majority of the fuel handling operations involving the transfer cask (i.e., fuel loading, drying, trailer loading, etc.) utilize standard techniques at Oconee for spent fuel shipment. The remaining operations (seal welding, transfer cask-HSM alignment, and DSC transfer) are unique to the ISFSI.

1.2.5 SAFETY FEATURES

The principal safety feature of the ISFSI is the containment provided by the DSC and the concrete shielding of the HSM. In addition to its structural and missile protection functions, this shielding reduces the gamma and neutron flux emanating from the IFAs inside a DSC so that the average outside surface dose rate on the HSM is less than 20 mr hr. Additional ISFSI features include:

- Filling the annulus between the DSC and transfer cask with demeneralized water and sealing it prior to lowering them into the spent fuel pool - Prevents contamination of the DSC exterior by pool water.
- Internal shield blocks inside the HSM which comprise the shielded ventilation plenum Reduces scatter dose out of the air inlet.
- 3. External shield blocks on the HSM roof Reduces scatter dose out of the air outlet.
- 4. Shield plugs on the DSC Reduces dose during DSC drying, helium filling and seal welding.
- 5. Double seal welds on each end of the DSC Prevents leakage of radioactive gases or particulates if the fuel rods should fail.

1.2.6 RADIOACTIVE WASTE AND AUXILIARY SYSTEMS

Because of the passive nature of the ISFSI, there are no radioactive waste or auxiliary systems required during normal storage operations. There are, however, some waste and auxiliary systems required during DSC loading, drying and transfer into the module. The Oconee waste systems handle the fuel pool water and air which are vented from the DSC during drying. Auxiliary handling systems (such as hydraulic pressure control, alignment, crane, etc.) are also required during the loading and transfer operation.





1.2 General Description of Installation

1.2.7 TABLES

- Accessible with Fuel Mast

Table 1-1 (Page 1 of 2). Design Parameters for the Oconee	ISESI
GENERAL DESIGN REQUIREMENTS	
· Maximum weight on crane hook	100 tons
Capacity (Casks Canister)	24 PWR Assys
- Maximum assembly weight	1682 lbs.
- Reference Fue! Assembly parameters:	
a) Nominal burnup	40,000 MWD MTU
b) Initial Enrichment (Maximum)	4 0° e
c) Maximum initial Uranium Content	472 kg assembly
d) Cooling Time	10 years nominal
e) Fuel Rod Array	15 x 15
• Fuel Cell Envelc pe (Minimum)	8.75 8.85 in
Effective multiplication factor	$K_{eff} < 0.95$
- Internal DSC atmosphere	Inert Gas (Helium)
Ambient temperature	-30°F to 116°F
Solar heat load (Maximum)	127 BTU hr-ft²
 Average doses at HSM surface during storage 	20 mr hr combined gamma and neutron
 Maximum Axial Midplane Dose at Transfer Cask Surface during Transport 	200 mr/hr combined gamma and neutron
- Maximum Loading Height (Fuel Pool)	15' 6" above pool floor
- Storage orientation	Horizontal
· Normal Operating Equilibrium Clad Temperature	340° C
 Assume Credit for Burnup for Criticality Computations² 	Based on 1.45% Initial Enrichment equivalent.



1.2 General Description of Installation

Table 1-1 (Page 2 of 2). Design Parameters for the Oconec	ISFSI	
Maximum Assembly Length (Includes Radiation Growth and Control Components)	173 in	
- Active Fuel Length	144 in	

Note:

1

1

f

1

- 1 Licensing basis design calculations assume a homogeneous source over the active fuel region (See Section 7.2.1, "Characterization of Sources" on page 7-9). Elevated dose rates in excess of 200 mrem hr over limited areas of the transfer cask surface may be observed. In particular, elevated gamma dose rates in excess of 200 mrem hr, centered on fuel assembly end fittings, can be anticipated based on initial DSC loading dose rate survey data. Supplementary shielding calculations performed subsequent to ISFSI operation demonstrate dose rates as high as 565 mrem hr centered on fuel or near assembly end fittings can be anticipated
 - 2 Primary licensing basis criticality control design feature is credit for 1810 ppm soluble boron in DSC cavity during wet loading operations. Fuel assembly initial enrichment burnup qualification procedures provide additional criticality safety margin.

Table 1-2. Summary of ISFSI Fuel Handling Operations

- 1. Clean the DSC, if necessary, and Load it into the Transfer Cask
- 2. Fill the DSC with borated water and Transfer C. sk annulus with demineralized water
- 3. Install the Inflatable Annulus Seal to seal the Cask DSC annulus
- 4. Lift the Transfer Cask Containing the DSC into the Spent Fuel Pool
- 5. Load the Fuel into the DSC
- 6. Place the Top Lead Shield Plug on the DSC
- Lift t. Transfer Cask Containing the Filled DSC out of the Spent Fuel Pool and Place it in the Decon Pit.
- 8. Remove the annulus seal.
- Lower the water level in the DSC transfer cask annulus to approximately 5 to 10 inches below the top of the DSC shell.
- Lower the water level in the DSC to approximately 4 inches below bottom surface of the top shield plug.
- 11 Seal Weld the Top End Shield Plug onto the "SC Body and perform NDE
- 12. Evacuate and Dry the DSC
- 13. Backfill the DSC with Helium
- 14. Seal Weld Covers for the Drain and Vent Line of the DSC and perform NDE
- 15. Place and Seal Weld the Top Cover Plate over the Shield Plug and perform NDE
- 16. Install the Transfer Cask Lid and Bolt in Place
- 17. Decontaminate the Transfer Cask Surface
- 18. Drain the water from the Cask DSC Annulus
- 19. Lift the Transfer Cask onto the Transfer Trailer and Lower it into the Horizontal Position
- 20. Tow the Transfer Trailer to the HSM
- 21. Remove the HSM Front Access Door
- 22. Align the Transfer Cask and the HSM
- 23. Remove the Transfer Cask Lid and Bottom Access Plate
- 24. Push the DSC into the HSM Using the Hydraulic Ram System
- 25. Retract Hydraulic Ram Arm and reposition transfer cask
- 26. Replace the HSM Front Access Door and Tack Weld in Place



1.9

System	Parameters	Value		
Transfer Cask	Nominal Cavity Diameter Nominal Cavity Length Payload Capacity (Maximum) Reference Heat Rating Shielding (Surface Dose) at Axial Midplane	68 in 188 in 90,000 lbs. 15.8 kw (.66 assembly) 200 mr hr average		
Transfer Cask Movement	Liftable by Crane Rotatable by Crane from Vertical to Horizontal	200,000 lbs. maximum Has rotation trunnions		
Transfer Cask Lid	Removable in Horizontal Position	5,400 lb.		
Trailer and Skid	Truck Transportable Transfer Cask Lid Must Protrude Past End of Trailer and Skid	15.25 cm (6 in.)		
	Capacity (Transfer Trailer) (Transfer Trailer Skid)	109,000kg (120 tans) 100,000kg (110 tans)		



1.2 General Description of Installation

1.3 GENERAL SYSTEMS DESCRIPTIONS

The major systems, subsystems, and components of the Oconee ISFSI are listed in Table 1-4 on page 1-15. The following subsections briefly describe the principal systems and components and their operation.

1.3.1 SYSTEMS DESCRIPTIONS

1.3.1.1 DSC Design

The NUHOMS-24P DSC is shown in Figure 1.3-1 of the Topical Report for the NUHOMS-24P System (Reference 2 on page 1-21). The DSC is sized to hold twenty-four irradiated pressurized water reactor (PWR) fuel assemblies. The main component of construction is a stainless steel cylinder with a nominal 67 inch outside diameter. The nominal overall length is 187 inches, excluding grapple ring.

The components of the internal basket of the DSC are defined in Teble 1-4 on page 1-15 and are also shown on Figure 1.3.1 of Reference 2 on page 1-21. The basket is comprised of twenty-four square cells. The structural component of the cells is type 304 stainless steel.

Structural support is provided by circular stainless-steel spacer disks. The basket is designed so that there is one disk under each spacer prid of the fuel assembly. Longitudinal support is provided by the four support rods which run the length of the DSC.

The DSC is equipped with two shielded end plugs so that when the canister is inside the transfer cask or the horizontal storage module, the radiation dose at the ends is limited and they are accessible for handling. The end shield plugs are constructed of lead surrounded by a steel body.

The DSC has redundant seal welds at the top and bottom. The bottom cover plates are welded to the DSC body during fabrication and the top cover plates after fuel loading. Also, all connections (drain and vent ports) will be redundantly sealed. This assures that no single failure of the DSC end plates will breach the DSC. Furthermore, there are no credible accidents which would breach the main body of the DSC.

Criticality safety during wet loading operations is assured by 1) the design of the basket structure which maintains a minimum separation between fuel assemblies. 2) technical specification procedures which assure a minimum boron concentration in excess of 1810 ppm is maintained within the DSC storage cavity during wet loading and unloading operations, and 3) procedures which limit the reactivity of fuel assemblies loaded into the DSC to an established maximum through vet fication of initial enrichment and exposure history.

1.3.1.2 Horizontal Storage Module

An isometric view of a unit of four HSMs is shown in Figure 1.3-1A of Reference 2 on page 1-21. The HSM is fabricated from reinforced concrete and structural steel which will be constructed in place at the storage location. The thick concrete top, front, and sides of the HSM provide adequate neutron and gamioa shielding to achieve an average 20 mr hr surface dose. Nominal closure door surface doses are less than 100 mr hr. The transfer cask surface has an average dose rate of less than 200 mr hr for the locations where workers must perform loading and unloading operations.



Thick shield walls (3.0 ft, thick) are provided on the outside walls of the modules at the end of the unit to provide shielding on the sides. Sufficient (2.0 ft, thick) shielding between modules (to prevent scatter in adjacent modules during loading and retrieval) is provided by the interior module walls.

The HSM provides fuel cooling by a combination of radiation, conduction and convection. The air enters at the bottom of the HSM and passes around the DSC and exits through the flow channels in the top shield slab. Heat is conducted out of the DSC into the natural convection air flow. Heat is also radiated from the DSC to the HSM walls where the natural convection air flow removes the heat. Figure 1.3-2 of Reference 2 on page 1-21 shows the flow path and typical conditions. The passive cooling system of the HSM was designed to assure that peak cladding temperatures are less than 340°C (644°F) during long term storage for average normal ambient temperatures of 70°F. The fuel can withstand short term temperatures of up to 570°C (1.058°F) during operational and accidental transients with no anticipated adverse effects. However, calculations show that temperatures remain well below 570°C at any time during normal operation or any postulated accident.

The HSMs are independent, passive systems for the dry storage of irradiated fuel assemblies. Therefore, the HSMs are designed to ensure that normal operation and credible site hazards do not impair their function. To this end, the HSMs are designed to the following loads:

- 1. Winds and Tornado (includes missile) Oconee FSAR, Chapter 3. "Design of Structures, Components, Equipment, and Systems" on page 3-1
- Seismic Oconee FSAR. Chapter 3, "Design of Structures, Components, Equipment, and Systems" on page 3-1
- 3. Flood Oconee FSAR, Chapter 2, "Site Characteristics" on page 2-1
- 4. Snow and Ice ANSI A58.1-1982.
- 5. Combined Load (dead weight, li loads, temperature) ACI 349-85.

The HSMs are placed in service on a load bearing foundation. Earth work is required to prepare the storage site for a level foundation and access area.

1.3.1.3 Transfer Cask

The transfer cask used with the ISFSI provides radiological shielding during the DSC drying operation and during the transfer to the HSM. Both neutron (Bisco NS-3, a cementatious material) and gamma (lead) shielding are incorporated into the cask design. For the Oconee ISFSI, the transfer cask has a nominal 188 inch long internal cavity with a nominal 68 inch internal diameter. Figure 1.3-2A of Reference 2 on page 1-21 shows the major components of the transfer cask.

1.3.1.4 Transfer Trailer

The transfer trailer has a capacity of 120 tons. The transfer trailer carries the transfer cask skid and the loaded transfer cask. The transfer trailer is designed to ride as low to the ground as possible to minimize the HSM height. Four hydraulic jacks are incorporated into the transfer trailer design to provide vertical movement for alignment of the cask and HSM. The transfer trailer is pulled by a conventional tractor. Figure 1.3-3 of Reference 2 on page 1-21 shows a typical transfer trailer arrangement. Also, as discussed in Section 8.2.5 of Reference 2 on page 1-21, the design basis drop height for the NUHOMS-24P Transfer cask is 80 inches. This analysis bounds the Oconee transport conditions. The nominal travel height of the transfer trailer deck is 41 inches which corresponds to a cask drop height of 59 inches. During transit from the fuel building to the HSM site, the trailer deck will be automatically leveled by the trailer's hydraulic suspension units. The maximum design travel for these units can raise the trailer deck height to 52 inches, which corresponds to a drop height of 70 inches. Mechanical stops attached to each



suspension unit cylinder ensure that the cask cannot be lifted to a height greater than 70 inches above the ground

1.3.1.5 Transfer Cask Skid

The transfer cask skid is similar in design and operation to existing transport skids. The major differences are:

- 1. No equipment interferes with access to the top of the transfer cask when it, the borizontal position.
- The skid is mounted on a smooth bearing surface and hydraulic positioners provide horizontal alignment with the HSM. A restraining bolt system is provided to prevent movement during trailer towing.
- 3. The entire skid is mounted on a trailer.

The above features are shown on Figure 1.3-4 of Reference 2 on page 1-21.

1.3.1.6 Horizontal Hydraulic Ram

The horizontal hydraulic ram is a hydraulic boom with a capacity of \$0,000 lb. and a reach of 6.55m (21.5 ft.) The ram will be mounted on a separate trailer for transportation and will be mounted on a surface supported tripod during the DSC pushing insertion or removal operation. Figure 1.3-5 of Reference 2 on page 1-21 shows the hydraulic ram.

1.3.1.7 System Operation

The primary operations (in sequence of occurrence) for the Oconee system are shown schematically in Figure 1.3-6 of Reference 2 on page 1-21 and are described below

- 1 <u>Transfer Cask Preparation</u> Cask preparation includes taking smears of the cask interior to ensure that the DSC will remain radiologically clean. These operations are done in the decontamination area inside the spent fuel pool area. The operations are standard cask operations and have been previously performed by Oconee personnel. Detailed procedures for these operations are described in Chapter 5, "Storage System Operations" on page 5-1.
- 2 <u>DSC Preparation</u> The internals and externals of the canister are verified to be clean. This ensures that the newly fabricated canister will meet existing Oconee specific criteria for placement in the spent fuel pool
- <u>Placement of DSC in Transfer Cask</u> The empty DSC is inserted into the transfer cask. Proper alignment is assured through the use of alignment marks on the cask and each DSC.
- 4 <u>Transfer Cask Lifting and Placement in the Spent Fuel Pool</u> The DSC transfer cask annulus is filled with clean demineralized water. The DSC cavity is also filled with borated water from either the spent fuel pool or an equivalent source of borated water. This prevents an inrush of pool water when they are placed in the spent fuel pool. This will also prevent contamination of the DSC outer surface by the pool water. The DSC transfer cask annular region is then sealed with an inflatable seal at the top to prevent mixing. The water-filled transfer cask with the DSC inside is then placed into the spent fuel pool.
- 5. <u>DSC Loading</u> Twenty-four spent fuel assemblies are placed into the DSC basket. This operation is identical to existing Oconee spent fuel shipping cask loading operations. These assemblies will be preselected to control reactivity and decay heat using the administrative controls on burnup, initial enrichment, and decay time detailed in Section 10.2.5. "Administrative Controls" on page 10-6.



- 6 DSC Top End Shield Plug Placement The DSC top end shield plug is placed inside the DSC using the overhead crane with lift beam attached. The top end shield plug is suspended from the transfer cask lift beam by cables and is emplaced as the set set set are sengaged to the transfer cask trunnions.
- Trans Cask Lifting out of the Pool The loaded transfer cask is lifted out of the spent fuel pool and placed in the decontamination pit. This operation is identical to existing Oconee cask lifting operations.
- 8. <u>DSC Sealing</u> The water level in the DSC transfer cask annulus is then lowered approximately 5-10 inches Swipes are taken over the DSC exterior at the DSC upper surface and around the circumference. The water level in the DSC is lowered away from the inside surface of the top end shield plug. Then a seal weld is applied to the outer surface of the top end shield plug. This provides the primary seal for the DSC.
- 9. <u>Transfer Cask DSC Drying</u> A pressure line is connected to the DSC and the water inside the canister is forced out by helium pressure. The water, which is removed from the transfer cask and the DSC, is returned to the spent fuel pool or routed to the Oconee radioactive waste processing equipment. The pressure line is then used to draw a vacuum to facilitate drying until the water content meets the design criteria.
- Helium Filling In order to ensure that no fuel and or cladding oxidation occurs during storage, the DSC is filled with helium (He). To accomplish this, a portable helium gas bottle is connected.

The DSC is then filled with He gas. After the DSC is filled with the inert gas, the filling line is removed and the DSC ports are plugged and welded closed.

- 11. Final DSC Sealing The top cover plate is positioned and seal welded. This provides a redundant seal at the upper end of the DSC. The lower end also has redundant seal welds, which were provided and tested during fabrication. This operation provides the double seal integrity of the DSC.
- <u>Transfer Trailer Loading</u> After helium filling and seal welding, the transfer cask lid is positioned and bolted in place. The transfer cask is then lifted onto the transfer cask skid mounted on the transfer *vailer and secured.
- 13. <u>Transfer</u> Once loaded and secured, the transfer trailer is towed to the HSM. This movement takes place completely within the Oconee plant site and owner controlled area.
- 14. <u>Transfer Cask HSM Preparation</u> At the Oconee ISFSI the transfer trailer is backed into position and the HSM front access cover is raised and removed. Next, the transfer cask lid is removed. An optical alignment system and the hydraulic skid positioners are used for the final alignment of the transfer cask and HSM.
- 15. <u>HSM Loading</u> After final 7 mment the bottom cover plate on the transfer cask is removed, and the DSC is then pushed into the SM by the hydraulic ram.
- 16. <u>Storage</u> After the DSC is positioned inside the HSM, the hydraulic ram is released from the DSC and retracted. The transfer trailer is pulled away and the HSM front access door is closed and tack welded. The DSC is now in storage within the HSM.
- 17 <u>Retrieval</u> For retrieval, the HSM access door is raised and removed and the transfer cask is positioned as previously described and the hydraulie ram is used to pull the DSC into the transfer cask. All coupling attachment, alignment, and closure operations are done in the same manner as previously described, but in reverse order. Once back in the transfer cask, the DSC and its cargo of canistered irradiated fuel assemblies are ready for shipment to a permanent repository or other storage location. Provisions will be made to return the DSC to the Oconee spent fuel pool if necessary.



1.3.2 TABLES

Dry Storage Canister	
DSC Basket	
Guide Sleeves (24)	
Spacer Disks (8)	
Support Rods (4)	
DSC Shell	
Shielded End Plugs (Top and Bottom)	
Cover Plates (Top and Bottoni)	
Drain and Fill Ports	
Grapple Ring	
Horizontal Storage Module	
Reinforced Concrete Walls, Roof, and Balanat	
DSC Structural Steel Support Assembly	
DSC Seismic Retainer	
Cask Docking Flange and Tie-Down Restraints	
Heat Shield	
Shielded Front Access Door	
Ventilation Air Openings (One Inlet, Two Outlets)	
Shielded Ventilation Air Inlet Plenum	
Ventilation Air Outlet Shie'ding Blocks	
Transfer Cask	
Cask Structural Shell Assembly	
Bolted Top Head Assembly	
Cask Lifting Trunnions	
Lead Gamma Shielding	
Neutron Shield Assembly	
Ram Access Penetration Cover Plate	
Transfer Trailer	
Heavy Industrial-Grade Trailer	
Cask Support Skid	
Skid Positioning and Alignment System	

Table	1-4 (Page	2 of	21.	Major	Systems.	Subsystems and	Components of	the Oconee	ISFS	51
-------	-----------	------	-----	-------	----------	----------------	---------------	------------	------	----

Hydraulic Ram System

Hydraulic Cylinder and Supports

Hydrau'ic System

Grapple Assembly



1.4 IDENTIFICATION OF AGENTS AND CONTRACTORS

The prime contractor for design and analysis of the Oconee ISFSI i: Pacific Nuclear Fuel Systems, Inc. of San Jose. California. HSM construction is the responsibility of the Duke onsite construction organization. Fabrication of transfer equipment and DSCs is also the responsibility of Pacific Nuclear Fuel Systems. Inc.

1.4 Identification of Agents and Contractors

Oconee ISFS: Safety Analysis Report

1-18

1-19

1.5 MATERIAL INCORPORATED BY REFERENCE

The Topical Report for the Nutech Horizontal Modular Storage (NUHOMS-24P) System for Irradiated Nuclear Fuel, originally submitted to the Nuclear Regulatory Commission by Nutech Engineers, Inc. (now Pacific Nuclear Fuel Systems, Inc.) on February 26, 1988 and amended in July 1989 is hereby incorporated into this SAR by reference.
1.5 Material Incorporated by Reference

Oconee ISFSI Safety Analysis Report

1.6 REFERENCES

- 1. Duke Power Company, Oconee Nuclear Station, Independent Spent Fuel Storage Installation, Environmental Report
- Topical Report for the Nutech Horizontal Modular Storage (NUHOMS-24P) System for Irradiated Nuclear Fuel, NUH-002, Revision 1A, July 1989.

.

Chapter 2. Site Characteristics

2-1

CHAPTER 2. SITE CHARACTERISTICS

Chapter 2. Site Characteristics

Oconee ISFSI Safety Analysis Report

2.1 Geography and Demography

Oconec ISFSI Safety Analysis Report

2.1 GEOGRAPHY AND DEMOGRAPHY

2.1.1 SITE LOCATION

The Independent Spent Fuel Storage Installation (ISFSI) is located on the Oconee Nuclear Station Plant site. The site is located in Oconee County, South Carolina, approximately 8 miles northeast of Seneca. South Carolina at latitude 34°-47°-38.2° N and longitude 82°-53°-55.4° W. Lake Keowee is located to the north and west of the site. The Corps of Engineers' Hartwell Reservoir is south of the site and Duke's Lake Jocassee lies approximately 11 miles to the north. Figure 2-1 on page 2-9 (based on Figure 2-1 on page 2-13 of Oconee FSAR) shows the site location with respect to neighboring states and counties within 50 miles.

2.1.2 SITE DESCRIPTION

Figure 2-2 on page 2-10 (based on Figure 2-4 on page 2-16 of Oconee FSAR) shows the site, property line, exclusion area, site structures and general features of the area. Figure 2-3 on page 2-11 is a detailed site layout showing the ISFSI location in relation to major site features. There are no industrial, commercial institutional or recreational structures within the site boundary. A visitors center and the Keowee Hydroelectric Station, both owned by Duke, are located within 1 mile of station center. Duke does not own the vacated Old Pickens Church and Cemetery, a small, historic property located east of the station which is not currently being used.

The topography immediately surrounding the ISFSI (Figure 2-3 on page 2-11) consists of relatively flat terrain which has been grassed or graveled over and is routinely maintained by the station. Routine maintenance of the immediate site vicinity assures that erosion will be minimal and that fire hazards due to burning vegetation are also minimized.

2.1.2.1 Legal Responsibilities for Site

All the property within the 1 mile radius exclusion area including mineral rights is owned by Duke except for the small vacant rural church plot belonging to Old Pickens Church, rights-of-way for existing highways and approximately 9.8 acres of U.S. Government property involved with Hartwell Reservoir.

The Hartwell property is either a portion of the Hartwell Reservoir or subject to flooding and not suitable for other uses. Duke has obtained from the owners of the church plot and from the United States the right to restrict activities on these properties and to evacuate them of all persons at any time without prior notice if, in its opinion, such evacuation is necessary or desirable in the interest of public health and safety.

The property which is within the exclusion area and which is not owned by Duke is shown on Figure 2-2 on page 2-10.

2.1.2.2 Other Activities Within the Site Boundary

Duke Power Company owns and operates the Oconee Nuclear Station and the Keowee Hydroelectric Station. The ISFSI is located within the owner controlled area of the nuclear plant. ISFSI operations have been considered for impacts upon th. Geonee station's facility operating licenses. Duke submitted a request, pursuant to 10CFR Part 50, to amend these licenses for the three Oconee Units to permit Duke to operate the ISFSI. This amendment request concludes that with certain minor modifications all aspects.

2.1 Geography and Demography

of ISFSI operation which are conducted within the existing Oconee station can be conducted safely while meeting the criteria for a "no significant hazards" finding.

All ISFSI operations are performed by the existing Oconec workforce. Only the transfer equipment used for the storage system is dedicated exclusively to ISFSI operations. No individual or group is dedicated exclusively to the ISFSI. Operational control of the ISFSI includes procedure: for the spent fuel pool loading steps and the subsequent transfer to the ISFSI. All procedural steps necessary for preparing the DSC transfer cask for transport from the fuel building to the ISFSI are completed. At this point other procedures for transport of the DSC transfer cask and emplacement of the DSC into an HSM are used.

ISFS1 operations required the following fuel building modifications:

- 1. Enlarging the opening of the cask decontamination pit covers.
- Shortening the projection from the spent fuel pool wall of the cooling system intake pipe. This is needed to provide clearance for the transfer cask in the spent fuel pool cask pit.
- The addition of a microdrive to the fuel erane positioning system to aid in the precision placement of the transfer cask.

The following auxiliary equipment is used exclusively for DSC transfer eask operations within the fuel building

- 1. Transfer cask lift yoke and extension momber-
- 2 Vacuum drying equipment.
- 3. Automatic welding equipment.
- 4. Slings for the transfer cask lid-
- 5. Cask pit depression platform.

Additional description of the ISESI and Fuel Building systems and facility is included in Section 4.4. "Operating Systems" on page 4-21

Other non-plant related activities are limited to the highways through the Exclusion Area. Duke's Visitors Center, recreation on the lakes, and the Old Pickens Church and Cemetery which are historical landmarks and will not be used for regular services.

2.1.2.3 Arrangements for Traffic Control

Arrangements have been made with the South Carolina State Highway Department to control and limit traffic on public highways in the Exclusion Area should it become necessary in the interest of public health and safety

2.1.3 POPULATION DISTRIBUTION AND TRENDS

The population distribution is based on the 1970 census. Table 2-1 on page 2-7 gives the population distribution within 10 miles of Oconec. The majority of citizens live in the cities of Walhala, Seneca, Clemson and Central, S.C. The area is largely rural and sparsely populated. The population projections for the 0-50 miles distribution around Oconec is given for year 2020 in Table 2-2 on page 2-8. These projections are based on the 1980 census. The population within the 10-mile radius is projected to approximately double by the year 2020 with no individuals permanently residing within the 1-mile radius of Oconec.

2.1.3.1 Transient Population

It is expected that Lake Keowee's 300 mile shoreline will be fully developed by the early 1990's at which time the estimated transient population will be 36,000. This estimate is based on development of lakeside lots, public access areas, and expanded commercial activities to take advantage of expanded recreational opportunities. There will not be any cottages within the Exclusion Area.

The visitors center, located on Duke Property just north of the plant and within the Exclusion Area, was host to \$10,000 people during its first 25 months of operation.

There are no large industries within 5 miles of the site therefore no industrial transients.

2.1.4 USES OF NEARBY LAND AND WATERS

Residential development of Lake Keowee's shoreline is expected to be the major use of the nearby land. Commercial development is anticipated to increase in response to the residential development. The waters of Lake Keowee are used for fishing, boating and swimming by the public through various public and private recreational areas.

The following description of land use and local red populations in Pickens and Oconee Counties in the 10-mile EPZ of the Oconee Nuclear Station is based on the Oconee Nuclear Station Emergency Plan as of August 1, 1988.

Pickens County hes within the 10-mile EPZ of the Oconee Nuclear Station. Involved are approximately 157.08 square miles of county territory and approximately 30.000 people. Also included are approximately 300 dairy cattle, 10 milk-producing goats, 243 head of swine, 2.938 head of beef cattle and 15 head of meat-producing goats.

Also, involved in the 10-mile FPZ are approximately 256 acres of vegetables. 47 acres of apples, and a large number of residential vegetable gardens

This area has approximately 1.297 acres of hay crops and 4.670 acres of pasture grass.

A large portion of <u>Oconee County</u> lies within the 10-mile EPZ of the Oconee Nuclear Station. Included in this zone are approximately 165.498 square miles of land and approximately 26.000 people, with the largest concentration in Seneca. Oconee County's 654 square miles are divided into 22.665 acres of cropland, 285.605 acres of woodlands, and approximately 127.333 acres that fall into a general category of 'all other' There are a total of 13 dames in the 10-mile EPZ.

The largest portion of land is devoted to crops such as soybeans, cotton, hay, wheat, small grain, and corn, apples, forestry, poultry, beef or dairying.

Production of meat, agricultural crops and milk for the 5-mile radius of Oconee Nuclear Station for 1980 was as follows

meat = 118 tons crops = 310 tons milk = 86.300 gallons

This data has not significantly changed since it became available in 1980.

2.1 Geography and Demography

There are two schools located within the 5-mile radius. Six Mile Elementary School - 522 students, and Keowee Elementary School - 299 students. Two special care institutions are located within the 5-mile radius. Harvey's Love and CareHome and Six Mile Retirement Center nursing homes have a total of 80 patients.

2.1 Geography and Demography

2.1.5 TABLES

Sector	0-1 Mile	1-2 Miles	2-3 Miles	3-4 Miles	4-5 Miles	5-10 Miles	lota
N	0	0	0	0	0	4()	4(
NNE	0	0	0	38	22	60	120
NE	9	0	0	115	235	2,000	2,350
ENE	0	22	38	108	112	681	961
Ŧ	0	0	0	140	417	670	1.22
ESF	0	0	51	70	131	1,326	1,578
SL	-0.7	0	.80	6	70	8,472	8,628
881		0		0	45	7,792	7,83
8	0	19	29	6	140	2.027	2.221
SSW.	0	6	0		112	7,000	7,118
SW	0	19	0	128	166	538	851
WSW	- 0	$B \rightarrow B$	80	181	35	1,102	1,411
II.	0	0	150	38	102	1,419	1,709
WNW	0	Picae.	22	51	26	1.456	1,558
NW	0	0	0	13	32	920	96
NNW		1.1	3	13	16	881	916
TOTAL	6	85	453	907	1.661	36,384	39,490



2.7

2.1 Geography and Demography

s 5.10 Miles 10.20 Miles 20.30 Miles 30 4 1.70 96 5466
- 429 Ilya 565
4 2767 7416 73.
8 2689 26130 7690
1 2850 51930 1653
9 47.20 8814 391
C9 6526 (0221) 5
2 2440 12406 5
4 4037 5661
13059 (1302) 1
02501 13281 1
4 3756 7123
211/2 7202 1
6 1247 1387
6 16.21 6417
0 1 8677 871
9 SS01SI 01538 9

2-8

2.1.6 FIGURES



Figure 2-1. General Location



2.9





Figure 2-2. Site Plan

-

1

2.1 Geography and Demography

d



Figure 2-3. ISFSI Layout



e,

2-11

2.1 Geography and Demography

Oconee ISFSI Safety Analysis Report

2.2 NEARBY INDUSTRIAL, TRANSPORTATION, AND MILITARY FACILITIES

2.2.1 INDUSTRIAL AND MILITARY FACILITIES

There are no large industrial or military facilities or activities within 5 miles of Oconee. No other nuclear facilities dictuding university research reactors are presently located within a 50-mile radius of Oconee Nuclear Station.

2.2.2 TRANSPORTATION ROUTES

Figure 2-2 on page 2-10 shows the major transportation routes within 1 mile of Oconee. There are no oil or gas pipelines within 5 miles of the site. The nearest railroad line or spur is located at Newry, SC which is outside the 5-mile radius from the plant

The nearest airport is the Clemson-Oconee Airport located approximately 9 miles to the south of the plant. The runway is oriented FNE-WSW. Pickens County Airport is located approximately 10 miles to the cast of Ocoree Nuclear Station. The runway is oriented in a NE-SW direction. Anderson County Airport is located approximately 23 miles SSE of the plant. It has two runways oriented as follows: NE-SW and NSW-SSE. The orientation of the NNW-SSE runway is not in a straight line toward Oconee Nuclear Station. The above information is based on the "Atlanta Sectional Aeronautical C....rt Scale 1 500,000" 37th Edition. September 25, 1986, published by the U.S. Department of Commerce. No structures which could cause damage as described in Reg. Guide 3.48, para 2.2 are located near the plant.

2.2.2.1 Description of Products and Materials

The highways passing through the 1 mile radius exclusion area are SC Routes 130 and 183 which carry local traffic only with infrequent trucking of hazardous chemicals and explosives since the general area is nonindustrial

Only small amounts of chlorine are stored on-site since chlorine is not used for condenser cleaning at Oconec. No individual container contains more than 150 lbs. of chlorine. The chlorine is used for drinking water purification and waste treatment, with three to five 150 lb, containers typically being in use. The maximum total number of containers on hand at any time is approximately ten.



2 - 13

2.2 Nearby In., Transp., and Military Fac.

Oconee ISFSI Safety Analysis Report

2.3 METEOROLOGY

2.3.1 REGIONAL CLIMATOLOGY

Western South Carotics is f_{ac} outh of major storm tracks but experiences higher precipitation amounts than the east cose, cose to $A \rightarrow b$ is ation in the lee of the Appalachian Mountains. A semi-permanent belt of high pressure would's which there are the regional climate. During the fall season, the area has a high probability of expression G_{ac} , thoughter stagnation during which the dilution rate for effluents is low due to low wind straight.

The Oconec plast sd, is identical on Lake Keowee which was established to provide cooling for the three existing Oconec b solution mass and future steam generating units as well as storage for Jocassee (pumped storage) and bc source (concertaional) hydroelectric stations. The topography in the vicinity of the site is moderately relating wind the keyal air flow is influenced to some extent by the contour of the lake. The prevailing wind are devided between the southwest and northeast quadrants due to the lake orientation and large scale prevailing $s(t) = t^2 - t^2$.

A complete develoption of regional and local wind data, including normal and extreme parameters can be found in Section 2.3 of the FSAR.

2.3.2 LOCAL METEOROLOGY

2.3.2.1 Data Sources

The accident usalysis nateorological data base, discussed in the Oconec SER Section 3.2.4. Units 2 and 3, is for the period March 15, 1970 - March 14, 1972. Joint frequency tables of wind direction, wind speed and atmospheric stability are shown in Table 2-3 on page 2-18.

2.3.2.2 Topography

Figure 2-4 (m page 2-24 shows the detailed topography within 5 miles of the storage site.

2.3.3 ONSITE METEOROLOGICAL MEASUREMENT PROGRAM

Meteorological measurements include wind direction and speed, horizontal wind direction fluctuation, temperature, vertical temperature gradient, and rainfall. The relative position of instruments with respect to station yard is noted in Figure 2-5 on page 2-25. Relative elevations of both surface levels and instrument levels are depicted in Figure 2-6 on page 2-26.

Wind measurements are made with the Packard Bell Model W S 101B series wind direction-speed system with starting thresholds of 0.7 and 0.6 miles per hour for direction and speed, respectively. Wind direction and speed are recorded in the control room on Esterline Angus Model A 601 C strip chart recorders with a system accuracy of \pm 5.4 degrees for direction and \pm 0.45 miles per hour for speed. Temperature and delta temperature measurements are made with the Leeds and Northrup 8100 Series 100 ohm resistance temperature device with Packard Bell Model 327 thermal radiation shields. Temperature and delta temperature are recorded on the Leeds and Northrup Speedomax W recorder with a system accuracy of \pm 1°F for temperature (at 10 m level) and \pm 0.5°F for delta temperature (46 m level referenced to the 10 m level). For data prior to February 24, 1977, delta temperature was measured at the 46 m level and the 1.5



2.3 Meteorology

m level. Rainfall is measured near the meteorological tower with the Belfort Weighing Rain Gauge Model 5-780 with an accuracy of ± 0.03 in and ± 0.06 in for zero to five and five to ten inch totals respectively.

Operational measurements consist of near real-time digital outputs in addition to the previously described analog system. An entirely new set of instrumentation has been installed including the measurement of dew point (at 10 m level); a supplemental low-level wind system (at 10 m level) provides input for emergency dose assessment (see Figure 2-5 on page 2-25 and Figure 2-6 on page 2-26). The rain gauge has been relocated near the supplemental wind system.

Instrument specifications for operational measurements are:

1. Wind Direction

- a. Manufacturer Teledyne Geotech
- b. Time averaged digital accuracy ± 3° of azimuth
- c. Time averaged analog accuracy \pm 6° of azimuth
- d. Starting threshold 0.3 m sec at 10° initial deflection
- e. Damping ratio 0.4 at 10° mitial deflection
- f. Distance constant 1.1 m.
- 2. Wind Speed
 - a. Manufacturer Teledyne Geotech
 - b. Time averaged digital accuracy \pm 0.27 m sec for speeds less than 27 m sec
 - c. Time averaged analog accuracy \pm 0.40 m sec for speeds less than 27 m sec
 - d. Starting threshold 0.45 m sec
 - e. Distance constant 1.5 m
- 3. Temperature
 - a. Manufacturer Teledyne Geotech
 - b. Time averaged digital accuracy = 0.3°C
 - c. Time averaged analog accuracy ± 0.5 C
- 4. Deha Temperature
 - a. Manufacturer Teledyne Geotech
 - b. Time averaged digital accuracy ± 0.10° C.
 - c. Time averaged analog accuracy ± 0.15°C
- 5. Dew Point
 - a. Manufacturer General Eastern
 - b. Time averaged digital accuracy ± 0.4°C
 - c. Time averaged analog accuracy $\pm~0.6^\circ\mathrm{C}$
- 6. Precipitation
 - a. Manufacture: Teledyne Geotech

11

- b. Digital accuracy ± 6% of total accumulation at 15 cm hr
- c. Analog accuracy ± 9% of total accumulation at 15 cm hr
- d. Resolution 0.25 mm

Near real-time digital output: of meteorological measurements are summarized for end-to-end 15 min. periods for use in a near real-time puff-advention model to calculate offsite dose during potential radiological emergencies. The Operator Aid Computer (OAC) system computes 'e 15 min. quantities from a sampling integral of 60 sec. It calculates 15 min average values for high _ 1 supplemental low level wind direction and speed: 15 min. averages are also calculated for delta temperature, ambient temperature and dew point temperature. Total water equivalence is computed for precipitation. A¹ 15 min. values are stored with a 24 hr recall. Permanent archiving of data from the digital system is made by combining the 15 min. quantities into one hour values.

2.3.4 DIFFUSION ESTIMATES

2.3.4.1 Basis

The design two-hour X-Q at the Exclusion Area Boundary (EAB) for accidental releases is 4.5E-4 (sec m³).

2.3.4.2 Calculations

The calculation of a two-hour X Q value to estimate radiological doses from potential accidental releases from the storage site (See Figure 1-1 on page 1-4) is based on a plant design condition of Fasquill Type F stability with a wind speed of 1m sec as proposed in the Oconec Safety Evaluation Report. Section 3.2.4. Units 2 and 3 The equivalent design condition [95 percentile hourly average X Q] for the ISFS1 is a Pasquill Type F stability with a wind speed of 0.65 m sec. The ca' ulation assumes a gaussian material distribution from a ground level release with essentially a point source geometry.

$$X/Q = [\overline{u} \pi \sigma, \sigma_s]^{-1} = a.5 L - 4(sec m^2)^*$$

Where D

ø, (1.0 mi.)	*	crosswind concentration distribution standard deviation = 57	7.00
e. (1.0 mi.)		vertical concentration distribution standard deviation = 19 m	1

mean wind speed at 10 m = 0.65(m sec)

* Slade, D. H. (ed.) 1968: Meteorology and Atomic Energy 1968, T1D-24190. National Technical Information Service, Springfield, Va.



2.3 Meteorology

Oconce ISFSI SPCrty Analysis Report

2.3.5 TABLES

-

82.00 60"8 63.78 81-2 6115 95"5 59-5 5315 96-12 101.10 65"# 434 5% 92 543 822 955 254 854 410 615 2445 24 19101 *4 88 8 134 2% 9.153 Class 10-01 60.00 54"8 5274 52"0 69"5 10'4 86"2 - -18"0 45.4 \$34 00.0 . 2 2 . -14 ** 22 25 DN-5'246 -Stability 01.0 ----28"4 41.14 67.54 10'0 10.9 56'64 80"2 61'6 90.78 \$34 44 28 48 23 10 \$25 08 6-215 4 . -the second second 81'8 1010 68"6 14'8 50 8 10"0 14" 4 65"8 6.22 58'8 62-0 \$30 - -153 15 5-262 +2 . 34 612 -. . -84 . ----3 60"@ 10.4 10.12 64.10 41.15 55"0 58° R 12"4 16 3 58"8 62"1 534 -#-Speed . . -4.0 22 . 56 -. 140.5 1244 81822 * 0110 4.5-10 23"2 64"# 55.18 20 " 5 -050-60.0 61"0 21.14 25.75 64'6 134 5.5 4 +1 28 41 44 11 23 44 . 100 (Gine \$ 242 and 2011 134 55"5 -25-2214 62"8 43 "0 26 "8 23"8 42"0 95"# 66"8 84 "2 44 \$52 14 885 24 4'672 4 \$ 24 44 10 44 16 Direction i de la com 10-0 00"0 68.9 65.4 6" 35 28'8 44" 6 41.10 16"8 10"8 \$9"2 230 -#55-5-242 25 2.4 54 543 64 . 646 224 -----60' 8 41.4 94"1 274 -5-50"# 50"0 0176 67.54 16.18 5574 46.4 89"0 . 2 +1 48 5.5 45 11 4.9 2.1 642 1544 9"041 1 Wind 87.95 61-10 -355--00 60.0 10.12 55"8 80.08 22.5 46 " 5 44.4 20'8 134 ٠ -24 14 26 44 44 44Z 54 5-258 . Ju i -25-661.6 64.0 75*8 01-0 174 60"# 10.15 20"0 55"8 55"5 15-8 80 "0 22 18 622 8-515 Frequencies 05 16 52 586 -14 --5919 25"0 60"8 80"4 92-0 25.0 11-0 24.4 \$34 -853-50"3 60.90 5'66 44 455 6-295 . * . . æ 22 28 68 46 10.04 64"0 64"6 65"8 55 "0 ----6178 22-8 45.10 66"8 160.10 65-8 134 -3-. . ٠ 2 . 41 24 25 4.8 ٠ 6.84 24 0.04 Joint 0.00 334 - 543-50"2 25.18 28"8 15"3 22.0 0710 50'0 22-3 60"6 10"0 . #1 2.4 25 42 15 24 . 645 -6.58 1 . 45'8 55"0 92"8 8276 22.4 44.4 20.0 42*1 3.74 - -80"9 85"8 36'6 6) . 4 . 23 62 44 -421 415 Ges 6.24 50 86"8 68'8 0410 50"0 20.74 64"4 14" 9 62"4 65"8 10.5 64-8 . 134 -----52 6-22 . * * -61 -94 1046 . . . --(Page 00"0 85'6 44.46 60"8 54"4 32.4 11.4 61'6 24-0 \$74 - 10-65"8 5.60 16.5 29 5.5 255 -41046 . ---. ------SAN 5"8-4 40"8-5"8 44"8-5"2 48"2-5"8 48"8-5"5 48"8-5"8 48"8-5"8 48"8-5"8 48"8-5"8 45"1-55" 100 Man 2"124 2"12-1"61 8"42-8"91 2"91-9"91 6"91-9"21 6"21-1"61 2"61-6"2 5"5-8"5 2-6-8-1 196653 MARS 5 265150 8-2-8-5 ri. SIN 13 BIGGE GRIA 265 235 编编员师 0415 0b 862061 2-59-15 able V STRAFTE AL ARTHONY the station is way '12"face than we' le'fact GEBARE MELEDARI BREENE EINERE 13468 13468

SASAS SHOTS THE READ WARS

EEEEE SAGELYADISSO DE 101 10101

	and some officer and the	the second se	14 March 14	
A Description and	16 1 6 1	Sec. Sec. Barrier	A second second	A CONTRACTOR
 Revenues 	1761 761	26.261616.5	A CALVING	PU C 1 K M 1
			Contraction of the second second	

.

.

2.3 Meteorology

ATE BE BEFORE 5-66-52	100 2"124 2"12-5"8 4"	0.0 0.00 0.00		10-0 00-0 10		aara 10-0 10	1 1 1 1		8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8		10'1 04'5 4'04				••••		1 0 0 1	A 1 M IN CONTRACT OF A 1 M IN	04-70 10-00 50-		10° 0.07 0.08	•	10°0 0°00 0°0		1010 0018 101		101 a la a la		14-2 14-2 14-	5 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4		847 B
	14.4-14.7 14.5-1					0.0× 0.0	•	6.03	• :		**		•	A015		A LANK			4° 10 0		6.01		0.01 0.		6.01		6.31 0.	*	9.63	•	20.2	
	CL055 12.4-14.5	•	00 ° 0	****		a.a		6.03	•	0.00	•		•	00.0	•	10.5	•	12.2	10.0		0.01		2.03	4	8.63		80'6		8.01		8.93	
3.900 from + 1	0 10.1-12.3 • 4.5-9.40	•	8.90	10.0		69	•	·		0.63		0.0	•	6.6	•			10-0	11										24-8		50°6	
18 16 16 1 B.L	1.9-10.		6.03	10.0		90° C	12	99.0				e.a		8°*8		5.0		B. 6							8.8		4.4		3.6		#~#	 and the second s
READ CR.COM	8.1-4.8		1979			10.0		10.0		10.01	23	80.9	11	868		4074		1040		a number of the second s	•••				10.0	0	08.0		10.6		54.4	
	875-871		8.85	•			**	18.8		8.04		56'3		e.14		80.9		10.0	63				***		e.es		10.9				8.03	
1	1.8-8.1		0.62	•						6.03		9.93		00		10.0		10.01	**	0.64	***			36-6			9.91		8.83		6.35	
*******	19161	44	8.14	1	6.20			8.76		4.7.8		22.22		·		8.78			£	25-0	=	15.4		3.24			===		82 B		e. 10	8.00
34 91	-	1	12	1	174	8		8 2	1	22	1	:24		12		12	1	12	1	374	ş	134		814		2					22	 104
Punker	1140			\$. 42	- 114	6.10		61. S				-16		- 14-		5		0	5.30	- 495	0.12	-14-	2.7.5				5.24	-940	9-14			-

*

2.3 Meteorology

SI Safety Analysis Report Oconec

	100	
τ.	28	
÷	25	
	-	
	6	
÷		
Ξ.	100	
а.		
۰.	2	
2	32	
Ε.	-	
ε.	20	
÷		
ε.	100	
÷.		
1.		
	25	
۰.	20	
2	×.	
Т.	T	
1	100	
÷	-100	2
1	1	B
Έ	22	
1		8 H.
	-	
	8	8 C
ε.	-20	2
т	100	ε
2	20	£
ε.	2	F
÷		
1	-	ε.
2		
٠	-	
а.	- 22	
2	- 200	81.1
Ξ.	de.	8.
т	-	
÷.	Sec.	
ъ	100	
	-	
1	22	
т.	- 20	
	- Said	
5	100	
6.	5	£1 -
Ξ.	-34	£
E	- 200	8 C (
τ.	- 25 -	2
3	12	
	100	8. T
	-	£
ъ	-	
8	166	
	100	£
8		2
5		
10		
2	-	2
а.	160	8
2	-	a
	36	
1	-	
18	-	2-1-
	100	
1	100	2
	100	
	100	
	-	
1	orders.	1
	THE	
	-	
1	100	1.
18	die.	1
	and the second s	-

Table

63*# 67	54-0 5	\$176 52	12"8 45	84.18 10	19'8 16	****	44'-1 111	62*5 <u>911</u>	1978 	85"5 858 88"8	134 84 134	1014 1101
10-0	99.6	20.0	85'B		89°8 8	20-0	9970	90."S 4	16'8 9	01'9 92	13.4 84	- Puil- 5'585
1 62°8	19'0	20.0	10'0	10"0	10.0	10.0	18.8	90'8 8	01'0 95	62°0 96	574 (M	-**- @*\$\$6
w.e	10-8	20.0	8-00	10-0	10.9	10.0	10'0	9°0	19-0 1	45°0 98	174 88	- PMP- 5-242
60"B	00'0	6.62	50.0	10°0	ca-a	£8*3 6	89*8 5	95 10-9	56'8 8	95 "8 95	13d 99	-8- 6-822
29.0	66.8	50'0	36.6	50.0	60°0	19.0	58°9	69.8 63	10'5 5	82°8 86	334 28	-#5#- 6"298
	40.0	50.0	44-8	81 °8	****	8878 61	43"8 52	91'9	16"# 2	89.8	130 Gm	-#5- 8*622
58°5	10'0	34.0	60.0	96'8	05'0 58	50°.8 8	1120 95	50'B 5	20"6 6	55°9 65	134 68	-#\$1- \$*202
	99"8	8-65	28.0	48'6	50.0	£0.0	50°8 1	58°9	50°8 8	81 78 69	920 980	4.905
8 68*8		62"3	10'0	10'0	18-8	67°5 8	29"3 65	91'8 93	9 9	85.8 85	174 24	-345- 6*268
6 66''ë	32"6	6.90	89.0	88'0	10.6	sore	50°.0	49'8	68"N	5878 66	134 69	-25- 4*653
00.0	80'8	90"0	46"8		86'8	20.9	58'8	80"8 25	58'6 9	93-9 52-8	134 84	-959- 1-288
		30'8	18 '9	50-0	90'0	60°8 9	61.0 61.0	91"@ 92	9479 9	19 19	134 96	-8- 8*85
40'9	8:35	20"0	00"0	***** *	45°0 53	93' 9	10.6	56"8 4	88-0 9	85 "# 85	134 88	-262-
aare	10.0	14-9	80°0	61"8 41	53"# 01	40°0	68*# #1	93 10.8	50°8 8	9018 54	174 68	- 34- 8*64
-	86°8	8° 66	00'0	9°04	ça's 9	80'8 15	60.0	68*8 8	10"0 2	96'96 69	324	-3mi-
86-3		10.0	80.8	10.0	18"8 8	58°8 6.	20.0	20"0	48°9 85	95.0	174 68	8.04
1.76 6"6-	-4 29'6-6"	8 as'8-6'	1 69-1-6-9 1 69-1-6-9	49"9-6" 8 "96-9"	5 60'5-6"5 21 5'23-1"0	69"9-5"E	43*5-5*2 8*6-5*8	6*5-6*8 6*5-6*8	1-6-9-3	14165 186268		1011331 9359
24-48-5	965-861	40 31 10 12 10 2 30 34 10 10 10 10 30 30	*980330 *88	88 8 67 822 51 95 8 680 81	2245 + 10223 1924		9146 bå	in 1931 1 34 1975	1421 20 JAN	6 71000 68 #5 45	84 o ¥2 6696	A197102-01

2-20

3		d,	Ŀ
6			
덻			
- 1			

-	60	664	i.	
10				Ŀ
				B
				9
				8
- 16				r.

11 13 2 <th2< th=""> 2 2 2</th2<>
m 11 m 11 m 10 11 21
16 17 10 27 11 14 18<

2.3 Meteorology

Passestances are a

1010

245.55

SALE AND SO BORSEANSING

2.3 Meteorology

Oconee ISSSI Safety Analysis Report

<u>4</u>	<u>i</u>	21	<u>67.0</u>	55 W	6610 665	903 3 C - 8	41°6 162	66°3	166-2	65°81 9592 28°8 5	174 04 174 08	1038 1979
	80.0	00°3	10.0	50.0	65°#	11-5	24"0	58*1 553	82.8	\$5.5	124	
18.9	3	10.0	10.0	3	*	+1	45	645	52'8	992	174	0.252
8	10°9 2	10.0	6979 6	****	44.5	11-0	62°4 62	19'8	41'0 42	211"3 458	274 94	
68"6 5	59'6 2	*****	90.0 83	10.0	01'0 68	\$1"S	0-10 1+	0%"0 85	975# 96	42"5 5#5	134 84	
00"# #	9879 0	10°G 2	60"8 8	40'D	61'6 61'	12"0	6-30 54	9.24	40°0 41	44"0 245	434 08	-======================================
16"4	80"B	50°70 8	69"B 8	61 °6 51	41°#	*2"# 56	22.0	96'0 49	11'9	32.4	134 0m	-#5- 0"520
94°9	30"8 8	1419	20~6 1	90'0	59-00 23	45 11.4	4750 55	12.6	50"8 9	0° 86 854	134 6%	- #15-
00'# #	00"0 0	68-5	00°0 0	50'0 5	\$1~# \$2	91'8 55	8578 92	41°4 11	41'2 51	54 °4 803	134 64	-5-
00'0 0	96.5	54°8 8	09°5	e 20.6	68'8 5	23-5	85°8	91'0 92	40-8 01	85"@ 18	1.34 64	-#55
8 8 8	98°90 0	8.06 8	00.00 5	60°0	20°5	40"6 61	#1"# #2	57°0 16	90.06	85'D	174 0%	-35- 8*56
60'9 8	00"S	80°8 8	50°6	00.9	10.0	£8".6 3	90°-9 21	13'0 54	40'9 11	16.8	174 Ch	- 35.5
60°8	80'8 8	94.05 9	00"0 0	89.2 ·	20"3 6	60°8 5	12-8 16	12-9	13'0 43	19'0	634 08	->- 0'64
28"# 2	10°5	00"0 2	50'0 5	8.00	60'8 6	5.0°.0	227-0 15	56°8 85	80'0 21	\$4 "9 908	134 100	-342
60°5	0414	04-6 8	10~6 0	10.0	60'8 9	43°4 63	35.4	95'0 19	95 'B 62	41'5 943	574 109	- 24-
	90°0	08''B 9''	10'0	10.0	10.0	20-0	65"# 94	50'0 54	41.8	8%*1 612	-174	- 200 fg 5 * 2 4
98'à 0	69'8 0	80'0 8	0.60	8 99'8	9à'8	37°6 26	58*1 951	28-1	92'0	995	174 84	0.01
	< 65"6 - 5"B	44'8-5's	45"2-5"5 8 4"92-5"9	60 '0-5'6 5'5'-9'2	\$9"\$-5"9 1 1"21-9"0	4.01-9.1	81-5-5°2 8'1-9*5	*** 2-5*2 5*5-5*5	49"1-59" 4"1-6"1	W161		80119

2.22

	18 6"4-C	80"8	• 6	. eo. e	- 80 - 60 - 60	B 00''B	9.90	eo' e	* CO *			G 80. *	19.8	e	80''5	90'8	10.0	1.
	45"4-5"4	* *			e .e	* 0.4	0.00	e.e.e	e.eo	•	8.00	•:	8 9	* .0 e	• 8		8 8	1
1. 40	1.5-9.49	0.00 0.00	•	e.e.e	. 8	6.83	8°.08	• · ·	0 6 G	• .00	0.00.0	e.		9.01	•.00.	1 8,91	8 9.80	[
1001 DC(04	6.9-1.49		0.00	• •••	0.90	0. 0 . 0	• • • •	ee	*		8.90	9.00	e 9		* 19-18	8.90 S	a. 00 a	*
***	1.1-1.1	0.00	8 c 6 ° 0		8 ° 8 ° 8	10.01		0 °0	. 9 8	8 ° 8 °	0 00 °0	10.6	6.02	4 U U	8 8.82	* °0	- .	1
AD SPEED SA	1.1.1.1.0	8 9-34	÷.00	10.0	10.01	1.9.9	10.6	8	* : • •		*.0.*	9 ° 8 °	8°.0	9.01	* 0 · 0	4 9.63	\$.e3	(ⁱ
	9.9-4.49	48 8.32	8 6.0	* · · 0	3.09	10.0	10.0	0.16 0.16	9.65 e	6.01	10. C	52	94 6	46°.6	\$ 0.0 F	4 4 4	42.0	£
	3-2-5-2	1110	36 6.75	43.0	34.0	11	• • • •	44	82 8.1.8	11.8	• • •	9.92	62°€	11.6	41.9	: : : :	4. B	116
	3.5-9.9	11.00	64 5.16	1.4	31.8	48.0	::.	42°8	82.0	3.5 9.4 B	81 ²³	87.8	82°8	37.0	61. 0	8.98 8.38	109	H.
	*********	52 0.24	38		10.0	1 .95	8.94 8	11 0.00	11 °°		8, 08	84 9.13	83.15	91 89	8. 18 8. 18	9.24	35 0.43	E
SFC TIM	13141	310	1.90	84.8 8.68	85.0	93 0. 33	15 .0	102 0.71	88 8.45	85 85	54 D. 43	241'B	111	::	818 91 8	111	242	5.07 1976
81 TR	13.54	85	80	85		85	554	85	08	85	85	08	-	82	504	82	54	122
-	AC CO A	* · · #	- 14	49.0	6 1. S	1.96	- 35 8	19.0	12.1	-1-		-14-0	1.15	0.9	1.15	13.0		AL.M.

2.3 Meteorology

2.3.6 FIGURES





2.3 Meteorology



Figure 2-5. Relative Positions of Meteorological Instruments



1

2-25

2.3 Meteorology

Oconee ISFSI Safety Analysis Report



Figure 2-6.

1

1

Relative Elevations of Meteorological Instruments

2.4 HYDROLOGIC ENGINEERING

2.4.1 HYDROLOGIC DESCRIPTION

2.4.1.1 Site and Facilities

The location and description of Oconec presented in Chapters 1 and 2 include reference to figures showing the general arrangement, layout and relevant elevations of the station. Station yard grade is 796 ft. mean sea level (msl). The mezzanine floor elevation in the Turbine. Auxiliary, and Service Buildings is 796.5 ft. (msl). Exterior accesses to these buildings are at elevation 796.5 ft. (msl).

All of the man-made dikes and dams forming the Keowee Reservoir $r \rightarrow an$ elevation of 815 ft. msl including the intake channel dike. The crest of the submerged weir in the intake canal is at elevation 770 ft. msl

Flooding at the ISFSI will not occur. Figure 2-2 on page 2-10 shows the location of the ISFSI at the Oconee site, and Figure 2-11 on page 2-60 shows the relative location and topography of the ISFSI yard at Elevation 825.0 and the surrounding terrain features, including the Keowee dam and dikes. The Probable Maximum Flood level for Lake Keowee, as defined in Section 2.4.2.2, "Flood Design Consideration" on page 2-28, is Elevation 808.0, which is seventeen feet below the ISFSI site yard level of Elevation 825.0. Also, the peak flood level due to a postulated failure of the upstream Jocassce Dam is Elevation 813.12, as discussed in Section 2.4.5.1, "Flood Protection Measures for Oconee Station Seismic Class 1 Structures" on page 2-31. Thus, since all of the man-made dams and dikes forming Lake Keowee are constructed to an elevation of 815.0 and since the ISFSI site elevation of 825.0 is above the maximum lake level which can be maintained, there is no potential for the reservoir level reaching the ISFSI site by overtopping. Therefore, flooding of the ISFSI will not occur.

The ISFS1 yard is surrounded by drainage intercept ditches sized to prevent local overland flow from reaching the ISFS1 site. In addition, stormwater drainage is provided in the paved areas of the ISFS1 site.

Therefore, flooding of the ISFSI site cannot occur either due to reservoir overflow or local intense precipitation.

2.4.1.2 Hydrosphere

The main hydrologic features influencing the Oconee plant site are the Jocassee and Keowee Reservoirs. Lake Jocassee was created in 1973 with the construction of the Jocassee Dam on the Keowee River. The lake provides pump storage capacity to the reversible turbine-generators of the Jocassee Hydroelectric Station. Jocated approximately 11 miles north of the plant. At full pond, elevation 1110 ft. msl, Lake Jocassee has a surface area of 7565 Ac. a shoreline of approximately 75 mi, a volume of 1,160,298 Ac-ft., and a setcl drainage area of about 148 sq mi.

Lake Keowee was created in 1971 with the construction of the Keowee Dam on the Keowee River and the Little River Dam on the Little River. Its garmary purpose is to provide cooling water for the plant and water to turn the turbines of the Keowee Hydroeicstric Station. At full pond, elevation 800 ft. msl. Lake Keowee has a surface area of 18.372 Ac. a shoreline of appacetimately 300 mi, a volume of 955,586 Ac-ft., and a total drainage area of about 439 sq mi. The Jocassee and Keowee Reservoirs and the hydroelectric stations located at these reservoirs are owned and operated by Duke.



2.4 Hydrologic Engineering

The area presently provides for a few raw water users. The City of Greenville and the Town of Seneca take their raw water supplies from Lake Keowee. The Town of Anderson, the Town of Clemson, the Town of Pendleton, Clemson University, and several industrial plants take their raw water supplies from Hartwell Reservoir, downstream of Lake Keowee.

Greenville's raw water intake is located approximately 2 miles north of the plant on Lake Keowee. Seneca's raw water intake is located approximately 7 miles south of the plant on the Little River Arm of Lake Keowee. Anderson raw water intake is located approximately 40 river miles downstream of the Keowee tailrace.

The existing raw water intakes for Greenville, Seneca, and Anderson are shown and located relative to the site on Figure 2.4.1 in the ISFSI Environmental Report.

2.4.2 FLOODS

2.4.2.1 Flood History

Since Oconce is located near the ridgeline between the Keowee and Little River valleys, or more than 100 ft, above the maximum known flood in either valley, the records of p: it floods are not directly applicable to siting considerations.

2.4.2.2 Flood Design Consideration

In accord. . with sound engineering practice, records of past floc ds as well as meteorological records and statistical procedures have been applied in studies of floods routed through the Keowee and Jocassee Reservoirs as a basis for spillway and freeboard design

The spillway capacities for Lake Keowee and Jocassee were selected in accordance with the empirical expression for design discharge

$$Q = C \cup DA$$

Where Q = peak discharge in cfs

D A = drainage area in square miles

C = 5000, a runoff constant judged to be characteristic of the drainage area

The following tabulation gives pertinent data on this design flood flow.

2.4 Hydrologic Engineering

Lake Keowee(1)	Lake Jocassee	
439	148	Drainage area at damsite, sq mi
25,200	21,000	Maximum recorded flow at nearby USGS gsgcs, cfs D A
Newry Gage. D A 455 sq mi)	(Jocassee Gage, D A 148 sq mi	
8-13-40	10-4-64	Date of maximum flow
1939-1961	1950-1965	Period of record
105,000	61,000	Spillway design discharge, cfs
800	1,110	Full Pond elevation
815	1.125	Crest of dam elevation
0	0	Surcharge on full pond for design discharge
4	2	Number of spillway gates
38 ft x 35 ft	40 ft. x 32 ft	Size of spillway gates Discharge capacity, cfs
107.200	45,700	Spillway
	16,500	(2 units Dependable flood flow of 4) through units
	and the second se	Total discharge capacity, cfs
107,209	62,200	

Note: 11 Little River and Keowee River Arms

•

1.

The above discharge capacities assume no surcharge above normal full pond level. Statistical analyses have shown design reservoir inflows for both Lake Keowee and Lake Jocassee equal to respective design discharge capacities outlined above to have recurrence intervals less frequent than once in 10,000 years.

The maximum wave height and wave run-up have been calculated for Lake Keowee and Lake Jocassee by the Sverdrup-Munk formulae. The results of these calculations are as follows:

Wave Height	Wave Run-Up	Maximum Fetch	I.ake	
3.70 ft.	7.85 ft.	8 miles	Keowee (Keowee River Arm)	
3.02 ft.	6.42 ft	4 miles	Jocassee	
3.02 ft.	6.42 ft.	4 miles	Keowee (Little River Arm)	

The wave height and wave run-up figures are vertical measurements above full pond elevations as tabulated above.

Studies were also made to evaluate effects on reservoirs and spillways of maximum hypothetical precipitation occurring over the entire respective drainage areas. This rainfall was estimated to be 26.6 inches within a 48 hour period. Unit hydrographs were prepared based on a distribution in time of the storms of October 4-6, 1964, for Jocassee and August 13-15, 1940, for Keowee. Results are summarized as follows:



2.4 Hydrologic Engineering

Oconec ISFSI Safety Analysis Report

Keowee	Jocassee	
147,800	70,500	Maximum spillway discharge, cfs
808.0	1114.6	Maximum reservoir elevation
7.0 ft.	10.4 ft.	Freeboard below top of dam

While spillway capacities at Keowee and Jocassee have been designed to pass the design flood with no surcharge on full pond, the dams and other hydraulic structures have been designed with adequate freeboard and structural safety factors to safely accommodate the effects of maximum hypothetical receipitation. Because of the time-lag characteristics of the runoff hydrograph after a storm, it is not considered credible that the maximum reservoir elevation due to maximum hypothetical precipitation would occur simultaneously with winds causing maximum wave heights and run-ups.

The maximum Keowee tailwater level during hydro operation has been calculated to be elevation 672.0 ft. (msl), which is 124 ft, below the nuclear station yard elevation 796.0 ft. (msl) and 153 ft, below the ISFSI vard elevation of 825 ft. (msl)

The maximum discharge calculated, due to hydro operating, is expected be 19,800 cfs. The minimum discharge calculated with no units operating, is expected to be 30 cfs

In summary, the above results of flood studies show that Lakes Keowee and Jocassee are designed with adequate margins to contain and control floods which pose no risk to the ISFSI site.

2.4.3 PROBABLE MAXIMUM FLOOD ON STREAMS AND RIVERS

2.4.3.1 Probable Maximum Precipitation

See Section 2.4.2.2, "Flood Design Consideration" on page 2.28.

2.4.3.2 Runoff and Stream Course Models

See Section 2.4.2.2. "Flood Design Consideration" on page 2-28.

2.4.3.3 Probable Maximum Flood Flow

See Section 2.4.2.2. "Flood Design Consideration" on page 2-28.

2.4.3.4 Coincident Wind Wave Activity

See Section 2.4.2.2. "Flood Design Consideration" on page 2-28.

2.4.4 POTENTIAL DAM FAILURES, SEISMICALLY INDUCED

Duke has designed the Keowee Dam. Little River Dam. Jocassee Dam. Oconee Intake Canal Dike, and the Intake Canal Submerged Weir based on sound Civil Engineering methods and criteria. These designs have been reviewed by a board of consultants and reviewed and approved by the Federal Energy Regulatory Commission in accordance with the license issued by that agency. The Keowee Dam, Little River Dam. Jocassee Dam. Intake Canal Dike, and the Intake Canal Submerged Weir have also been designed to have an adequate factor of safety under the same conditions of seismic loading as used for design of Oconee.



The construction, maintenance, and inspection of the dams are consistent with their functions as major hydro projects. The safety of such structures is the major objective of Duke's designers and builders, with or without the presence of the nuclear station or ISFS1.

2.4.5 FLOODING PROTECTION REQUIREMENTS

2.4.5.1 Flood Protection Measures for Oconee Station Seismic Class 1 Structures

The Oconee Station plant yard elevation is 796.0 ft. msl and the ISFSI yard elevation is 825 ft. (msl). All of the man-made dikes and dams forming the Keowee Reservoir are constructed to an elevation of 815.0 ft. msl with a full pond elevation of 800.0 ft. msl. However, Class 1 structures and components at the station are not subject to flooding since the Probable Maximum Flood (PMF) would be contained by the Keowee Reservoir. The minimum external access elevation for the Auxiliary. Turbine, and Service Buildings is 796.5 ft. msl which provides a 6 in water sill. Also, the plant site is provided with a surface water drainage system that protects the plants facilities from local precipitation.

In the Oconee PRA study, a postulated failure of the upstream Jocassee Dam resulted in a peak flood elevation at Keower Dam of Elev. 813.12, which gives 1.9 feet available freeboard. Although the connecting canal between the two arms of Lake Keower would lengthen the travel time of the flood wave, it is conservatively assumed that the water level resulting at Oconee Intake Dike would be the same as for Keower Dam.

2.4.5.2 Flood Protection Measures for ISFSI Site

The site for the ISFSI is elevated well above the nominal plant yard grade at E1. 825.0. Flooding of the ISFSI is not a credible event; therefore, no flood protection prevention measures are necessary.

2.4.6 ENVIRONMENTAL ACCEPTANCE OF EFFULENTS

The only liquid used for the ISFSI is during preparation of the DSC and transfer cask within the confines of the plant Auxiliary Building. No liquids are used during the actual operation of the ISFSI.

2.4.7 SUBSURFACE MYDROLOGY

The Independent Spent Fuel Storage Installation provides for the storage of spent nuclear fuel in a dry condition. Therefore, there will be no consumption of groundwater or impact to the groundwater system as a result of installing the ISFSI at the Oconee Station.

2.4.7.1 Groundwater Usage

The completed field survey of approximately 30 wells performed in the late 1960's determined that groundwater usage is almost entirely from the permeable zones within the saprolite with only minor amounts obtained from the underlying fractured bedrock. Yields from these shallow wells are low, generally less than 5 gpm, and are used to supply domestic water for homes and irrigation of lawns, gardens, and limited amounts for livestock. With only a few exceptions, the wells are hand dug, equipped with bucket lift and or jet pump, and 40 to 60 ft. deep. At present, there is no industrial demand for groundwater within the area. The only appreciable groundwater draft observed is being supplied by eight wells for Keowee High School, located four miles west of the site.

2.4 Hydrologic Engineering

2.4.7.2 Regional Groundwater Conditions

The Oconee Station lies within the drainage area of the Little and Kec⁻⁻ \circ Rivers which flow southerly into the Seneca River and subsequently discharge into the main drainage, course of the Savannah River. The average annual rainfall at the site area is approximately 53 in

The deposits of the Little and Keowee drainage basin are generally of low permeability which result in nearly total runoff to the two rivers and their numerous tributary creeks. Runoff occurs soon after precipitation, particularly during the spring and summer months when the soil percolation rates are exc. Go by the short term but higher yielding rainfall periods. The area is characterized by youthful ne. streams and creeks which discharge into the mature Little and Keowee Rivers.

Throughout the area, groundwater occurs at shallow depths within the saprolite (residual soil which is a weathering product of the underlying parent rock) soil mantle overlying the metamorphic and igneous rock complex (Reference 1 on page 2-87). Refer to Section 2.5, "Geology and Seismology" on page 2-43. This saprolite soil, which ranges in thickness from a few fact to over 100 ft, is the aquifer for most of the groundwater supply. Wells are shallow and few exceed a total depth of 100 ft. Depths to water commonly range from 5 to 40 ft below the land surface. Seasonal fluctuation is wholly dependent of the rainfall and the magnitude of change may vary considerably from well to well due to the limited areas of available recharge. Average fluctuation is about 3 to 5 ft. Both surface water and groundwater in this area are of low mineral content and generally of good quality for all uses.

To determine the general groundwater environment surrounding the plant area, groundwater levels were established in numerous domestic wells and exploratory drill holes during the original program in the late 1960's within a four-mile radius. Additional data was obtained from interviews with local residents regarding specific wells and discussions with State and Federal personnel. The results of the groundwater level survey are shown on Figure 2-7 on page 2-38. The results demonstrate that local subsurface drainage generally travels down the topographic slopes within the more permeable saprolite soil zones toward the nearby surface creek or stream. Gross drainage is southward to the Little and Keowee Rivers which act as a base for the gradient.

Because the topography and thickness of the residual soil, overlying bedrock control the hydraulic gradient throughout the area, and further, the relief is highly variable within short distances in is not possible to assign a meaningful average gradient for the 15 square mile area surveyed. In all small areas studied within the four-mile radius, the groundwater hydraulic gradient is steep and conforms to the topographic slope. Water released on the surface will percolate downward and move toward the main drainage channels at an estimated rate of 150 to 250 ft. per year.

The gradient throughout the area represents the upper surface of unconfined groundwater and therefore is subject to atmospheric conditions. Confined groundwater occurs only locally as evidenced by the existence of isolated springs and a few exploratory drill holes which encountered artesian conditions. These examples do not reflect general conditions covering large areas but merely represent isolated local strata within the saprolite soil which contain water under a semi-perched condition and or permeable strata overlain by impermeable clay lenses which have been breached by erosion at its exit and recharged short distances upslope by vertical percolation

The plant area is on a moderately sloping, northwest trending topographic ridge which forms a drainage divide between the Little and Keowee Rivers located approximately 0.5 mile to the west and east, respectively. Groundwater levels at the plant site, measured during the 1966 drilling program and subsequently in four piezometer holes drilled for pre-construction monitoring purposes, ranged from elevation 792 ft. (msl) to 696 ft. (msl). The slope of this apparently free water surface is predominantly southeasterly toward the Keowee River and its tributary drainage channels. An average hydraulic gradient

to the southeast of approximately 8.0 percent was plotted along a line of measured wells. This closely conforms to the existing topography as expected. Refer to Figure 2-8 on page 2-39 for measured water levels and typical water table profile.

2.4.7.3 Groundwater Quality

The surface water and groundwater of the area is generally of good quality (Reference 2 on page 2-87). Of the wells surveyed, none were noted where water treatment is being conducted. Temperature of well water measured ranged from a low of 46 to a high of 59 degrees. The majority of readings were from 50 to 53 degrees Fahrenheit.

Water contains different kind amounts of mineral constituents. Temperature, pressure and length of time water is in contact constituents present. Be safe imate contact with the host rocks for longer periods of time, they have a n incent a. I mineral content than surface waters. The mineral content of natural surface safe imate contact time caused by rapid runoff in the mountainous areas.

Tabulated below are the surface v.e., the sample state of the parts per million from the Keowee River near Jocassee, South Carolina. The water sample state and analyzed by the U.S. Geological Survey, Water Resources Division in June 1965.

7.8	Carbonate (CO ₃)	0.0
0.01	Bicarbonate (HCC ₃)	7.0
1.0	Sulfate (SO ₄)	1.0
0.1	Chloride (Cl)	0.6
1.2	Fluoride (F)	0.1
0.4	Nitrate (NO ₃)	0.1
15.0	Phosphate (PO.)	0.0
3.0		
6.6		
13.0		
	7.8 0.01 1.0 0.1 1.2 0.4 15.0 3.0 6.6 13.0	7.8 Carbonate (CO3) 0.01 Bicarbonate (HCO3) 1.0 Sulfate (SO4) 0.1 Chloride (Cl) 1.2 Fluoride (F) 0.4 Nitrate (NO3) 15.0 Phosphate (PO4) 3.0 6.6 13.0 13.0

Soil surveys conducted by the U.S. Department of Agriculture in cooperation with the South Carolina Agricultural Experiment Station assign pH values of between 5.0 and 6.0 for the Hayesville and Cecil soil series which are present at the site area (Reference 3 on page 2-87). Surface water samples taken from the Keowee River within one mile of the site have a pH of 6.5 to 7.0. It is expected goundwater at the site has a pH ranging between 5.5 and 6.0.

The cation exchange potential can be evaluated by knowing the SAR (Sodium Absorption Ratio), sat ition extract values, and the pH of the soil. Two samples of sappolite soil were obtained from drill holes used in determining field permeability values and tested for Sodium Absorption Ratio (SAR). The results are tabulated as follows:



2-33
2.4 Hydrologic Engineering

Saturation Extract Values Milligram-equivalent per 100 grains of soil

Sample No.	pH	Cond. (mhos)	Calcium	Magnesium	Sodium	SAR
1	5.8	5	0.015	0.000	0.0108	0.122
2	5.7	7	0.010	0.000	0.6166	0.235

Considering the amount of soil that is available is so great, it is evident that many times the amount of strontium and or cesium contained in the waste could be absorbed. Further, the distribution coefficient for ion exchange of radionuclides with the sediments is dependent on the pH of the water in the formation (Reference 4 on page 2-87). The distribution coefficient is a ratio of the reaction of these radionuclides that are absorbed on the soil and the fraction remaining in solution. It is expected that the soils surrounding Oconec have a ratio in the range of 80 to 150, and consequently a sub-tantially lower average velocity for any radionuclide to that of natural water will result.

The estimated maximum rate of movement of water through the soils is about 0.75 feet per day. Using this rate in relation with the above distribution coefficient, bulk density and porosity of the soil, and ratio of the weight of soil to volume of groundwater it indicates the radionuclide velocity will be about .0015 that of groundwater. Using a safety factor of five for variance in flow and competition for exchangeable sodium ions, it would require more than 1000 years for strontium or cesium ions to migrate a distance of one-half mile. In summary, the movement would be so extremely slow that the saprolite soil is an effective natural barrier to the migration of radionuclides.

2.4.7.4 Program of Investigation

Permeability tests were performed in borings in the late 60's as part of the original site investigation program to determine permeabilities of the soil underlying the site. The tests were run according to the Bureau of Reclamations Field Permeability Tests. Designation E-19. Figure 2-9 on page 2-40 shows the arrangement of the field test equipment along with a brief description of the procedure used in determining the soil permeability test results. Test results are from 5 borings as presented in Table 2-4 on page 2-37. The formulae used in the calculations of the k values are shown in Figure 2-10 on page 2-41. Field permeability tests conducted within the saprolite soil yielded values ranging from 100 to 250 ft. yr. The permeability tests were performed in holes of varying depths to determine, if the zoned typed weathering of the saprolite soil affects vertical permeability. Based on the test results, inspection of nearby road cuts, and a study of the exploratory drill logs, it is concluded that the surficial saprolite possesses lower permeability values than that found in the deeper strata. This correlates with the general profile of the saprolite in that the later stages of weathering produce a soil having a higher clay content than the more coarse-grained silty sand sediments below. This natural process of weathering results in the formation of a partial barrier to downward move ment of the surface water.

2.4.7.5 Groundwater Conditions Due to Keowee Reservoir

As previously discussed, the groundwater levels at the plant range from elevation 792 ft. (msl) to below elevation 696 ft. (msl). The Keowee Reservoir operates with a maximum pool elevation of 800 ft. (msl). This results in raising the surface water elevation to that datum on the northern and western portions of land adjoining Oconee. It also raises the existing groundwater table for those local areas bordering the reservoir where formerly the groundwater surface was below elevation 800.0 ft (msl). The reservoir materially contributes in establishing a potentially larger recharge area and where it effects the groundwater results in a more stable hydraulic gradient with less seasonal iluctuation than formerly existed.



Preliminary studies indicate that Keowee Reservoir has created the following groundwater conditions at Oconee.

- 1. Groundwater continues to migrate downslope through the saprolite soil on a slightly steeper gradient in a southeasterly direction toward the Keowee River base datum.
- 2. There are two topographic divides which separate the nuclear station from the nearby reservoir: (1) a one-half mile wide north-south stretch of terrain west of the site, and (2) a narrow 500 ft, wide ridge north of the site. Original groundwater measurements in drill hole K-12, located atop the northern ridge, show water table conditions exist at about elevation 810 ft. (msl)
- 3. There should be no reversal of groundwater movement at the site, and all water percolates downward and away from the plant area.
- The construction of Keowee Dam and Reservoir has not created adverse groundwater conditions at the plant site.
- 5. Infiltration of domestic wells, located bey id the plant one-mile exclusion radius, by surface water from the site is not possible under the groundwater conditions imposed by Keowee Reservoir.

2.4 Hydrologic Engineering

2.4.8 TABLES

		ė	8	in,	6	
4						ķ
8						
18						
	8				ş	ŗ

Acil to.	4 €	• (i)	4 -	1. (ft)	Q (ft'/min)	- 2	W.T Condition	
VA 4W2	1.8.1	2.50	1:531	27.0	0.175	21.5	1.0w	
CWLINW2	14.0	0.833	16.8	0.15	0.133	20.5	High	
13W1	51 L	0 8 7 3	7.422	27.0	0.275	20.03	Low	
1 MST-VD	14.0	61.8.3.3	2 6v. N	11.1	0.240	20.5	High	
Z ANSE VA	12.25	0.833	14.7	3.03	0.1.0	21.19	High	

 $\frac{h}{r}$ < \sim 10, not acceptable

2. $\frac{h}{r}$ < 10, possibly acceptable

3. I or manual incremental test, k = 7.4 x 10° ft.min.

Oconee ISFSI Safety Analysis Report

x 10.*

x 10, x x 10, x x 10, x 2.4 Hydrologic Engineering

2.4 Hydrologic Engineering

2.4.9 FIGURES





0

2.4 Hydrologic Engineering

1



Figure 2-8. Groundwater Survey at Station Site



2-39



2.4 Hydrologic Engineering



Figure 2-9. Well Permeameter Test Apparatus



2.4 Hydrologic Engineering



Figure 2-10. Formulae For Determining Permeability



2-41

2.4 Hydrologic Engineering

.

Oconee ISFSI Safety Analysis Report

2.5 GEOLOGY AND SEISMOLOGY

Specific soil testing has been performed at the designated location for the ISFSI. The data obtained from this testing is utilized in the foundation design of the ISFSI (See Section 2.5.5, "ISFSI Foundation" on page 2-51). It should be noted that foundation conditions at the ISFSI site are typical of those encountered in the general station area. The following sections discuss the Oconee site geology and seismology.

2.5.1 BASIC GEOLOGIC AND SEISMIC INFORMATION

Geologic and seismic investigative studies for Oconee Nuclear Station include the following:

- 1. a review of the available geological and seismological literature pertaining to the region:
- 2 a geological reconnaissance of the site, performed primarily for the purpose of evaluating the possibility of active faulting in the area;
- geophysical explorations and laboratory tests to provide parameters for evaluating the response of foundation materials to earthquake ground motion.
- 4. an evaluation of the seismic history to aid in the selection of the design earthquake that the station might experience, and
- 5. the development and recommendation of seismic design parameters for the proposed structures.

The geologic field work at the site was performed concurrently with the drilling for the original plant site. The site reconnaissance is a continuation of the geologic field work done for the Keowee Dam. Local outcrops, though scarce, are examined and the rock types, joint and foliation orientation noted.

The original plant structures are founded on normal Piedmont granite gneisses. The construction characteristics of the residual soils overlying the rock that form the foundation for the ISFSI are known and present no problems in design or construction. The rock underlying the site, below surface weathering, is hard and structurally sound and contains no defects which would influence the design of heavy structures.

The southeastern Piedmont rocks are highly stable seismologically, and the Oconee Nuclear Site should be one of the nation's most inactive areas with respect to earthquake activity.

2.5.1.1 Regional Geology

The regional structure is typical of the southern Piedmont and Blue Ridge. The region was subjected to compression in the northwest-southeast direction which produced a complex assortment of more or less parallel folds whose axes lie in a northeast-southwest direction. The Blue Ridge uplift was the climax of the folding, and it was accompanied by major faulting, along a line stretching northeast through Atlanta and Gainesville. Georgia and across South Carolina, 11 miles northwest of the site. This has been termed the Brevard Fault.

The age of these uplifts has not been agreed on by geologists. The concensus of geologic opinion seens to require a period of severe deformation followed by at least one additional period of less severity. Probably all occurred during the Paleozoic Era, but it has been suggested that the last major uplift was as late as the Triassic (180 million years ago) when the Coastal Plain to the east was downwarped. A number of

investigators have maintained that the major detormative movements occurred at least 225 million years ago. However, all the resulting stresses have not yet been fully dissipated.

There is no evidence of any displacement along these faults during either historic times or during the Geologic Recent Era as indicated in displacements in the residual soils that blanket the region. While the well known Brevard Fault passes 11 miles northwest of the site, there is no indication of a major fault in the immediate vicinity of the site. Furthermore, the major faults of the region are ancient and dormant, except for minor adjustments at considerable depth. Therefore, there is no indication of any structural hazard to foundations.

The site is underlain by crystalline rocks which are a part of the southeastern Piedmont physiographic province. This northeastward - trending belt of ancient metamorphic rocks extends northward from Alabama east of the Appalachians, and in South Carolina crosses the state from the Fall Line on the east to the Blue Ridge and Appalachian Mountains on the west. These rocks are generally recognized as being divided into four northeast-southwest trending belts in the Carolinas. From southeast to northwest they are the Carolina slate belt. Charlotte belt. Kings Mountain belt, and Inner Piedmont belt. The Oconee Nuclear Site is in the western, or Inner Piedmont Belt.

The Piedmont metamorphic rocks of the site were formed under many different combinations of pressure and temperature, and represent a complex succession of geologic events. The formerly accepted concept that the Piedmont consists only of the deep, worn-down roots of ancient mountains now seems untenable. The older theory that the rocks were exclusively of igneous origin is being replaced by the proposition that they represent highly metamorphosed sediments which have been folded, faulted, and injected to result in one of the most complex geologic environments in the world. It can be said with certainty, however, that these rocks represent some of the oldest on the continent. The new techniques of dating by radioactive decay have placed the age of the metamorphic episodes that produced these rocks as occurring from 1.100 my (million years) to 260 my ago. The successive northeastward trending bands of rocks vary greatly in lithology from granitic types to highly basic classifications, with gneisses and schists being the predominant classifications petrographically. In summary, the regional geology of the Oconee Nuclear Site can be accepted as typical of the southeastern Piedmont - narrow belts of metamorphic rocks trending northeast, with the foliation dipping generally to the southeast.

2.5.1.2 Site Geology

2.5.1.2.1 Geologic, History, Physiography, and Lithography

The rock present at this site is metamorphic. It is believed to be Precambrian in age: thus, it was formed over 600 million years ago. The complete history of this region is quite complex and has not been fully unravelled. However, it is the concensus of the geologic opinion that the formation consisted of thick strata of sedimentary rocks which were later downwarped and altered by heat and pressure. This first rock formed is termed the country rock.

More than one episode of regional metamorphism transformed the rock into metasediments with accompanying injection and mobilization by plastic flow.

Since the formation of the country rock, most of the mass has been altered or replaced by injection of granite gneiss, biotite hornblende gneiss, and one or possibly more pegnatite dikes.

It is not definite which is the younger: the granite gneiss injection or the bioute hornblende gneiss injection. The limited evidence points to the granite gneiss as the younger of the two



The pegmatite dikes are the youngest rock known at this site. One such dike is exposed in the road cut on the east side of the state highway passing through the site. It clearly shows the pegmatite cutting through the older rocks, and thus, demonstrates that it is the youngest.

Regional metamorphism, folding, and some minor faulting occurred concurrently much of this early time.

This site is located within the Inner Piedmont Belt, at this locality the westernmost component of the Piedmont Physiographic Province. The topography of the area is undulating to rolling, the surface elevations ranging from about 700 ft. to 900 ft. The region is moderately well dissected with rounded hilltops, representing a mature regional development. The area is well drained by several intermittent streams flowing away from the center of the site in a radial pattern.

The local geology of the Oconec Nuclear Site is typical of the southeastern Inner Piedmont Belt. The foundation rock is biotite and hornblende gneiss striking generally northeast, with the foliation dipping southeast. The rock is overlain by residual soils, which vary from silty clays at the surface, where the rock decomposition has completed its cycle, to partially weathered rock, and finally to sound rock.

The strike of the foliation planes or bands of mineral segregation is north 6 degrees to 15 degrees east with an average dip of 22 degrees to 28 degrees to the southeast. However, due to the local folding or warping at this site, minor variations in the strike and dip of the foliation will occur within the site.

There have been periods of erosion and perhaps even continuous erosion since the close of the Paleozoic Era. The rock now encountered at this site represents the deeper portions of the original metamorphic complex.

The rock encountered at this site is of three main types; light to medium gray granite gneiss, light gray to black biotite hornblende gneiss and white quartz pegmatite with local concentrations of mica, both muscovite and biotite varieties.

The dominate rock type at this site is the light to medium gray granite gneiss. This rock type is generally moderately hard and hard below the initial soft layers encountered in the rock surface. Joints in this rock are brown iron stained in the upper softer layers, but in the deeper harder rock, the joints are not stained. This helps illustrate that the jointing at this site does not control the weathering or decomposition of the rock.

The second most abundant rock type is the biotite hornblende gneiss. The rock is generally weathered or softer to a greater depth than the granite gneiss. This is probably due to the higher percentage of biotite mica. Biotite mica is a potassium magnesium-iron aluminum silicate. The iron content of the biotite mica causes the rate of decomposition to accelerate. However, generally at the deeper portions of the original plant borings, the biotite hornblende gneiss hardness increases to moderately hard or harder. Only a few thin soft layers were noted in this rock in the deeper portion of the original plant borings but not in the ISFSI site boring logs which are presented and discussed in Section 2.5.4, "Subsurface Materials" on page 2-50.

A few layers of hard quartz pegmatite with local concentrations of mica were recorded. The thickness of the pegmatite layers are generally less than three feet. These pegmatite layers are dikes. A dike is a sheetlike body of igneous rock that fills a fissure in the older rock which is encountered while in a molten condition. There is an exposure of mica-quartz pegmatite dike on the east side of the state road cut passing through this project. This dike exposure is about 3.5 ft, wide, but due to the lack of knowledge of orientation of the dike, the exact width cannot be computed. The quartz pegmatite encountered in the original station borings probably represent other smaller dikes of the same material. These dikes are of



hard, sound and durable material and should cause no concern to construction or foundation requirements.

2.5.1.2.2 Rock Weathering

Rock weathering at the Oconee Nuclear Site is about normal for Piedmont biotite gneisses. The range of depth before sound rock is reached is 0 to 35 ft. for the ISFSI foundation. Yard grade is nominally at elevation 825.0 msl, with the bottom of the foundation at elevation 822.0 msl. The resulting residual materials - clays, silts, and weathered rock - are structurally strong, and are used in situ for the foundation of this structure.

2.5.1.2.3 Jointing

The rock at the Oconee site is moderately jointed. All of the visible rock outcrops were studied in attempting to determine the correct orientation of the joint patterns.

Some moderately good rock outcrops were found and several joint pattern orientations measured. While studying and logging the original site rock cores, all of the joint dips were recorded. The dips of the joint patterns recorded in the rock cores were associated with the dips measured in the rock outcrops.

The rock has apparently not been subjected to stresses causing high concentrations of joints. The core borings indicate that jointing is widely spaced, and has not influenced the weathering pattern. Joints are about equally divided between strike and dip joints, with occasional oblique joints.

2.5.1.2.4 Ground Water

Subsurface water is typical of Piedmont area. The top of the zone of saturation, or water table, follows the topography, but is deeper in the uplands and more shallow in valley bottoms. It migrates through the pores of the weathered rock, where the feldspars have disintegrated and left intersticial spaces between the quartz grains. Additional water is contained in the deeper fractures and joints below the sound rock line. The water table is not stationary, but fluctuates continually as a reflection seasonal precipitation. Additional information on ground water is included in Section 2.4.7, "Subsurface Hydrology" on pag. 2-31. Groundwater elevations encountered during the ISFS1 site borings are noted on the boring logs. Section 2.5.4. "Subsurface Materials" on page 2-50.

2.5.2 VIBRATORY GROUND MOTION

A seismological study for the Oconee Nuclear Site has been performed to determine the design and hypothetical earthquakes for the site and the ground motion associated with them. Details are discussed in Section 2.5.2, "Vibratory Ground Motion" on page 2-214 of Reference 5 on page 2-87.

2.5.2.1 Earthquake History

The largest earthquakes close to the site occurred near Charleston in August. 1886, some 200 miles from the site. Two shocks occurring closely in time, had an intensity estimated to be about Modified Mercalli IX at the epicenter and were perceptible over an area of greater than two million square miles.

Aftershocks of the main earthquake had intensities ranging up to Modified Mercalli VII. These shocks may be associated with a downfaulted Triassic basin under the coastal plain.

There have been two moderate earthquakes in the immediate vicinity of the plant since construction began.



In 1971, an earthquake occurred near Seneca. South Carolina. The descriptions of this event which occurred at 07:42 (EST) on July 13, 1971 have been examined from various sources. A MM intensity VI was assigned to the event by USGS based primarily on the report of a cracked chimney near Newry, about 10 km south of the present epicentral area. A detailed examination of the buildings and chimneys by Sowers and Fogle (1978) convinced them that the chimney in question had been broken and in a state of disrepair before the shock. They assigned an intensity IV (MM) to the shaking at Newry.

The July 13, 1971 event at 07.4 J AM EDT was preceded by a felt shock at about 4:15 AM EDT and followed by at least one felt aftershock at 7:45 AM (Sowers and Fogle, 1978).

On August 25, 1979 (9.31 PM EDST, Aug. 26) a magnitude 3.7 earthquake occurred in the vicinity of Lake Jocassee, South Carolina. This MM intensity VI event was felt in an area of about 15,000 sq. km and was recorded locally on the three statish—ike Jocassee seismographic network, and regionally on seismic stations in South Carolina. North Carolina, Georgia, Tennessee, and Virginia. During the period (August 26, 1979 - September 15, 1979) 26 aftershocks were recorded and they ranged in magnitude from +.60 to 2.0.

A list of earthquakes in the region is provided in Table 2-5 on page 2-55.

2.5.2.2 Geologic Structures and Tectonic Activity

The region (defined as North Carolina and South Carolina, and parts of Georgia, Alabama, Tennessee, and Virginia) is comprised of three large northeast-southwest trending tectonic zones: The coastal plain, the crystalline-metamorphic zone and the overthrust zone.

The site is located nearly in the center of the crystalline-metamorphic zone, which consists of six generally recognized metamorphic belts. From southeast to northwest these are. The Carolina slate belt, Charlotte belt, Kings Mountain belt, Inner Fiedmont belt, Brevard belt, and Blue Ridge belt. The site location is within the Inner Piedmont belt. The rocks in the belts consist of metamorphosed sediments and volcanics that have been folded, faulted, and intruded with igneous rocks. These belts are delineated by differing degrees of metamorphism. Generally, the degree of metamorphism becomes progressively less from the northwest to the southeast.

The oldest meiamorphic rocks are located in the Blue Ridge belt. The more easterly belts of younger rocks have undergone progressively less metamorphism

To the north and west are found a series of fault systems. Since these faults are both numerous and extensive, they can be grouped together and referred to as the overthrust zone. These faults no doubt resulted from the formation of the Appalachians

The great system of thrust faults in the overthrust zone and most of the known faulting within the crystalline-metamorphic zone apparently occurred during the last period of metamorphism (260 million years ago).

During the Triassic Period (180 to 225 million years ago), sediments were deposited over parts of the exposed metamorphic belts. These deposits and the older metamorphics were intruded by a system of northwest-trending diabase dikes and were faulted by northeast-trending normal faults in the late "riassic Time (200 million years ago). Some of the older faults within the crystalline-metamorphic zone riasy have been active of this time.



2-47

From the late Triassic time until the present, the coastal plain has accumulated a sedimentary cover over its crystalline-metamorphic bedrock. These sediments overlap the bedrock and thicken toward the outheast, effectively masking any ancient faulting.

It is considered possible that igneous activity has occurred in the region after the Triassic because volcanic bentonitic clays of Eocene (approximately 50 million years ago) and possible Miocene age (12 million years ago) have been mapped in the sediments of the coastal plain in South Carolina. The source of this volcanic activity is presently unknown.

Faulting: The names, distances and directions from the proposed site, and the probable age of the known faulting in the region are as follows:

Name	Distance-Direction Form Site	Probable Age Millions of Years
Brevard Fault	11 Miles NW	260
Dahlonega Fault	40 Miles W	260
Whitestone Fault	47 Miles NW	260
Towaliga Fault	96 Miles S	260
Cartersville Fault	104 Miles W	260
Gold Hill Fault	115 Miles E	260
Goat Rock Fault	140 Miles 8W	260
Triassic, Deep River Basin, N.C. and S.C.	140 Miles E	200
Triassic, Danville Basin, N.C.	145 Miles NE	200
Crisp and Dooly Counties. Ga.	190 Miles SW	12 to 70
Probable Triassic Basin Charleston, S.C.	200 Miles SE	200

The first seven faults are all associated with the last metamorphic period. The Brevard, Whitestone, Dahlonega, and Cartersville faults apparently form an interrelated system. This system separates the eastern metamorphic belts from the Blue Ridge metamorphic belt and the overthrust zone on the west.

The Towaliga Goat Rock, and Gold Hill Faults, and the Kings Mountain belt apparently form another interrelated alignment within the eastern metamorphic belts. The Kings Mountain belt is not considered a fault. Its association and alignment in relation to the three known faults mentioned and the location of earthquake epicenters within the area bounded by these features, lead to the conclusion that these features f^{α}_{α} is an interrelated alignment.

There is no surface indication that any of these three faults have been active since the Triassic Period (200 million years).

Two fault locations in the region have been thoroughly investigated by borings. These are the Cartersville fault near the Allatoona Dam, and the Oconee-Conasauga fault in Georgia. These faults were found to be completely healed and not to have moved in many millions of years.

The Triassic basins of the Carolinas and further north may be due to the release of the compressional forces which formed the Appalachians These basins are down-faulted grabens which are filled with

Triassic sediments. Two earthquakes in the vicinity of McBee, South Carolina, may be related to an extension of a Triassic basin which has been inferred in the Chesterfield-Durham area.

The earthquake activity near Charleston. South Carolina, may indicate an active fault in that region. However, no evidence of surface faulting has been found.

2.5.2.C Correlation of Earthquake Activity with Geologic Structures or Tectonic Provinces

The region surrounding the Oconec Station site can be divided into three major areas on the basis of the regional tectonics and the seismic history. These major seismic areas are:

- 1. the overthrust zone and Blue Ridge metamorphic belt:
- 2. the crystalline-metamorphic zone, exclusive of the Blue Ridge belt, and
- 3. the coastal plain

The greatest number of recorded shocks have occurred within the overthrust zone and the Blue Ridge metamorphic belt northwest of the Brevard. Whitestone. Dahlonega, and Cartersville fault system. The epicenters in this area are generally widely scattered.

There have been a small number of earthquakes within the crystalline-metamorphic zone, exclusive of the Blue Ridge metamorphic belt. These earthquakes, extending from central Georgia to North Carolina, may be associated with the Towaliga, Goat Rock, Gold Hill, Kings Mountain alignment.

The coastal plain has experienced few earthquakes outside of the Charleston area. Four shocks, at Wilmington, North Carolina and Savannah. Georgia, have occurred but are unrelated to any known faulting, although the Wilmington shocks were adjacent to the Cape Fear Arch.

The only earthquake which does not closely fit this system of seismic areas is the 1924 shock in Pickens County. South Carolina (MM V Intensity). However, it is likely that this earthquake is associated with the overthrust-Blue Ridge seismic area.

2.5.2.4 Maximum Earthquake Potential

The assignment of probable future earthquake activity can only be based upon the previous record and the known geology of the area. Although the seismic history of the region is fairly short, a reasonable picture of the seismicity of the area becomes apparent from a study of the epicenter locations and the regional tectonics.

There are three significant zones of seismic activity in the general vicinity of the site; the Brevard and related faults zone, the overthrust zone, and the Towaliga, Goat Rock, Gold Hill, Kings Mountain alignment.

An evaluation of the earthquake activity and the regional geology can result in the selection of a series of maximum-sized shocks which are likely to occur in these variou, areas. Conservatively, we can assume that the previous maximum-sized shock on a particular fault zone can occur during the economic life of the power station and ISFSI at perhaps the nearest approach of the particular fault system to the site.



Oconec ISFSI Safety Analysis Report

Zone	Location	(MM) intensity at Epicenter	Estimated Magnitude (Richter)
Brevard Fault Zone	11 Miles NW	VI	Less than 41, 10 5
Overthrust	75 Miles NW	VIII	Less than 51/2 to 6
Towaliga. Goat Rock Gold Hill. Kings Mountain Alignment	30 Miles SE	VII-VIII	Less than 5½ to 6

2.5.2.5 Seismic Wave Transmission Characteristics of the Site

Static and dynamic engineering properties of the soil and rock materials that underlie the general plausite area are discussed in Section 2.5.4. "Stability of Subsurface Materials and Foundations" on page 2-12. of Reference 5 on page 2-87. Design response spectra that include considerations of the thickness and distribution of these materials are discussed in Section 2.5.2.8. "Design Response Spectra" on page 2-219 of Reference 5 on page 2-87.

2.5.2.6 Maximum Hypothetical Earthquake (MHE)

The MHE acceleration value is 0.15 g for structures founded on overburden. The design response spectra are covered in Section 2.5.2.8, "Design Response Spectra."

2.5.2.7 Design Base Earthquake

It is considered likely that the shocks listed in Section 2.5.2.4. "Maximum Earthquake Potential" on page 2-49 could occur no closer than the indicated distances from the site during the life of the planned facilities. Since the magnitudes of these shocks are fairly small, the distance from the epicenter becomes extremely important. Ground accelerations would diminish rapidly with the distance from the epicenter. Although larger earthquakes occur within other fault zones, the highest ground accelerations at the site would be experienced from an earthquake along the Brevard fault zone. The assumption of a shock of less than Richter Magnitude five occurring along the Brevard fault zone at its closest location to the site (11 miles), would give ground motions on the order of five percent of gravity at the site. Vertical ground accelerations, as contrasted to the horizontal accelerations, would be only slightly less than five percent of the gravity in the competent rock at the site.

2.5.2.8 Desig Response Spectra

The Oconee FSAR provides that the maximum ground acceleration for structures founded on over burden (MHE) is .15g (Reference 5 on page 2-87, Section 2.5.2, "Vibratory Ground Motion" on page 2-214 and Figure 2-51 on page 2-244). The accelerations considered and used for the design of the NUHOMS-24P system envelope the MHE acceleration (Reference 6 on page 2-87).

2.5.3 SURFACE FAULTING

This information is discussed in Sections 2.5.), "Basic Geologic and Seismic Information" on page 2-43 and 2.5.2, "Vibratory Ground Motion" on page 2-46.

2.5.4 SUBSURFACE MATERIALS

2.5.4.1 Exploration

A grid pattern of borings was established to provide the maximum amount of submation for determining the foundation and soil conditions and permit flexibility in final ISFSI layout, alignment, and elevation.

The general site area is shown on Figure 2-11 on page 2-60 and the site and boring layout is shown on the Boring Plan. Figure 2-12 on page 2-61.

The drilling, sampling, and rock coring were performed in accordance with methods specified by the American Society for Testing and Materials

"Penetration 3 sting and Split Barrel Sampling of Soils" - D-1586-64T

"Diamond Core Drilling for Site Investigation" + D-2311-62T

"Thin Walled Tube Sampling of Soils" + D-1567-63T

Boring logs are given in Figure 2-13 on page 2-62 throug. Figure 2-26 on page 2-85.

2.5.4.2 Groundwater Conditions

Section 2.4.7, "Subsurface Hydrology" on page 2-31 provides a discussion of the existing groundwater conditions at the Oconee site.

It is anticipated a the removal of the overburden due to ISFSI construction will nave little, if any, effect is the water table. If the water table elevation does change, it is anticipated that it will drop slightly. The sent elevation of the water table at the ISFSI site varies from elevation 797 feet at the south end to ation 822 feet at the north end.

h, drastatic uplift will not occur during the life of the ISFSI because the foundation of the HSMs and associated pavement is at or above the water table. There may be some seepage through the cut into the hillside, however, adequate drainage is provided around the ISFSI site to carry away seepage.

At the south end of the ISFSI site, the elevation of the water table is far below the foundation of the HSMs. At the north end, the foundation of the HSMs will be near the water table elevation. However, the HSM structure at the north end of the ISFSI site is partially founded on rock. Therefore, there will be no reduction of shear resistance due to potential seepage along that bedding

5.5 ISFSI FOUNDATION

A specific soil testing (results and locations presented in Section 2.5.4. "Subsurface Materials" on $pa_{ge} = 2.50$) and foundation evaluation has been performed at the ISFSI site to assist in the development of the insitu static soil bearing pressure. Fourteen (14) soil borings were taken in and around the ISFSI site. The location of these borings is shown in Figure 2-12 on page 2-61. A line of boring was taken along the length of the future foundation of the HSMs. From these borings several undisturbed samples were taken. Several tests, including the triaxial shear test, were performed on selected undisturbed samples. The results from the triaxial shear test provided essential information used to determine the ultimate and allowable bearing capacity. (The triaxial shear tests were performed in accordance with the Corps of Engineers Manual EM10-2-1906, Appendix 10).

After inspection of the boring logs, soil samples, and tests, the worst case soil data were selected and used in the Meyerhoff bearing capacity equation to determine the ultimate soil bearing capacity, which is approximately 12.0 kips square foot. To obtain the allowable static soil bearing capacity, a factor of

safety of 3.0 was applied to the ultimate capacity, which yields the allowable bearing pressure of 4.0 kips square foot (Reference 9 on page 2-87).

The largest applied static bearing pressure was calculated by first determining the dead weight of the HSM with a fully loaded DSC and then dividing by the area of the foundation. This maximum applied static bearing pressure was computed to be 3.3 kips square foot, which is less than the allowable soil bearing pressure of 4.0 kips square foot.

As shown by the boring logs, the HSM foundation will to a large degree rest entirely on either firm soil or partially weathered rock with penetration blow counts ranging from n = 12 to refusal. A conservative analysis was performed to determine the worst-case settlement of an HSM array. Both a 2x3 array and a 2x10 HSM array were considered. This analysis indicates that the worst-case differential settlement will cause the 2x10 HSM array to experience a differential settlement of about 3.0 inches along the North-South axis. Differential settlement in the Last-West direction will be negligible.

These settlements are accounted for in the foundation design. The foundation was analyzed as a finite module using the computer code McAuto STRUDI. (Reference 7 on page 2-87).

This computer code models settlements by the use of calculated soil springs which provide consideration for the settlements. Considering the small magnitude of this settlement, the integrity and radiological shielding of the HSM will not be adversely impacted. The foundation structure consists of a 3 ft. reinforced concrete mat. Typical Hold reinforcement is shown in Figure 8.1-9 of Reference 6 on page 2-87.

The limiting calculated maximum stresses and allowable stresses for loadings as defined by Reference 6 on page 2-8" envelope the site foundation stresses for the Oconec ISESI site. These forces are for the accident condition assuming blocked vents and bound all other loading combinations.

2.5.6 LIQUEFACTION

Potential liquefaction of soils under the Oconee ISFSI foundation area is not a concern because all of the foundation materials are non-liquefiable. The three foot thick concrete mat bears entirely on either firm soil or partially weathered rock having Standard Penetration Test blowcounts ranging from N = 12 to refusal. Figure 2-11 on page 2-60 shows the longitudinal profile of the ISFSI foundation level in relation to both the original ground and to partially weathered rock, based on site borings.

2.5.7 SLOPE STABILITY

The ISFSI site includes cut slopes along both sides of the ISFSI site access road, and along the west, north, and northeastern sides of the ISFSI site as shown in Figure 2-11 on page 2-60. Fill slopes are located along the southeastern and south sides of the ISFSI site. The maximum vertical cut is approximately fifty feet and the maximum vertical fill is approximately ten feet. The maximum ISFSI slope is two horizontally to one vertically.

The stability of slopes associated with the ISFSI site were modeled by a program that utilizes the circular arc analysis method of slices. The program postulates a failure arc through the soil embankment or foundation, computes the soil mass driving moment and the soil mass resisting moments associated with the postulated failure arc, and men determines the resulting safety factor by dividing the total resisting moment by the total driving moment. The computer program allows the compute ion of a large number of safety factors associated with many postulated failure arcs (Reference 8 on page 2-87).

The slope stability analyses were performed using the maximum ISFSI site slope of two horizontal to one vertical. Actual site soil engineering parameters, based on laboratory testing of soil samples, were determined. (Reference the site boring records presented in Figure 2-13 on page 2-62 through Figure 2-26 on page 2-85.) The Seismic Design Input Criteria specified in Section 3.2.3.1. "Input Criteria" on page 3-10 were used as input in determining the seismic behavior of the ISFSI site slopes.

The inimimum safety factors calculated for any postulated failure arc of the vertical cut and fill slopes of the ISFSI site are as follows:

Slope Loading Condition	Minimum Calculated Safety Factor	
55 feet vertical cut slope, static	1.62	
55 feet vertical cut slope, dynamic	1.22	
10 feet vertical fill slope, static	2.06	
10 feet vortical fill slope, dynamic	2.03	

Therefore, the stability of the ISFSI site slopes is ensured since the minimum safety factor is greater than 1.0 for all slopes for all analyzed conditions.

2.5 Geology and Seismology

2.5.8 TABLES

		-	
	2		÷
趨			
125			201

	-	
	10	
	τ.	
	-a -	
	10	
	52	
	æ.,	
	(78) H	
	100	
	1	
	X	
	100	
	10.	
	100	
	127	
	24	
	20	
	22	
	81	
	100	
	100	
	100	
	8	
	2	
	100	
	22	
	3	
	10	
	ALC: N	
	· .	
	20	
	- TE -	
	100	
	100	
	1007	
	-	
	72.	
	24	
	32	
	2	
	Mit .	
	- 22	
	8	
	. R . 1	
	1	
	1.00	
	22	
	360	
	2	
	- EE .	
	10	
	12	
	-26	
	- RE -	
	-	
	12.	
	18	
2.	46.	
	-	
	-	
	R	
	1	
	-	
	1 2.3	
	mt ? an	
	ant Lat	
	cant La	
	ficant Eau	
	ificant Ea	
	nificant Eau	
the second second	milicant La	
the second second	emilicant La	
the second second second	Significant Ea	
the second se	Semilicant La	
the second second second	. Semilicant La	
A DESCRIPTION OF A DESC	(). Significant La	
the second se	5). Significant Lat	
the second se	S). Significant Eau	
the second from the second second second	if 5). Significant Ear	
the second se	of 5). Significant Eat	
the state of the second s	of 5). Significant Eat	
the state of the second s	I of S). Significant Lat	
the second se	I of S). Significant Eau	
the second se	c 1 of 5). Semificant Eau	
the second	pe I of 5). Significant Eau	
the second	the 1 of 5). Significant Eat	
the second	age I of S). Significant Eau	
the second	Page I of S). Significant Eat	
the second	(Pape 1 of 5). Significant Eat	
「「「」」」」」「「」」」」」」」」」」」」」」」」」」」」」」」」」」	(Page I of 5). Significant Lat	
the summer is a second to be a secon	5 (Page I of S). Significant Eat	
the second	-5 (Page 1 of 5). Significant Eat	
「「「「」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」	2.5 (Pape 1 of 5). Semificant Ea	
the shine where the second state of the second	2.5 (Page 1 of 5). Significant Ea	
「「「「「」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」	2.5 (Page I of 5). Significant Ea	
「「「「」」」「「」」」」」」」」」」」」」」」」」」」」」」」」」」」	r 2-5 (Page 1 of 5). Significant ha	
the second second in the second	le 2.5 (Page 1 of 5). Significant Ea	
「「「「」」」「「」」」」」」」」」」」」」」」」」」」」」」」」」」」	ble 2.5 (Page I of 5). Significant Eat	
「「「」」」「「」」」」」」」」」」」」」」」」」」」」」」」」」」」」」	able 2.5 (Page 1 of 5). Significant Ea	
「「「「」」」「「」」」」」」」」」」」」」」」」」」」」」」」」」」」	Table 2.5 (Page 1 of 5), Significant Eat	

		Intensity		Epicentral Loc	ation	
Vear	Date	(Modified Mercalli)	Locality	NIat.	W.I.ong.	Perceptible Area (Square Miles)
1843	Barnary 4	IIIA	Western Fennessee	2.58	0.05	400,000
18.57	December 19	Not Listed	Charleston, S.C.	12.8	79.8	Not Listed
15.72	June 17	A	Milledgeville, Ga.	111	83.3	Not I asted
1874	Tchnary 10 April 17	N	McDowell County, N.C.	152	128	Local
1875	Newcmber 1	IA	Northern Georgia	XXX	82.5	25,000
1875	December 22	NB.	Arvenia, Virginia	37.6	78. S	(800) (05
1877	November 10	>	Western N.C. and Eastern Tennessee	5.54	84.0	5,000
1879	December 12	N	Charlotte, N.C.	35.2	50 G	Not Listed
1884	January 18	A	Wilmington, N.C.	34.3	78.05	1 ocal
1885	August 6	A.M.	North Calofina	36.2	81.6	1 ecal
1886.	February 4	A	Alabama	32.8	88.0	1,600
1886	August M.	N-XI	Charleston, S.C.	671	20.63	2,000,000
1886	October 22 October 22	NI NI	Charleston, S.C. Charlestea, S.C.	12.9 12.9	80.0 80.0	30,000
1886s	November 5	IA	Charleston, S.	32.9	80.0	30,000
1889	July 19	IA	Memphis, Tenn	35.2	0.05	Local
1897	April 30	IV-VI	Tennessee and III	Not I isted	Not Listed	Not Listed
1897	December 18	Λ	Ashland, Virginia	217	77.5	7,500
1900	October 31	v	Jacksonville, I la.	30.4	81.7	Local

2.5 Geology and Seismology

2-56



		Intensity		Epicentral Los	cation	
Усяг	Date	(Modified Mercall)	Locality	N.f.at.	W.Long.	(Square Miles)
1002	October 18	v	Southeastern Lenin and Northwestern Ga	35.0	85,3	1,500
1903	January 23	V1	Georgia and S.C.	32.4	81.1	10,000
1964	March 4	V	Eastern Tenn.	35.7	83.5	5,090
1995	January 27-8	VII	Alabama	34	86	250,000
1907	April 19	N .	South Carolina	32.9	80.0	10,000
1011	April 20	v	North Carolina- South Carolina Border	35.2	82.7	600
1912	June 12	VII	Summerville, S.C.	32.9	80.0	35,000
1912	June 20	V	Savannah, Georgia	32	81	Not Listed
1913	January 1	VII VIII	Union County, S.C.	34.7	81.7	43,000
1913	March 28	VH	Eastern Tennessee	36.2	83.7	2,700
[9]3	April 17	V	Eastern Tennessee	35.3	84,2	3,500
1914	January 23	v	Eastern Tennessee	35.6	84.5	Local
1914	March 5	VI	Georgia	<u>tt 5</u>	83.5	50,000
1914	September 22	v	South Cerolina	33.0	80.3	30,000
1915	October 29	v	North Carolina	35.8	52.7	1,290
1916	February 21	VI	Western N.C.	35.5	82.5	200,000
1916	August 26	v	Western N.C.	36	81	3,800
1916	October 18	VII	Alabama	33.5	86.2	100,000
1917	June 29	v	Alabama	32.7	87.5	Local
1918	June 21	v	Tennessee	36.1	84.1	3,000

	-	÷.	į,		
- 482			ь		
				ь	
				53	
				90	

		ð
		1
		2
		G
		"
		Z
		3
		ş
		4
		2
		ą
		4
		ŝ
		Å
		4
		7
		â
		1
		2
		3
		ŝ
		ć
		1
		3
		1
		3
		1
		3
		1
		4
		7
		3
		3
		Ŧ
		1
		à
		-
		3
		j
		h
		1
		4
		ų
		2
		ĥ
		ú
		3
		**
		1
		1
		ş
		1
		M
		۴

-

		Intensity		Epicentral Lo	cation	
Year	Date	(Modified Mercalli)	Locality	N1at.	W.Long.	Perceptible Area (Square Miles)
8161	October 15	A	Western Lennevee	35.2	2.68	20(000)
0261	December 24	N	Lastern Tennessee	\$	85	1 ocal
1924	October 20	N	Pickens County, S.C.	35.0	82.6	56,000
926	July 8	Ы	Southern Mitchell County, N.C.	15.9	82.1	Local
1927	June 16-	A	Alabarca	M7	86.0	2,500
8201	November 2	М	Western N.C.	u m	82.6	46,000
11.61	May 5	IA-A	Northern Alabama	33.7	86.6	6,500
1161	December 19	IV.VI	Summerville, S.C.	0.1.0	80.2	Local
5261	January I	v	North Carolina- Georgia Border	151	81.6	2,000
1939	May 4	>	Anniston, Ma.	33.7	858	Not 1 isted
1941	November 16-	IA:A	Covington, Icnn.	32.6	59.7	Local
1945	June 13	×	Cleveland, Tenn.	38	84.5	Not Listed
1945	July 26	м	Murray Lake, S.C.	34.3	81.4	25,000
2361	November 19	V	Charleston, S.C.	32.8	80-0	Not Listed
1952	July 16	- IA	Dycrsburg, Tenn.	36.2	89.6	Not Listed
1954	Janc , 22	*	Athens and I towah, T.cnnesee	15.3	54.4	Not Listed
1954	April 26	Λ	Memphus, Tenn.	35.2	1.05	Not I isted
1955	January 25	и	Tenn-Arkansas- Missani Boider	35.6	603	30,000

Oconee ISFSI Safety Analysis Report

2.5 Geology and Seismology

	-	
- 14	38V	
- 22		
100		

Oconee ISFSI Safety Analysis Report

Perceptible Area (Square Miles) Not Listed Not Listed Not Listed Vot Listed Not Listed 1,7083 Net Listed Vot Listed R, 300 005'11 8,100 4,100 25,600 Local 400 2,800 4,500 Local 3.500 400 Not 1 isted W.Long. 89.5 5.68 2.63 80.2 835 5.02 89.5 S.4.D. 1.68 84.5 81.5 LLL 8.9 I. 528 5 63 89.6 12 2 2 F piccutral Location Table 2.5 (Page 4 of 5). Significant Earthquakes in the Southeast I nited States (Intensity V or Greater) Not Listed NIM. 34.2 36.2 34.5 34.5 10.50 11.52 32.55 35.6 2.44 1.22 1.15 5.58 32 -\$ 8 12 ŝ Alabama-Tennessee Border Tennessee-Arkansis Border Lasterna Central Tennessee Othon County, Tenn. Dver County, Tenn. Dyer County, Tenn. 1 weality Northern Alabama Fastern Tennessee I emessee Border Wilmington, N.C. Northeastern S.C. Near Coast, S.C. North Carolina-Anderson, S.C. South Carolina Western N.C. Western N.C. Finley, Tenn. Finley, Tenn. Vargenia-N C Finley, Tenn. Border Intensity (Modified Mercalli) IA 2 1A 11/2 17 12 17 10 13 3. 17 5 IN 20 20 2 > 20 20 2 chruary 262 November 24 December 13 Veptember 28 December 21 September 7 September 5 **Actober 26** lamuary 28 October 20 August 12 Jamuary 28 August 3 Manch 29 March 5 May 13 April 23 Figner 23 April 8 Indy 2 Date 0961 1997 1958 1958 656l 6561 1959 65.61 1960 Year 1957 1561 1991 1958 5561 5564 2261 556.8 956 9561 1561

2-58

		Intervity		Epicentral Los	cation	- 111 A
10	Date	(Modified Mercal5i)	Locality	NLst.	W.Long.	(Square Miles)
0%	April 15	N	Eastern Tenn.	2.52	84	000-1
100	Arred 21	A	Lake County, Tenn	36.3	89.5	Incal
	14 14	•	Charleston, S.C.	11	402	Local
10	100 cm (100	IV.VI	Sentera, S.C.	34-35	8.2.83	Local
1	CT MINE	M	Lake Joessere, S.C.	35	83	5,800

2.5 Geology and Seismology

.

2.5.9 FIGURES







Figure 2-12. Site Boring Plan

2-61

Ococce ISFSI Safety Analysis Report

8081	NG	
DE C I	1044 9	1.04
216.31	1.1976-0-1	1.000

1	Description	1 51	IP ELEY	Remarks
.0	Red micaceous silty fine to medium sand		881.80	M + 13
.0	Strong brown micaceous fine to medium sandy silt		876.80	N + 6
10.0	Gray/brown micaceous silty fine/ coarse sand		871.80	N * 11
15.0			866.80	Undisturbed Sample 17.6'-19.5'
0.0	Black/gray micaceous slightly silty tine/coarse sand w/gravel		851.80	N * 12
22.6	Brown micaceous silty fine/ coarse sand Black/reodish brown very micaceous fine to medium sandy silt		859.20	N * 6
25.0	Brown/white micaceous silty fine to coarse sand		856.8	N = 10
30.0	Black/gray micaceous silty fine to coarse sand		851.8	M = 100
	S. CARLES .		10.1	

Figure 2-13 (Part 1 of 2). Core Boring Record

2.5 Geology and Seismology

BORING DESIGNATION 1

35.0 E	Elack light gray micaceous slightly silty fine to coarse sand			846.80	N =100 Undisturbed Sample
40.0		1 1		100	37.6 * 37.8
	Light brown/light gray micaceout slightly silty fine to coarse sand			841.80	N = 49
45.0	Top: Reddish brown micaceous fine sandy silt. Bottom: Light brown to gray (light) micaceous slightly silty fine to coarse sand			836.80	¥ * 100
50.7	Carbice fishtail refusal			831.10	
55.0		42.9	NX		
60.0		84.5	нx	821.80	
60.2	Water Table			821.60	
70.0		98.0	NX	811.80	
19.4	Coring Terminated			802.44	

Figure 2-13 (Part 2 of 2). Core Boring Record

-

2-63



BORING DESIGNATION 2

D.0 Brownish red micaceous silty Fine to coarse sand BB1.62 N = ?9 5.0 Brownish red micaceous fine to medium sandy silt, Black/light gray silty sand at bottom of sample B76.62 N = 10° 10.0 Black strong brown micaceous silty fine to coarse sand B71.62 N = 10° 14.5 Carbide fishtail refusal B67.12 15.0 0.0 NX B66.62 20.0 0.0 NX B66.62 25.0 0.0 N = 17 35.0 Silty fine to medium sand B51.02 35.0 Brown to light gray micaceous silty fine to coarse sand B41.62 40.0 Brown to light gray micaceous silty fine to coarse sand B41.62	1	Description	1 I	Size	Eley	Remarks
5.0 Brownish red micaceous fine to medium sandy silt, Black/light gray silty sand at bottom of sample 876.62 N = 10° 10.0 Black strong brown micaceous silty fine to coarse sand 871.62 N = 100 14.5 Carbide fishtail refusal 867.12 N = 100 15.0 0.0 NX 866.62 20.0 0.0 NX 866.62 25.0 0.0 NX 856.62 30.6 Re-enter Hole w/fishtail Light brown to gray micaceous silty fine to medium sand 851.02 N = 17 35.0 Brown to light gray micaceous silty fine to coarse sand 841.62 N = 100	0.0	Brownish red micaceous silty fine to coarse sand			881.62	N * 29
10.0 Black strong brown micaceous silty fine to coarse sand 871.62 N * 100 14.5 Carbide fishtail refusal 867.12 15.0 0.0 NX 866.62 20.0 0.0 NX 866.62 25.0 0.0 NX 856.62 30.6 Re-enter Hole w/fishtail 851.02 N = 17 35.0 851.62 35.0 861.62 40.0 Brown to light gray micaceous silty fine to coarse sand 841.62 N = 100	6.0	Brownish med micaceous fine to medium sandy silt, Black/light gray silty sand at bottom of sample			876.62	N * 10 ⁴
14.5 Carbide fishtail refusal 867.12 15.0 0.0 NX 866.62 20.0 0.0 NX 861.62 25.0 0.0 NX 856.62 30.6 Re-enter Hole w/fishtail Light brown to gray micaceous silty fine to medium sand 851.02 N = 17 35.0 8:36.6 Undisturbed sample 	10.0	Black strong brown micaceous silty fine to coarse sand			871.62	N * 100
15.0 0.0 NX 866.62 20.0 861.62 25.0 0.0 NX 856.62 30.6 Re-enter Hole w/fishtail Light brown to gray micaceous silty fine to medium sand 0.0 NX 851.02 35.0 8:66.62 Undisturbed sample 38.1' - 39.9' 8:66.62 40.0 Brown to light gray micaceous silty fine to coarse sand 841.62 N = 100	14.5	Carbide fishtail refusal			867.12	
20.0 25.0 0.0 HX 861.62 30.6 Re-enter Hole w/fishtail Light brown to gray micaceous silty fine to medium sand 851.02 N = 17 35.0 8:6.62 Undisturbed sample 38.1' = 39.9' 40.0 Brown to light gray micaceous silty fine to coarse sand 841.62 N = 100	15.0		0.0	NX	866.62	
25.0 0.0 NX 856.62 30.6 Re-enter Hole w/fishtail Light brown to gray micaceous silty fine to medium sand 851.02 N = 17 35.0 856.62 Undisturbed sample 38.1' = 39.9' 40.0 Brown to light gray micaceous silty fine to coarse sand 841.62 N = 100	20.05				861.62	
30.6 Re-enter Hole w/fishtail Light brown to gray micaceous silty fine to medium sand 851.02 N = 17 35.0 8:6.62 Undisturbed sample 38.1' = 39.9' 40.0 Brown to light gray micaceous silty fine to coarse sand 841.62 N = 100	25.0		0.0	нх	856.62	
35.0 8<6.62	30.6	Re-enter Hole w/fishtail Light brown to gray micaceous silty fine to medium sand			851.02	N * 17
40.0 Brown to light gray micaceous silty fine to coarse sand 841.62 N = 100	35.0				86.63	Undisturbed sample 38.1' - 39.9'
	40.0	Brown to light gray micaceous silty fine to coarse sand			841.63	N * 100

Figure 2-14 (Part 1 of 2). Core Boring Record



3

D

2.5 Geology and Seismology



BORING DESIGNATION 2

1	Description	ROD BIT 5 Size Eley	Remerks
5.0	Brown to light gray micaceous silty fine to coarse sand	836.62 N * 100	
3.7	Carbice fishtail refusal	827.92	
6.0		826.62	
0.0		97.0 KX 821.62	
0.3	Water Table	621.32	
0.0		100 D NX 811.62	
8.8	Coring Terminated	802.82	
	Sec. Sec.		

Figure 2-14 (Part 2 of 2). Core Boring Record





Figure 2-15. Core Boring Record



2.5 Geology and Seismology

80	R)	NG
DE	51	GNATION

à

Hepth Ft	Description	ROD	B1 Siz	Eley		Remarks	
0.0		T		828.37			
5.0	Yeliowish brown/light gray micaceous silty fine to coarse sand			623.37	N * 37		
10.0	Brown/light gray micaceous silty fine to coarse sand			616.37	N + 100		
13.1	Carbide fishtail refusal			815.27			
15.0		59.	NX.	813.37			
18.1	Water Table			810.27			
20.0	Coring Terminated			808.37			
		1					

Figure 2-16. Core Boring Record



Oconee ISFSI Safety Analysis Report

BORING DESIGNATION 8-1

Depth	Description	ROD	BIT ELEY	Remarks
5t 0.0	Reddish brown, mics, silty, fine	+	626.32	N + 45
5.0	to coarse send (some ground)		623.32	Undisturbed Sample
10.0	Olive brown, mica, silty, fine		618.32	N = 10
	to medium sand		613.32	Undisturbed Sample
15.0	fine to medium send			17.4'-19.9' N * 51
20.05	Reddish yellow, mica, silty, fine to medium send		808.32	N × 44
25.0	Light olive brown/white, mica, silty, fine to medium sand		803 . 32	Undisturbed Sample 27.4'-28.6' N * 49
30.0	Light olive brown/white, mica silty, fine to medium send		798.38	Undistumbed Sample 32.4'-33.5' N * 100
32.5	Water Table		795.82	

Figure 2-17 (Part 1 of 2). Core Boring Record



2.5 Geology and Seismology

BORING DESIGNATION 8-1

epth	Description	R.D	81 512	F ELEY		marks
95.0	Light plive brown, mica, silty fine to coarse sand	+		792.32	N = 100	
40.0	Light clive brown/white, mics, silty fine to medium sand			788.32	N * 100	
46.7	Carbide fishtail resal			781.62		
50.0		\$5.4	NQ	778.32		
65.0				773.32		
60.0		0.5	NQ	768.32		
64.5	Coring Terminated			763.82		
			5			
						BE CH
						경광성
	Reserves of supported and some sought some of the same support of the same sector of the same sector sector success of the same sector success of the	and the second		A Design over the same dataset	Contraction of the local division of the loc	COLUMN A DESCRIPTION OF TAXABLE PARTY & STREET

Figure 2-17 (Part 2 of 2). Core Boring Record


Oconee ISFSI Safety Analysis Report

BORI

BORING DESIGNATION B-2

	631.03	lindicturbad Campia
		2.5'=5.0'
Strong brown, mica, silty fine to medium send	826.01	N = 8 Undisturbed Sample 7.5'-10.0'
Light pole brown, mica, silty, fine to medium ssand	821.01	N * 15 Undisturbed Sample 12.5'-15.0'
Yellowish brown, mica, silty, fine to medium sand	816.01	Undisturbed Sample 15.0'-16.7' N * 23
Strong brown, mica, silty, fine to medium sand	811.01	N * 100
No Description	806.03	N * 100
Very pale brown/yellowish brown mica, silty, fine to coarse sand	801.01	N * 30
Water Table	798.81	
	<pre>Strong brown, mica, silty fine to medium send Light pole brown, mica, silty, fine to medium send Yellowish brown, mica, silty, fine to medium sand Strong brown, mica, silty, fine to medium sand No Description Yery pale brown/yellowish brown mica, silty, fine to coarse sand Water Table</pre>	Strong brown, mica, silty fine 826.01 to medium sand 821.01 Light pole brown, mica, silty, 821.01 Yellowish brown, mica, silty, 816.01 Tine to medium sand 816.01 Strong brown, mica, silty, fine 811.01 No Description 806.01 Water Table 798.81

Figure 2-18 (Part 1 of 2). Core Boring Record



2.5 Geology and Seismology





Figure 2-18 (Part 2 of 2). Core Boring Record



2-71

0

Oconee ISFSI Safety Analysis Report

	'Ya,				BORING DESIGNATION	8
th	Description	ROD 81	F Eley	Rea	merka.	-
0.0		-	820.98	Undisturbed 2.0'-4.5'	Sample	
5.0	Light yellowish brown/reddish brown, mica, silty, fine to medium sand		815.98	Undisturbed 4.5'+7.0' N * 12	Sample	
10.0	Light yellowish brown/strong brown, mica, fine to medium sandy silt		810.98	Undisturbed 12.0'-14.5' N * 11	Sample	
15.0			805.98	Undisturbed 17.0'-19.5'	Sample	
0.0	Light yellowish brown/strong brown mica, fine to medium sandy silt		800.98	Undisturbed 1 19.5'-22.0' N * 13	Sample	
3.9	water Table		797.08			
5.0			795.98	Undisturbed 27.0'-29.5'	Sample	-
30.0	White/pinkish gray, mica, silty, fine to coarse sand		7 90 . 98	N - 65 Undisturbed 32.0'-33.6'	Sample	

Figure 2-19 (Part 1 of 2). Core Boring Record



2.5 Geology and Seismology

1



epth Ft	Description	ROD	81' 512	Eley	Remarks
35.0	White/pinkish gray, mica.silty. fine to medium sand			785.98	N * 100
37.0	White/pinkish gray, mica, silty fine to medium sand			783.98	N * 27
40.1	Carbide fishtail refusal			780.88	
45.0		99	RQ	775.98	
50.1	Coring Terminated			770.88	

Figure 2-19 (Part 2 of 2). Core Boring Record



l

Oconee ISFSI Safety Analysis Report

BORING DESIGNATION 8-4

Depth	Description	ROO	81 51z	E ELEX	Resarks
0.0				878.53	Undisturbed Sample 2.4'+4.9'
5.0	Reddish brown/red. mica. silty. fine to medium very sandy clay			673,53	N * 19 Undisturbed Sample 7.4'-5.9'
10.0	Reddish brown/red. mica, silty. fine to medium very sandy clay			668.53	N * 100
	Light brown yellow/yellowish brown, mice, silty, fine to coarse send (with gravel)				N * 49
15.0	Light brownish yellow/yellowish brown, silty, fine to coarse sand			863.53	N * 100
20.2	Carbide fishtail refusal			856.33	
25.0		12.1	NQ	853.53	
30.0		0	NQ	848.53	
35.0				843.53	
40.0	Yellow/brownish yellow, mics, silty fine to medium sand			838.55	N * 100

14

Figure 2-20 (Part 1 of 2). Core Boring Record



2.5 Geology and Seismology

BORING DESIGNATION 8-4

	Description	T	SI	P Elex	Remarks
	Pale brown/light yellow brown, mica, silty, fine to medium send	Ī			N * 100
	Water Table			834.63	
	No Description			833.53	N * 100
	Carbide fishtail refusal			828.63	
				826.53	
		91	NQ	823.53	
1	Coring Terminated			818.63	
1					

36

Figure 2-20 (Part 2 of 2). Core Boring Record



Ď

Oconee ISFSI Safety Analysis Report

Rem Undisturbed 2.2'-4.7'	IT Eley	ROD	Description	
Undisturbed 2.2'-4.7'			the second s	an i
	853.63			.0
N = 12 Undisturbed 7.2'-8.2'	848.63		Red/Reodis. brown, mice, silty, fine/medium sand	.0
N * 12			Red/yellowish red. mica, silty, clay, fine to medium sand	
Undisturbed 12.2'+14.7	843.63			0.0
Undisturbed 14.7'=17.2' N = 11	836.63		Light yellow brown/brownish yellow, mica, silty, fine to coarse sand (with gravel)	.0
Undisturbed 22.2'+24.7	833.63			0.0
Undisturbed 24.7'-26.7' N * 100	825.63		White yellowish brown, mics, silty fine to coarse sand	.0
	825.38		Carbide fishtail refusal	.8
	823.68	0 10		0.0
	023.00	U NU		0.0
N = 12 Undisturbed 12.2'+14.7' Undisturbed 4.7'+17.2' N = 11 Undisturbed 22.2'+24.7' Undisturbed 24.7'+26.7' N = 100	1	643.03 836.63 833.63 828.63 825.38 823.68	843.03 836.63 833.63 828.63 825.38 0 NQ 823.68	Red/yellowish red. mica, silty, clay, fine to medium sand Light yellow brown/brownish yellow, mica, silty, fine to coarse sand (with gravel) White yellowish brown, mica, silty fine to coarse sand Carbide fishtail refusal 0 NQ 823.63

Figure 2-21 (Part 1 of 2). Core Boring Record



۵

-

1.7 Goology and Seismology



ŧ

Figure 2-21 (Part 2 of 2). Core Boring Record

2.77

BORI	NG	
DE S1	GNATI	ON .

B-1*

1	Description	Size Elex	BAGATES
0	Strong brown/reddish brown Mica, silty, fine to coarse sand	824.59	N · 12
5.0		619.55	Undisturbed Sample 7.0'-9.5'
10.0	Readish brown/brownish yellow. mics silty fine/coarse sand	614.59	N * 17 Undisturbed Sample 9.5'*12.0'
15.0		809.59	Undisturbed Sample 17.0'-19.5'
20.0	White/brown mics, silty fine/ coarse sand w/gravels	804.59	N = 59 Undisturbed sample
25.0	White/brown mica, silty fine to coarse sand	799.59	N * 100
26.2	Water Table	798.39	
30.4	Carbice Refusal Boring Temminated	794.19	
		1.	
	Second Second		

Figure 2-22. Core Boring Record



2.5 Geology and Seismology

BORING DESIGNATION 8-2*

Depth	Description	ROD	BIT	Eley	Remarks
Ft					
0.0	Red m silty, clayey, fine/ medium sand		31	. 38	N * 20
\$.0			66	4.38	Undisturbed Sample 6.9'-9.4'
10.0	Strong brown/white, mica. silty fine/medium sand		8	59.3E	N = B Undisturbed sample 9.4'-11.4'
15.0	Strong brown/dark brown, silty, mica, file/medium sand		8	54 . 38	N * 9 Undisturbed Sample 16.9'+18.9'
					Undisturbed Sample 18.9'-20.9'
20.0	White, brown, mica, silty, fine to medium sand		8	49.38	Undisturbed Sample 20.9'-22.9' N * 100
25.0			8	44.36	
30.0	Yellowish red/strong brown mica, silty fine.medium sand		8	39.38	N * 16
				-	
			-		and an international and a second state of the second state of the second state of the

Figure 2-23 (Part 1 of 2). Core Boring Record



Oconec ISFSI Safety Analysis Report

an ann an an an an Anna an Anna an Anna

BORING DESIGNATION B-2*

Denth	Description	ROO	SIZE	Elev	Renarks
Ft [1			
35.0			1	34.34	Undisturbed Sample 36.9'-38.9'
					Undisturbed Sample 38.9'-40.9'
40.0	White/brown, mica, silty, fine/ medium sand			329.34	N * 17
45.0	White/strong brown mica, silty. fine/medium sand		1	324.38	N = 21
50.0			1	819.38	Undisturbed Sample 51.9'-54.1'
	Brown/yellowish red, mica, silty, fine medium sand				N * 29
55.0	Yellowish red/brown, mica, silty, fine/coarse sand		ľ	14.38	N = 27
60.0	White/brown mica, silty, fine/medium sand			809.38	N = 100 Undisturbed Sample 61.9'-62.65'
67.0	Carbide refusal			802.36	
	Boring Terminated			-	
	an anna an				1

Figure 2-23 (Part 2 of 2). Core Boring Record



2.5 Geology and Seismology

BORING DESIGNATION 8-3*

	Description	Size Ele	Y Remarks
			친구가 가지가 것
.0	Red, mica, silty, clayey, fine/ medium sand	861.5	N * 16
.0		856.5	Undiscurbed Sample 7.0'-9.5'
	Dark brown/strong brown, mica, silty fine/coarse sand with gravels		N * 20
0.0	Brown/strong brown, mica, silty, fine/medium sand	\$51.5	4 N = 12
5.0		846.5	•
20.0		841.5	Undisturbed Sample 20.0'-22.5
			Undisturbed Sample 22.5'-24.4'
25.0	White.'light brown mica, silty, fine/ ~dium sand	836.5	N = 25 Undisturbed Sample 26.9'=28.8'
	White/light brown, mica, silty, fine/medium sand		K = 48
30.0	White/strong brown, mica, silty, fine/coarse sand	8.21 .	54 N = 46

Figure 2-24 (Part 1 of 2). Core Boring Record

Oconee ISFSI Safety Analysis Report

BORING DESIGNATION B-3*

th	Description	ROD BIT	Remains
35.0	Brown/dark brown, mica, silty, fine/medium sand	826.54	N * 21
40.0		821.54	N = 100
45.0		816.54	N # 100
50.9	Carbide refusal Boring Terminated	810.64	

Figure 2-24 (Part 2 of 2). Cure Boring Record



2.5 Geology and Seismology

BORING DESIGNATION B-4*

anth	Description	S	Size	Eley	Renurks
t i	and the second design of the				
0.0	Light gray/yellowish brown, mica silty, fine/course sand		8	14.66	N = 18
5.0				109.66	Undisturbed Sample 7.1'=9.6'
10.0	Red mica, silty, fine/medium very sandy clay			304.66	Undisturbed Sample 9.6'-12.1' N = 19
15.0				799.66	Undisturbed Sample 17.1'-19.5'
18.3	Water Table			796.36	
20.0	Red/Yellowish med, mica, finr/ medium sandy silt			794.66	Undisturbed Sample 19.6 -22.1' N = 12
25.0				789.66	Undisturbed Sample 27.1'-29.6'
					Undisturbed Sample 29.6'-30.1'
30.0	White/brown, mica, silty, fine to medium sand			784.63	N = 52 Undisturbed Sample 32.1'-33.5'

Figure 2-25 (Part 1 of 2). Core Boring Record

2-83

Oconee ISFSI Safety Analysis Report

BORING DESIGNATION B-4*

h	Description	Size Eley	Returks
35.0	Gray/white, mica, silty fine/ medium sand	779.66	N = 55
	White/brown, mica, silty, fine/ coarse sand		K * 37
40.0	Pinkish gray, mica, silty, fine/ coarse sand	774.66	X * 52
45.0	Light brown/reddish yellow, mica silty, fine/coarse sand	769.56	N * 100
50.0	Dark brown/yellowish brown, mica, silty fine/coarse sand	764.66	N = 100
57.4	Carbide fishtail refusal Boring Terminated	757.28	
		1	

 σ_{a}

Figure 2-25 (Part 2 of 2). Core Boring Record



2.5 Geology and Seismology

	Description	ROD	BIT Size Elev	Remarks
0	Dark brown/white, mica, silty fine/medium sand		617.17	N* 25
0			812.17	Undisturbed Sample 6.5'-8.0'
9.0	White/brown, mica, silty, fine/ coarse sand		807.17	N* 100
2.6	Carbide fishtail refusal Boring Terminated		804.57	
			-	

BORING DESIGNATION 8-5*

> Figure 2-26. Core Boring Record







2.6 REFERENCES

- 1. Geologic Notes, Division of Geology, State Development Board, Vol. 7, No. 5, September-October 1963.
- Chemical Character of Surface Waters of South Carolina, South Carolina State Development Board, (Bulletin No. 16C) 1962.
- Soil Survey Oconee County, South Carolina, United States Department of Agriculture, Series 1958, No. 25, February 1963.
- 4. Storage of Radioactive Wastes in Basement Rock Beneath the Savannah River Plant, DP-844 Waste Disposal and Processing (TID-4500, 28th Ed.), March 1964.
- 5. Oconee Nuclear Station, Final Safety Analysis Report.
- Topical Report for the Nutech Horizontal Modular Storage (NUHOMS-24P) System for Irradiated Nuclear Fuel, NUH-002, Revision 1A, July, 1989.
- STRUDI. Structural Design Language Release 5.3 McDonnell Douglas Architectural Engineering and Construction Systems Company P.O. Box 516, St. Louis, MO 63166.
- 8. Earth and Rockfill Dam Engineering, Sowers, G. F., Asia Publishing House, Bombay, 1961.
- 9. Soil Mechanics, T. William Lambe and Robert V. Whitman, 1969.



Chapter ... Principal Design Criteria

CHAPTER 3. PRINCIPAL DESIGN CRITERIA

Chapter 3. Principal Design Criteria

O-onee ISFSI Safety Analysis Report



3-2

3.1 PURPOSE OF THE OCONEE ISFSI

The purpose of the Oconee ISFSI is to insure the uninterrupted operation of the three unit Oconee Nuclear Station by providing additional long-term spent fuel storage capacity. The existing storage system consisting of two separate wet spent fuel pools is rapidly approaching a maximum operating inventory. The ISFSI utilizes the NUHOMS-24P System. NUHOMS-24P is comprised of a series of reinforced concrete HSMs which will each house a stainless steel, helium filled DSC containing 24 qualified spent fuel assemblies. The DSC top end shield plug and separate cover plate are both independently seal welded to provide total confinement of the irradiated fuel. A shielded transfer cask is used to transfer the DSC to the HSM from the spent fuel pool. During storage, the HSM provides radiation shielding and passive decay heat removal from the DSC.

3.1.1 MATERIAL TO BE STORED

Each DSC is capable of storing 24 PWR assemblies. The following subsections will address the physical, reactivity, thermal and radiological characteristics of spent fuel to be stored in the DSC.

3.1.1.1 Physical Characteristics

The physical characteristics of the reference 15×15 fuel arc listed in Table 3-1 on page 3-6. Additional information may be found in the Oconee FSAR. Section 4.0.

3.1.1.2 Reactivity Characteristics

The reactivity of the spent fuel assemblies must be limited for criticality control purposes. Reactivity is a function of both the initial enrichment and the discharge burnup. A reactivity equivalence curve which shows the acceptable combinations of initial enrichment and discharge burnup is given in Figure 10-1 on page 10-10. For criticality control, the spent fuel assemblies must fall into the acceptable range above the initial enrichment burnup curve in order to qualify for storage in the DSC. Despite the multiple verification steps and extensive administrative controls used to assule selection of qualified irradiated fuel assemblies, criticality control for a misloaded array of unirradiated fuel is maintained by assuring that the DSC is filled with borated water (≥ 1810 ppm boron) and submerged in a borated water spent fuel pool (≥ 1810 ppm boron) during loading and unloading operations.

In the event that unqualified IFAs or unirradiated assemblies are erroneously placed in the DSC, the double contingency principle is applied such that the negative reactivity worth of the (approximately 2000 ppm) soluble boron in the spent fuel pool water (from which the DSC cavity will be filled initially) is more than sufficient to maintain k-eff well below 0.95. Analysis shows that the soluble boron provides sufficient margin to maintain K-eff below 0.95 (0.98 under optimum moderator conditions) for 24 new, 4.0 wt % enriched fuel assemblies loaded into the DSC.

3.1.1.3 Thermal Characteristics

The heat generation is limited to 0.66 kw per fuel assembly. This value is based on storage of 24 assemblies per DSC with a nominal burnup of 40,000 MWD MTU, an initial enrichment of 4.0 wt % U-235 and a nominal decay period of ten years. Other combinations of burnup, initial enrichment and cooling times may also be acceptable upon further analysis demonstrating acceptable decay heat levels.



3.1 Purpose of the Oconce ISFS!

3.1.1.4 Radiological Characteristics

The DSC is designed for a maximum dose rate of 200 mr/hr at the surface of the top (with temporary neutron shielding if necessary during welding operations) and bottom end shield plugs. The HSM is designed for an average dose rate of 20 mr/hr at the surface of the module dropping down to a negligible level at the site boundary. Fuel with a maximum burnup of 40,000 MWD MTU, an initial enrichment of 4.0 w/o/U-235 and a decay of ten years will not exceed these dose values. Other combinations of burnup, initial enrichment and cooling times may also be acceptable upon further analysis demonstrating acceptable radiation dose rate levels.

3.1.2 GENERAL OPERATING FUNCTIONS

3.1.2.1 Overall Functions of the Facility

The Oconee ISFSI is designed to maximize the use of existing site features and equipment and minimize the need to add or modify equipment. The storage facility is located away from the existing plant security boundary such that a separate security "island" is created. The only services required from the station during the ongoing storage mode will be through security surveillance equipment tied in with the plant security center. The storage facility is included in routine daily security patrols. Power supply to the storage facility is retail. Other support services fro, the plant are necessary only during loading (and unloading) operations.

Following periodic delivery of the individual DSCs and construction of the HSMs, the DSC is loaded into the transfer cask and the two are lowered into the spent fuel pool. The DSC transfer cask is loaded with 24 spent fuel assemblies previously selected per criteria given in Section 10.3, "Operational Control and Limit Specification" on page 10.9. Once fuel loading is complete, the DSC is fitted with its top end shield plug and pulled out of the pool. The water level in the DSC is then lowered slightly and the top end shield plug is welded into place. Ti is is followed by further draining and eventual vacuum drying of the DSC cavity. The cavity is then back-filled with helium followed by further seal welding of both penetrations. An additional cover plate is welded over the top end shield plug, the cask lid bolted in place, and the transfer cask is then lowered to the transfer trailer and rotated to the horizontal position. Transfer from the spent fuel pool receiving area to the in lependent storage facility is done with the use of a separate tractor. The transfer trailer is then careful v aligned with the opening in the HSM to allow the hydraulic ram system to push the DSC out of the transfer cask and into the HSM. This method utilizes a small penetration at the bottom of the transfer cask to allow access to the DSC through the transfer cask bottom. A large access door is then lowered and tack welded in place to close off the HSM access.

The HSMs are constructed on a level, reinforced concrete slab designed for normal transfer and storage conditions and postulated accidents.

Once loaded and secured, the passive design of the HSM provides for sufficient radiation shielding, tornado missile protection, and decay heat removal capabilities for the stored spent fuel. The double seal welded DSC closure system together with multi-pass welding procedures provide a multiple barrier against releases of radioactive material.

A more detailed description of each NUHOMS-24P system component is provided in the following subsections.

3.1.2.2 Handling and Transfer Equipment

All components of the NUHOMS-24P system are designed to interface where necessary with all existing Oconee fuel handling storage equipment and facilities. This includes fuel pool receiving areas, radwaste



systems, overhead cranes, yoke and yoke extension, fuel handling bridge and mast, auxiliary hoists, water, power and gas supplies, and clearance restrictions.

The additional equipment required to support the operation of the NUHOMS-24P system includes the DSC, the transfer cask, the transfer trailer with hydraulic alignment mechanisms, the hydraulic ram assembly, the HSM and various miscellaneous tools, lids, gauges, hoses. Other equipment necessary to operate the system include a tractor to be used for moving the transfer trailer to and from the ISFSI, and a mobile yard crane for raising and lowering the HSM front access door. This equipment is further described as follows:

- <u>DSC</u> The DSC serves as the actual confinement vessel for the 24 fuel assemblies during both the storage mode and transfer operations. Seal welds on the top end shield plug and cover plate provide multiple containment of all radioactive products within or on the surface of the spent fuel assemblies. The top and bottom shield plugs also provide for biological shielding during DSC welding, drying, and backfilling, operations, transport of the fuel assemblies, and during all operations performed at the front end of the HSM.
- 2 <u>Transfer Cask</u> The transfer cask provides for dry transfer of the DSC from the Oconee fuel storage pool to the storage facility. The transfer cask utilizes a lead gamma shield and a solid neutron shield to maintain acceptable surface dose levels during transfer operations. A removable access plate at the bottom of the cask provides access to the DSC by the hydraulic ram during transfer of the DSC into or out of the HSM. The cask has a bolt on closure lid to keep the DSC in place during cask movement. Lift trunnions are provided at the top end of the transfer cask to interface with a lift beam which will in turn interface with the spent fuel pool overhead crane. These top lift trunnions together with bottom trunnions provide cask support on the trailer during transfer operations.
- 3. <u>Transfer Trailer</u> The transfer trailer allows for movement of the entire DSC transfer cask assembly to the ISFSI. It is designed with a positioning mechanism that moves the cask in the horizontal and vertical directions to ensure alignment with the HSM. Final alignment accuracy is verified by an optical alignment system.
- 4. <u>Hydraulic Ram</u> The hydraulic ram assembly is stored and transported on a separate trailer but is mounted on the ground during movement of the DSC into the HSM. The ram is aligned with the bottom access portal of the horizontally positioned cask and engaged to slowly push the DSC from the cask into the HSM. A grappling ring on the bottom of the DSC and grappling arms on the hydraulic ram allow for eventual retrieval of the DSC using the same operations in reverse.
- 5 <u>Horizontal Storage Module (HSM)</u> The HSM provides protection for the DSC during the storage mode and provides sufficie biological shielding from the stored spent fuel Passive decay heat removal results from air entering shielded air duets near the bottom of the structure, passing up and around the DSC and picking up heat before being exhausted through shielded vents at the top of the HSM. The HSM design includes a front access fitted with a carbon steel door and the coupling system for mating with the transfer cask. The HSM is fitted with a set of rails which serve as a bearing surface for movement of the DSC into and out of the module and as the primary support structure for the DSC during storage.

A more detailed description of these primary NUHOMS components, including design criteria, is provided in Chapter 4, "Storage System" on page 4-1.

3.1.3 TABLES

Table 3-1. Physical Characteristics of PWR Fuel Assemblies Based on Nominal Design					
Алтау	15 x 15				
Maximum Assembly Length (including radiation growth and control component) (in.)	173				
Weight (lb.)	1,682				
Number of Fuel Rods	208				
Number of Guide Tubes	16				
Number of Instrument Tubes	1				
Fuel Rod Length (in.)	153.69				
Active Fuel Length (in.)	141.8-144.0				
Maximur Distance between Grid Straps (in.)	21 7/32(1)				

Note: ⁽¹⁾Grid straps are placed on intervals of 21 $3/32 \pm 1/16$ inch. Thus the maximum interval is 21 7/32 inch. These tolerances do not accumulate. The spacers in the DSC are two inches wide and the fuel grid straps are 1 1/2 inch wide (higher for later zircaloy grid fuel). Therefore, fuel assembly support will be provided at the grid straps by the DSC spacer discs through the entire tolerance range of 20.97 inches (20 31/32) - 21.22 inches (21 7/32). The nominal value of 21.12 used in Revision 1 of the NUHOMS-24P Topical Report (Table 3.1-2) falls within this range.





3.2 DESIGN CRITERIA FOR ENVIRONMENTAL CONDITIONS AND NATURAL PHENOMENA

The Oconee ISFSI is designed to perform its intended safety function under normal and extreme environmental conditions. In general, the structural and mechanical safety criteria of the ISFSI are the same as or enveloped by the criteria specified in the NUHOMS-24P Topical Report.

Details of the HSM lightning protection are contained in Section 8.2, "Accidents" on page 8-5. Oconee foundation conditions are described in Section 2.5, "Geology and Seismology" on page 2-43.

3.2.1 TORNADO AND WIND LOADINGS

3.2.1.1 Applicable Design Parameters

The ISFSI was constructed within the existing boundaries of the Oconee Nuclear Station. As stated in Section 3.2.1 (Tornado and Wind Loadings) of Reference 1 on page 3-23, the most severe tornado and wind loadings specified by NRC Regulatory Guide 1.76 and NUREG-0800, Section 3.5.1.4, were selected for design consideration. Therefore, both the HSM and the transfer cask are designed in accordance with NRC Regulatory Guide 1.76 and NUREG-0800, Section 3.5.1.4.

As stated in Section 3.2.1.1 of Reference 1 on page 3-23, "... the maximum wind speed is 360 miles per hour, the rotational speed is 290 miles per hour, the maximum translational speed is 70 miles per hour, the radius of the maximum rotational speed is 150 feet, the pressure drop across the tornado is 3.0 psi, and the rate of pressure drop is 2.0 psi per second."

3.2.1.2 Determination of Forces on Structures

The tornado loading combination used for design of the HSM is:

$$y = 1/0 (1.0D + 1.0L + 1.0T_0 + 1.0W_t + 1.0P_t)$$

Where y = required yield strength of the structure

- \emptyset = concrete capacit; reduction factor
- 0 == 0.90 for concrete flexure
- $\emptyset = 0.85$ for shear in concrete.
- $\emptyset = 0.90$ for axial tension in concrete.
- $\emptyset = 0.70$ for tied compression members.
- $\emptyset = 0.90$ for fabricated structural steel.
- T_o = normal operating temperature.
- 1 = live loads on structure
- D = dead loads of structures and equipment
- $W_1 = stress$ induced by design tornado wind velocity (drag, lift, and torsion)





P, = Stress due to differential pressure

Shape factors will be applied in accordance with ANSI A 58.1 - 1982.

3.2.1.3 Tornado Generated Missiles

As described in Section 3.2.1.2 of Reference 1 on page 3-23, the determination of impact forces created by Design Basis Tornado (DBT) generated missiles for the HSM was based on the criteria provided by NUREG-0800, Section 3.5.1.4, III.4. Accordingly, three types of missiles were postulated. The velocity of the missiles was conservatively assumed to be 35 per cent of the combined translational and rotational velocity for the DBT or (0.35)(360), which is 126 miles per hour. For the massive high kinetic energy deformable missile specified in NUREG-0800, a 3,967 pound automobile with a 20 square foot frontal area impacting at normal incidence was assumed. For the rigid penetration-resistant missile specified, a 276 pound, eight-inch diameter blunt-nosed armor piercing artillery shell, impacting at normal incidence was assumed. For the protective barrier impingement missile specified, a one-inch diameter solid steel sphere was assumed.

The possibility of a tornado damaging a cask/DSC in transit to the HSM is a low probability event. The probability of a tornado occurring at the Oconec site and generating a missile that impacts the cask is less than 1x10⁻⁷ per transfer trip. This is based on site-specific tornado frequencies derived from 35 years of National Severe Storm Forecasting Center data and assumes a conservative exposure time to DBT effects of 24 hours. However, the transfer cask has been evaluated for the tornado wind speed and missiles specified for the HSM. The maximum DBT tornado wind speed of 360 mph produces a design pressure of 304 psf. The 3,967 pound automobile and 276 pound eight inch diameter shell missiles are also considered. The one inch diameter spherical missile effects are enveloped by the eight inch shell missile.

3.2.1.4 Ability of Structures to Perform

The ISFSI is designed to withstand the design basis tornado wind loads. All components of the ISFSI with the exception of the air outlet shielding blocks of the HSM are designed to withstand the tornado generated missile forces as described in Section 8.2.2 of Reference 1 on page 3-23. The loss of the air outlet shielding blocks is discussed in Section 8 of the NUHOMS-24P Topical Report. The HSM is anchored to the foundation slab to mitigate overturning and sliding effects using dowel rods of a size and spacing consistent with the HSM wall vertical reinforcement.

The possibility of total air inlet and outlet blockage by foreign objects or burial under debris during a tornado event is considered. The effect of facility burial under debris is presented in Section 8.2 of Reference 1 on page 3-23.

The transfer cask analysis for tornado wind speed and missile effects was performed for the cask secured in the horizontal position on the support skid and transfer trailer are following criteria were used to evaluate the adequacy of the transfer cask for the loads described in Section 3.2.1.3, "Tornado Generated Missiles."

- 1. Stability
- 2. Penetration Resistance
- 3. Stresses



The main components of the transfer cask considered in this analysis were the structural shell, and the top and bottom cover plates. Since the primary purpose of the solid neutron shield is biological shielding and since it is located on the cask exterior, it was conservatively assumed that the neutron shield will be

ruptured by a DBT missile strike and therefore was not considered in the structural analysis. A brief description of the analysis follows.

1. Stability Analysis

A stability analysis for the transfer cask mounted on the skid/trailer assembly was performed for the wind pressure loads and the massive missile impact.

For the wind pressure loads, the overturning moment was compared to the stabilizing moment to determine the factor of safety against overturning. A factor of safety of 3.1 was eclculated.

For the massive missile impact, it was conservatively assumed that the missile impacts the uppermost part of the cask. The angle of rotation (θ) of the cask/skid/trailer arrangement at inspact was calculated as 3.0 degrees assuming a rigid pavement. This calculation was based on the conservation of angular momentum, and also compared to the angle (θ_{up}) necessary for the cask/skid/trailer to tip over. Tip-over occurs when the center of gravity of the cask is directly above the point of rotation. This was calculated as 32.7 degrees. Since $\theta < \theta_{up}$, tip-over does not occur and the stability of the cask/skid trailer arrangement is maintained.

2. Penetration Analysis

Penetration due to the 276 pound rigid missile was calculated using two formulas obtained from the literature. The added energy absorbing affect of the neutron shield material was omitted from this calculation to give a more conservative result. The first approach, suggested by Nelms (Reference 4 on page 3-23) is for a lead-backed shell:

$$T = \left[\frac{KF}{2.4 S_0 D^{1.6}} \right]^{0.71} = 0.50 \text{ inches}$$

Where: T = Minimum required steel plate or shell thickness to resist penetration

 $KE = Kinetic energy = 1/2 mV^2$

m = Mass of missile = 276 g

= 0.714 lb. sec.²/in.

- v = Velocity of missile
 - = 2,218 in./sec.

 $S_u = -$ Ultimate strength of cask structural shell = 70,000 psi

D = Diameter of missile = 8.0 inches

The second formula used was developed by the Ballistic Research Laboratory (Reference 5 on page 3-23):

$$T = \frac{KE^{2/3}}{672D} = 0.52$$
inches

Where: $KE = Kinetic energy = 1/2 \text{ mV}^2$

m = Mass of missile

- = 8.57 lb. sec.²/ft.
- V = Velocity of missile



3.2 Dsgn. Crit. for Envir. Cond. and Natural Phenomena

Oconee ISFSI Safety Analysis Report

- == 184.8 ft./sec.
- D = Diameter of missile = 8.0 inches

Both methods produce a consistent result which shows a predicted penetration of 0.5 inches compared to the minimum structural shell thickness of 1.5 inches. Therefore the DBT missile will not penetrate the cask and the DSC will remain intact.

3. Stress Analysis

Conservative hand calculations were performed to determine the peak stresses in the cask shell, and the top and bottom cover plates due to DBT loads. A summary of the stress results is provided in the attached Table 3-2 on page 3-12. The analytical method for each of the load cases shown in this table is briefly described below.

- a. <u>Wind Pressure Loads</u>: A uniform line load of 2.18 Kips/ft. was applied to the full length of the cask. The correlation of Roark and Young (Reference 6 on page 3-23) Table 31, Case 9c was conservatively used to calculate membrane and bending stresses. The analyses of the three inch top and two inch bottom cover plates were performed using Case 10, Table 24 of Roark and Young. The top cover plate was assumed pinned at the edges while fixed edge supports were assumed for the bottom cover plate.
- b. <u>Massive Missile Impact</u>: Based on the conservation of angular momentum, the total force on impact was calculated to be 257 kips. This force was applied as a line load to the cask shell and as a pressure load to the top and bottom cover plates. The analysis method followed that described above for the wind pressure loads.
- c. <u>Penetration Resistance Missile</u>: The impact force due to the eight inch diameter, 276 pound missile was calculated from the conservation of momentum as 63.4 kips. Case 9a, Table 31 of Reference 6 on page 3-23 was used to calculate the membrane and bending stress for the cask shell while Cases 16 and 17, Table 24 of Reference 6 on page 3-23 were used to calculate the stresses in the top and bottom cover plates respectively.

3.2.2 WATER LEVEL (FLOOD) DESIGN

The grade level of the ISFSI is El 825.0. This elevation is 11.9 ft. higher than the calculated maximum flood water elevation at Oconee due to a postulated breach of the upstream Jocassee Dam (See Section 2.4.5.1, "Flood Protection Measures for Oconee Station Seismic Class 1 Structures" on page 2-31).

3.2.3 SEISMIC DESIGN

3.2.3.1 Input Criteria

The maximum horizontal and vertical ground acceleration (Maximum Hypothetical Earthquake - MHE) specified for the Oconee site is .15g (Reference 3 on page 3-23, Section 2.5.2.8, "Design Response Spectra" on page 2-219). The Oconee site accelerations are less than the analyzed values of .17g vertical and .25g horizontal used for NUHOMS components (Reference 1 on page 3-23) and thus are enveloped by the generic NUHOMS analysis.

The Oconee HSMs were designed using the seismic criteria from Reference 1 on page 3-23. As stated in 'section 3.2.3 of Reference 1 on page 3-23, "The maximum horizontal ground acceleration component selected for design of the NUHOMS-24P was 0.25g. The maximum vertical acceleration component selected was two-thirds of the horizontal component which is 0.17g. In order to establish the amplification factor associated with the generic design basis response spectra, various frequency analyses were performed

for the different NUHOMS-24P components and structures. The results of these analyses indicated that the dominant lateral frequency for the reinforced concrete HSM was 25 Hertz. The corresponding horizontal seismic acceleration used for design of the HSM was 0.32g. Conservatively assuming that the dominant HSM vertical frequency is also 25 Hz. produces a vertical seismic design acceleration of 0.22g."

The effects of a seismic event occurring during the transport of a loaded DSC resting inside the NUHOMS-24P transfer cask and secured to the transport skid/trailer was postulated. This load case is conservatively enveloped by the postulated normal transport load accelerations of ± 0.5 g acting in the vertical, axial, and transverse directions, applied simultaneously at the center of gravity of the transfer cask, as specified in Section 8 of Reference 1 on page 3-23. These accelerations envelope those which would result from a seismic event in the highly unlikely event that a design basis earthquake would occur during transport of the loaded DSC to or from the HSM.

3.2.3.2 Seismic System Analysis

The stresses in the Oconee HSM and the DSC due to the .15g horizontal and vertical motion for the MHE are enveloped by the results of the generic seismic analysis reported in the NUHOMS-24P Topical Report (Reference 1 on page 3-23). The maximum HSM reinforced concrete bending moments and shear forces in Table 8.2-3 of Reference 1 on page 3-23 envelope the seismic loads at Oconec.

The foundation of the HSM is also designed to withstand the forces generated by the MHE (See Section 2.5.5, "ISFSI Foundation" on page 2-51).

3.2.4 SNOW AND ICE LOADS

The NUHOMS-24P Topical Report specified a postulated live load of 200 pounds/ft² which conservatively envelopes the maximum snow and ice loads for the Oconee site.

3.2.5 COMPINED LOAD CRITERIA

Load combination criteria established in the NUHOMS-24P Topical Report for the HSM, DSC and DSC support assembly meet or envelop the load combinations required by the Oconee FSAR Section 3.8, "Design of Class 1 Structures" on page 3-99 (Reference 3 on page 3-23).

The HSM analyses summarized in Reference 1 on page 3-23 considered combinations of HSMs ranging from a single stand-alone module up to the maximum array size of 2x10. The finite element models used in the analyses are applicable to both side-by-side or back-to-back arrangements. Different DSC loading patterns were analyzed for each size of array to establish the worst case design loadings.

The analyses showed that the single HSM provides the governing case for load combinations containing tornado wind and missile loads, seismic loads and flooding conditions. The postulated failure mode for each of these cases is sliding or overturning of the HSM unit. The analyses also showed that the thermal loads for a 2x10 array control the reinforcement requirements for the walls, roof and foundation members for all intermediate array sizes.

Therefore. Reference 1 on page 3-23 presents a design configuration which envelopes the loads from a single HSM to a 2x10 array.



3.2.6 TABLES

Load Case	Load Description	Stress Category	Calcu			
			Cask Shell	Top Cover Piate	Bottom Cover Plate	Allowable ⁽¹ Stress (ksi)
1	Wind Pressure Loads	Primary Membrane	0.9	0.0	0.0	49.0
		Membrane + Bending	2.9	0.4	0.3	70.0
2	Massive Missile	Primary Membrane	6.4	0.0	0.0	49.0
		Membrane + Bending	20.5	19.7	17.5	70.0
3	Penetration Resistance Missile	Primary Membrane	4,9	n.a	0.0	49.0
		Membrane + Bending	20.3	13.2	22.2	70.0
4	Protective Barrier Missile	Primary Membrane	Bounded by Case (3) Above			49.0
		Membrane + Bending				7'J.0



3-12

3.3 SAFETY PROTECTION SYSTEM

3.3.1 GENERAL

The Oconee ISFSI is designed for safe and secure, long-term containment and storage of IFAs. The major components which assure that the safety objectives are met are listed in Table 3-3 on page 3-19. The major procedures which require special design consideration are:

- · Double Closure Seal Welds on DSC Ends
- · Radiation Exposure During DSC Closure and Drying Operations
- · Minimization of Contamination of DSC Exterior by Pool Water
- Minimization of Radiation Exposure During Transfer of the DSC from the Transfer Cask to the HSM

These items are addressed in the following subsections.

3.3.2 PROTECTION BY MULTIPLE CONFINEMENT BARRIERS AND SYSTEMS

3.3.2.1 Confinement Barriers and Systems

The Oconee ISFSI design incorporates multiple confinement barriers to ensure there will be no release of airborne radioactive effluent to the environment. The radioactivity which must be confined is from the IFAs themselves and DSC exterior contamination from IFA loading operations in the spent fuel storage pool. ISFSI multiple radioactivity confinement barriers are listed in Table 3-4 on page 3-20.

DSC exterior contamination is minimized by preventing spent fuel storage pool water from contacting the DSC exterior. DSC loading procedures (See Section 4.4.1, "Loading and Unloading System" on page 4-21) require that the annulus between the transfer cask and DSC be filled with demineralized water and sealed prior to immersion in the point fuel pool. Annulus sealing is accomplished by an inflatable seal between the transfer cask and DSC. The combination of the above operations provides assurance that the DSC exterior surface has less residual contamination than required for shipping cask externals (i.e., 10CFR 71.87(i)(1)). A surface swipe of the DSC exterior is taken while is in the cask decontamination pit to assure this level of contamination is not exceeded.

The annulus seal is an inflatable fabric reinforced elastomeric tube. An automotive-type valve stem permits inflation to approximately 25 psig. This design can accommodate the maximum variation in the annulus width (.5 to 1.5" at the cask flange). The seal is placed in the annulus and inflated prior to immersion in the fuel pool. It remains in place at least through the completion of top end shield plug decontamination. The seal may remain in place until just prior to DSC seal welding. The seal is then stored for future use, or discarded if it has become damaged.

The function of the annulus seal is to minimize the potential for DSC and cask contamination during fuel loading. It is not intended to be a "safety protection system" for the NUHOMS system. The seal provides an added assurance that minimizes the potential spread of contamination and therefore reduces personnel radiation exposures. The NUHOMS system will safely function with or without the seal, and as such, its correct placement and operation are not critical to the safety of t² e system. In the event that the seal should fail, the water filled annulus will minimize the spread of contamination below the top of the DSC.

3.3 Safety Protection System

Should the DSC surface become contaminated, clean demineralized water will be flushed through the DSC/transfer cask annulus until surface smears show that the contamination levels meet Technical Specification limits.

Transfer cask external contamination will also be controlled to minimize personnel radiation exposure and potential off-site radiological releases during cask handling operations outside the spent fuel pool. 49CFR 173.443(d), which governs contamination levels for off-site shipment in a closed, exclusive use vehicle, will be used as a guideline for establishing cask release limits.

Containment of radioactive material associated with IFAs is provided by fuel cladding, the stainless steel DSC body, and double seal welded primary and secondary closures. These multiple confinement barriers assure that any accidental radioactive releases from stored IFAs to the environment will be ALARA.

3.3.2.2 Ventilation - Offgas

The ISFSI design limits the temperature history of stored fuel rods, such that no fuel damage will occur under design basis conditions. Decay heat dissipation is discussed in Section 1.2.3.2, "Decay Heat Dissipation" on page 1-6 of this SAR. ISFSI response to abnormal cooling conditions (i.e. convective air flow blockage conditions) is provided in Chapter 8, "Accident Analyses" on page 8-1 of this SAR. There are no radioactive effluent releases during normal operations. Additionally, there are no credible accidents which cause releases of radioactive effluent from the DSC. Therefore, there are no offgas system or radiological effluent release monitoring requirements for the ISFSI.

The only offgas concern results from the DSC transfer cask purge and drying operations. During this operation, the gases purged from the DSC transfer cask internals are directed to the spent fuel storage facility HVAC system upstream of the Engineered Safety Feature (ESF) HEPA, and carbon filter units. The purged air and helium are ultimately released from the station unit vent. Potential radiological effluent releases are monitored by both spent fuel storage facility HVAC and unit vent monitors prior to release. This is the same method and system currently utilized for spent fuel shipping cask operations.

3.3.3 PROTECTION BY EQUIPMENT AND INSTRUMENTATION SELECTION

3.3.3.1 Equipment

1

1

1

1

The transfer cask and DSC are the only equipment considered safety related during normal and off-normal operations. The HSM is not safety related. However, the functions of the HSM are considered important to the safe operation of the ISFSI and therefore the HSM is designed, constructed, and tested in accordance with the Duke QA-2 Quality Assurance Program. The design criteria for all equipment comprising the ISFSI that is classified to be important to safety are summarized in Sectic 3.4, "Summary of Storage System Design Criteria" on page 3-21 of this SAR. Design code standards for ISFSI components are summarized in Table 3-5 on page 3-20.

The design criteria for the NUHOMS reinforced concrete HSM including its foundation and DSC support structure, the DSC and its internal basket assembly, and the transfer cask are provided in Section 3.2, "Design Criteria for Environmental Conditions and Natural Phenomena" on page 3-7 and summarized in Tables 3.2-1 through 3.2-9 of Reference 1 on page 3-23.

The Oconee lifting beams used for movement of the transfer cask within the fuel building are designed and procured as components important to safety. The lifting beams used in that part of the operation are controlled by 10CFR Part 50 and NUREG-0612 and are designed to ANSI 14.6-1986 criteria for nonredundant vokes.

The vacuum drying system described in Section 4.7.3 of Reference 1 on page 3-23 is not safety related. Failure of any part of this system can only result in a delay of operations, and not in a hazardous situation to the public or operating personnel. The welding materials required to make the closure welds on the DSC top end shield plug and top cover plate are purchased to the same ASME Code criteria as the DSC (Section NB Class 2). The actual equipment used for making the closure welds is purchased in accordance with standard industry codes such as ANSI, AWS and AISC.

As noted in Section 4.5.5, "Transfer Components" on page 4-32 of this SAR, all other components of the NUHOMS system, including the transfer cask skid, skid positioning system, and hydraulic ram system are required to perform their function to successfully transfer a DSC to and from the HSM. These systems are described in Reference 1 on page 3-23 with the design requirements further delineated in Chapter 4, "Storage System" on page 4-1 of this SAR. However, the failure of any of this equipment may cause additional operational effort but will not endanger he health and safety of the public or plant personnel. Therefore, these transfer components are not considered to be important to safety and are therefore designed, constructed, and tested in accordance with accepted industry standards.

In addition, the transfer cask, HSM, and DSC have been designed to meet very conservative design criteria including postulated conditions which envelop those which may result from the mechanistic failure of the transfer system equipment. Design conditions such as cask drop accident and jammed DSC have been included even though there is no plausible way for these worst case events to occur. Conservative bounding analysis for these conditions have been performed using minimum material yield strengths, strength reduction factors, and factors of safety in accordance with the stringent requirements of the ASME and ACI Codes. Even when applying this conservative criteria considerable design margin for these components and structures remains as evidenced by the analysis results comparisons with acceptance criteria contained in Reference 1 on page 3-23. Further, these components and structures are fabricated and constructed to the rigorous standards and methods of the ASME and ACI Codes under a 10CFR 50 Appendix B Quality Assurance Program. These include material qualification, welding and nondestructive examination, and strict surveillance and quality control inspection. The resulting high integrity of the Transfer Cask, DSC, and HSM provide more than adequate assurance that the health and safety of the public and plant personnel are protected.

3.3.3.2 Instrumentation

The Oconee ISFSI is designed to maintain a safe and secure, long-term containment and storage environment for IFAs using only totally passive components. Therefore, no safety related instrumentation is required for operation of the facility.

3.3.4 NUCLEAR CRITICALITY SAFETY

3.3.4.1 Control Methods for Prevention of Criticality

A combination of DSC fuel basket design and station administrative procedures assure subcritical conditions exist at all times. DSC fuel basket material properties and geometry are established to assure subcriticality assuming a full loading of IFAs with a specified minimum burnup that encompasses the majority of the available spent fuel inventory at Oconee. Oconee administrative procedures assure that only qualified IFAs are loaded for storage in a DSC and that a minimum soluble boron concentration of 1810 ppm is maintained within the DSC basket cavity during underwater loading unloading operations. IFA qualification for storage in a DSC is determined based on initial enrichment, burnup history and post-irradiation cooling time as governed by Oconee ISFSI Technical Specifications.

IFA qualification requirements are provided in Section 10.2.5. "Administrative Controls" on page 10-6 on Administrative Controls. Using special nuclear materials control and accountability (SNMCA) records



3.3 Safety Protection System

and the burnup results from the Oconee Operator Aided Computer, the specific data needed to characterize any given spent iucl assembly can be gathered. This includes the initial enrichment, discharge burnup, cladding defects (7 any), current storage location, and cumulative cooling time since reactor discharge. After verifying that all the spent fuel specifications of Reference 2 on page 10-11 are met, documentation of individual fuel qualifications will be transmitted to fuel nandling personnel. Oconee administrative procedures will require receipt of this qualification documentation, and independent verification of fuel assembly identification numbers prior to loading a given assembly into the DSC. In addition, all assembly serial numbers will be checked following the complete loading of 24 assemblies into the DSC.

The Oconee ISFSI Technical Specifications which govern IFA qualification for storage are given in Reference 2 on page 10-11. The administrative procedures outlined above will be used to ensure that the requirements for fuel qualification are met.

IFA qualification criteria do not include a specification on axial burnup distribution. The axial burnup profile used in analyzing the nonuniform axial burnup reactivity effects on fully loaded DSC spent fuel storage arrays is considered worst-case based on a comprehensive review of axial burnup profiles generated by the EPRI-NODE computer program (reference Section 3.3.4.3 of Reference 1 on page 3-23). Although some individual IFAs may not be enveloped by the worst-case axial burnup profile considered, the conservative treatment of nonuniform axial burnup in the Reference 1 on page 3-23 analysis and the averaging effects of loading up to 24 qualified IFAs per DSC provide adequate assurance that the K-eff of any loaded DSC configuration will not exceed the worst-case value presented in Reference 1 on page 3-23.

Criticality analyses applicable to Oconee have been performed which demonstrate that subcriticality is maintained within appropriate safety margins under the worst conditions. Worst-case conditions analyzed include fuel misload events and optimum moderation effects on reactivity. A bounding off-normal case of 24 misloaded unirradiated new fuel assemblies with a 4% U-235 initial enrichment combined with the reactivity effects of optimum moderation has been analyzed. A minimum fuel pool water boron concentration of 1810 ppm is assumed in the off-normal case calculations. Criticality analysis details are provided in Section 3.3.4 of Reference I on page 3-23.

3.3.4.2 Error Contingency Criteria

The design basis for preventing criticality in ISFSI spent fuel storage operations is taken from American National Standard Design Requirements for Light Water Reactor Spent Fuel Storage Facilities at Nuclear Power Plants, ANSI ANS-57.2-1983, ANSI ANS-57.2-1983 requires a demonstrated margin of subcriticality of $\geq 0.05 \ \Delta K$ under all credible conditions except under certain extreme off-normal conditions where a $\geq 0.02 \ \Delta K$ subcritical margin may be justified. Additionally, ANSI ANS-57.2-1983 requires all uncertainties be included in the final calculated K_{eff} value at 95.95 tolerance limits. See Section 3.3.4 of Reference 1 on page 3-23 for details on how these criteria are applied in demonstrating ISFSI criticality safety.

3.3.4.3 Verification Analysis

The analysis method which ensures ISFSI criticality safety uses the Criticality Analysis Sequence No. 2 (CSAS2) and the 123GROUPGMTH master cross-section library included in the SCALE-3 system of codes (Reference 2 on page 3-23). CSAS2 consists of two cross-section processing codes (NITAWL and BONAMI), a 1-D transport code for cell-weighting cross-section data (XSDRNPM), and a 3-D monte-carlo code (KENO-IV) for calculating the effective multiplication factor for a system.

In CSAS2 calculations involving the zero burnup intercept point, cross section processing and cell weighting of cross sections was performed assuming fresh fuel. For CSAS2 calculations involving irradiated fuel, cross section processing and cell weighting . cross sections was performed assuming irradiated fuel actinide and fission product isctopics.

Irradiated fuel fissile nuclide number density data was obtained from CASMO-2 (Reference 4 on page 3-23) calculations and input to the CSAS2 criticality code sequence (reference Section 3.3.4.2 of Reference 1 on page 3-23). The CASMO-2 lattice physics code has been used extensively in reactor physics calculations. Its ability to accurately predict fissile nuclide depletion and generation as well as neutron multiplication is well established in benchmark calculations (References 5 and 6 on page 3-23) and through its successful application in numerous licensed reactor physics and core reload design calculations.

The Shielding Analysis Sequence No. 2 (SAS2) included in the SCALE-3 package of codes was used to develop irradiated fuel fission product number density data for input to CSAS2. SAS2 is an industry recognized code which employs ORIGEN-S to perform fuel burnup, depletion and decay calculations.

A set of 40 critical experiments have been analyzed using the CSAS2/123GROUPGMTH reactivity calculation method to demonstrate its applicability to criticality analysis and to establish method bias and variability. The experiments analyzed represent a diverse group of water moderated, heterogeneous oxide fuel arrays separated by various materials (stainless steel. Boral, water, etc.) that are representative of LWR shipping and storage conditions. The method bias and uncertainty applied in the calculation of the final K-eff result is based on CSAS2 123GROUPMTH calculated results for the set of 40 critical experiments summarized in Table 3.3-6 of Reference 1 on page 3-23. All 40 critical experiments included in the method benchmark are similar 'o zero burnup nominal case flooded DSC conditions in that all are water moderated, low enrichment heterogeneous U02 systems. Additional benchmark calculations were performed to demonstrate that the irradiated fuel criticality equivalence method used is conservative when compared to the method bias basis U02 benchmark results (i.e., Reference 1 on page 3-23 Table 3.3-6 results). CSAS2/123GROUPGMTH benchmark results for systems containing Pu02-U02 mixed oxide fuel pins are provided in Table 3.3-7 of Reference 1 on page 3-23. Benchmark data representative of irradiated fuel assemblies was obtained from the results of CASMO-2 infinite lattice criticality calculations: the results of benchmark comparisons between CASMO-2 and CSAS2/CASMO2/SAS2 calculated K-inf values are provided in Table 3.3-8 of Reference 1 on page 3-23. Inspection of the benchi. rk results provided in Reference 1 on page 3-23 Table 3.3-7 and 3.3-8 demonstrates that the criticality/equivalence method used conservatively overpredicts K-eff for systems containing plutonium or irradiated fuel of the type proposed for Oconee ISFS1 storage.

Further details on the analysis method and the ISFSI verification analysis are provided in Section 3.3.4 of Reference 1 on page 3-23.

3.3.5 RADIOLOGICAL PROTECTION

The Oconee ISFSI is designed to maintain onsite and offsite doses ALARA during loading operations and long-term storage conditions. ISFSI loading procedures, shielding design, and access controls provide the necessary radiological protection to assure radiological exposures to station personnel and the public will be maintained ALARA. Further details on collective onsite and offsite doses resulting from ISFSI operations and the ISFSI ALARA evaluation are provided in Chapter 7, "Radiation Protection" on page 7-1 of this SAR.

Access to the spent fuel assemblies stored in the ISFSI is restricted by a security fence, and the thick walls and heavy door of the Horizontal Storage Module. Since there are no active systems in the storage module, there is no need for continuous monitoring of conditions. Appropriate monitoring will be
3.3 Safety Protection System

performed prior to loading or unloading Dry Storage Canisters inside the ISFSI fence. Appropriate monitors are in place inside the station to provide warning of radiation hazards while DSC loading and cask handling operations are performed in the fuel building and loading area. During transport, the transfer cask will be monitored to assure no danger to the health of the public or station personnel.

3.3.6 FIRE AND EXPLOSION PROTECTION

The ISFSI HSM and DSC contain no flammable material and the concrete and steel used for their fabrication can withstand any credible fire hazard. There is no fixed fire suppression system within the boundaries of the ISFSI; however, portable suppression equipment is provided within the fenced boundary. Also, the facility is located such that the station fire brigade can respond to any fire emergency using portable fire suppression equipment or the site's Fire Protection System, as described in Section 9.5.1, "Fire Protection System" on page 9-107 of the Oconee FSAR.

ISFSI initiated explosions are not considered credible since no explosive materials are present in the fission product or cover gases. Externally initiated explosions are considered to be bounded by the design basis tornado generated missile load analysis presented in Section 3.2, "Design Criteria for Environmental Conditions and Natural Phenomena" on page 3-7 of this SAR and Reference 1 on page 3-23.

3.3.7 MATERIALS HANDLING AND STORAGE

Major ISFSI materials handling and storage requirements include irradiated fuel and radioactive waste handling and storage. No hazardous chemicals or chemical reactions are involved in the normal ISFSI loading and storage processes.

All irradiated fuel handling outside the fuel stora^m pool is perforn ed with the fuel assemblies enclosed in a DSC. DSC handling equipment and handling procedures are described in detail in Chapter 4, "Storage System" or page 4-1 and Chapter 5, "Storage System Operations" on page 5-1 of this SAR, respectively.

Radioa: : generation, treatment and disposal is addressed in Chapter 6, "Waste Management" on page 6-1. SAR.

The design criteria for handling spent fuel outside the pool area is to keep the fuel enclosed in the DSC and the Transfer Cask or HSM at all times. There is no waste generation outside the fuel building for normal DSC transfer operations. Waste generated in loading and decontaminating the cask is handled by existing Oconee waste systems in the pool and decontamination areas.

The canister/cask design is such that fuel headling in the pool area is unchanged from normal fuel handling procedures. Specific criteria for selecting fuel assemblies for storage are detailed in Reference 2 on page 10-11. IFAs are selected to meet design criticality, cooling and radiation protection parameters. Once the assemblies are loaded into the DSC, there is no individual assembly handling. Thus, the only fuel handling procedures are those already in existence for the pool and the administrative criteria for selecting assemblies for storage. Damaged fuel assemblies are not normally considered for storage and would be handled according to existing pool procedures in the event damage occurred during DSC loading or unloading in the pool. (Fuel damage in the context of this discussion represents gross clad or structural failure and does not include pin-hole clad leaks.) Fuel handling operations will be monitored with existing pool area monitors.

All radic active waste generation is from cask decontamination and consists of liquid waste which is input into the cask decontamination pit drain and thus into the existing plant liquid radwaste system and solid waste which is collected for disposal via the existing plant radwaste facility.

REV: (30 JUNE 1991)

3.3 Safety Protection System

3.3.8 TABLES

Table 3-3. Ocone: ISFSI Major Components and Functions

• Transfer

Q.

......

ŝ

and

* Dry Storage Canister (DSC)

Guide Sleeves Spacer Disks Support Rods End Shield Plugs DSC Body End Cover Plates

 Horizontal Storage Module (HSM) Concrete Shielding

DSC Support Assembly

- · Foundation
- Transfer Components

Transfer Trailer Hydraulic Ram Trailer Optical Alignment System Cask Ousde IFA Transport, Shielding

Criticality Control, IFA Support, Cover Gas Containment, Padioactive Material Confinement, Shielding (Lead Plugs)

Shielding, DSC Support, DSC Tornado Missile Protection, DSC Cooling

HSM Foundation Support

Transfer Cask Movement, DSC Transfers

à

d

2.3 Safety Protection System

able 3-4. Deoree ISFSI Radioactive Material Confinement Barriers		
Radioactivity Source	Confinement Barriers	
Contaminated Spent Fuel Storage Pool Water	 Demineralized Water in DSC/Transfer Cask Annulus 	
	 inflatable Annulus seal between DSC and Transfer Cask 	
 Irradiated Fuel and Fission Gases) Fuel Cladding	
	2. DSC Body	
	 Seal Welded Primary Closure (Top End Shield Plug) 	
	4. Seal Welded Secondary Closure	

Item	Design Code	Design Criteria
Transfer Cask	ASME Section III Class 2	Presented in Ref. 3.1, Section 3.2.5.5
DSC	ASME Section III Class 1	Presented in Ref. 3.1, Section 3.2.5.2
HSM Including Foundation and DSC Support Structure	ACI 349-85 ACI 318-83 AISC	Presented in Ref. 3.1, Section 3.2.5.1
Transfer Trailer and Skid	Industry Standards ⁽¹⁾	Ref. 3.1, Section 1.3.1.4 and 1.3.1.5
Hydraulic Ram	Industry Standards ⁽²⁾	Ref. 3.1, Section 1.3.1.6
Cask Lifting Devices	ASME Section III. Subsection NF	None required at HSM site. Fuel bldg, lifts controlled by 10CFR Part 50 criteria.
HSM Site Electricity Power	NEC, NEMA, NEPA	Required for DSC transfer operations only.

Note:

- 1. See Sections 1.3, "General Systems Descriptions" on page 1-11 and 3.1.2.2, "Handling and Transfer Equipment" on page 3-4 of this SAR.
- 2. See Section 5.1, "Operation Description" on page 5-3 of this SAR.



3.4 SUMMARY OF STORAGE SYSTEM DESIGN CRITERIA

1. REFERENCE SPENT FUEL CHARACTERISTICS -

- a. 15x15 PWR Assemblies (24 Per Module/DSC)
- b. Decay Heat = .66 KW Per Assembly
- c. Nominal Burnup = 40,000 GWD/MTU
- d. Initial Enrichment = 4.0 weight % U-235
- e. Equivalent Zero Burnup Enrichment ≤1.45 weight % U-235 (criticality)

2. COMPONENT FUNCTIONS -

- a. DSC Provides Fuel Support, End Shielding, Heat Transfer, Criticality Control, and Confinement of cover gas and Radioactive Material.
- b. Transfer Cask Provides Shielding. DSC Loading, Handling and Transfer Mechanism, HSM Docking Functions, and Tornado Wind and Missile Protection.
- c. HSM Provides Shielding, Passive Decay Heat Removal, Structural/Seismic DSC Support and Environmental Protection, including Tornado Wind and Missile Protection
- d. Hydraulic Ram System Provides Mechanism for DSC Transfer From Transfer Cask to HSM and eventual removal of DSC from HSM.

3. ENVIRONMENTAL CONDITIONS -

- a. Maximum Tornado:
 - 1) wind speed = 360 miles per hour
 - 2) rotational speed = 290 miles per hour
 - 3) translational speed = 70 miles per hour
 - 4) pressure drop across the tornado = 3.0 psi
 - 5) rate of pressure drop = 2.0 psi per second
- b. Tornado Missiles @ 35% of the Combined translational and rotational DBT velocity = 126 miles per hour.
 - 1) 3967 pound automobile with a 20 square foot frontal area
 - 2) 276 bound, eight inch diameter blunt-nosed armor piercing artillery shell
 - 3) one-inch diameter solid steel sphere
- c. Flood Design: Not Applicable
- d. Seismic Forces = .17g Vertical. .25g Horizontal (NUHOMS components)

= .15g Vertical, .15g Horizontal (Oconee site conditions)

e. Snow Ice Loads = 200 Pounds Per Square Foot

4. SAFETY PROTECTION -

- a. Normal Operating Clad Temperature ≤ 340°C
- b. Material Confinement Multiple Barrier Concept

3.4 Summary of Storage System Design Criteria



- c. Purged-gasses Passed Through Spent Fuel Pool Vertilation System During Fuel Loading Otherwise Not Applicable.
- d. Criticality Control Through Burnup Credit and 1810 ppm Soluble Boron Credit $K_{\rm eff} < 0.95,$ $K_{\rm eff} < 0.98$ (off-normal)

3-22

3.5 REFERENCES

- Topical Report for the Nutech Horizontal Modular Storage System for Irradiated Nuclear Fuel (NUHOMS-24P), Rev. 1A, dated July, 1989, NUH-002.
- "SCALE-3: A Modular Code System for Performing Standardized Computer Analysis for Licensing Evaluation." NUREG CR-0200, ORNL, Revision 3, December 1984.
- 3. Oconee Nuclear Station Final Safety Analysis Report (FSAR).
- "Structural Analysis of Shipping Cask. Effects of Jacket Physical Properties and Curvature and Puncture Resistance", H. A. Nelms, Vol. 3, ORNI. TRM-1313, Oak Ridge National Laboratory, Oak Ridge Tennessee, June, 1968.
- "Design of Structures for Missile Impact." R. B. Linderman, J. V. Rotz, and G. C. K. Yeh, Topical Report BC-TOF-9-A. Bechtel Power Corporation, Revision 2, September 1974.
- "Formulas for Stress and Strain," R. J. Roark and W. C. Young, Fifth Edition, McGraw-Hill, New York, New York, 1975.
- "CASMO-2 A Fuel Assembly Burnup Program," Malte Edenius, et. al., STUDSVIK/NR-81/3, March 1981.
- 8. "CASMO Benchmark Report," M. Edenius, et. al., RT-78, 6293, STUDSVIK, March 1978.
- "CASMO Benchmarking Against Experiments in Racl: Geometries," M. Edenius, et. al., NR-81/61 STUDSVIK, November 1981.

3-23

Chapter 3. Principal Design Criteria

Oconee ISFSI Szfety Analysis Report





CHAPTER 4. STORAGE SYSTEM

The Oconee ISFSI is located within the existing Owner Controlled Area on the Oconee Nuclear Station site. The storage area is located west of the existing intake structure. Figure 4-1 on page 4-5 depicts the site layout in relation to other plant features and defines the onsite route that the transfer cask and trailer will travel in moving loaded storage canisters from the Fuel Buildings to the ISFSI.

The Oconee ISFSI utilizes the NUHOMS-24P storage system which provides for the horizontal, dry storage of irradiated nuclear fuel assemblies. The fuel assemblies are contained in a DSC made of stainless steel which is placed inside a reinforced concrete HSM for long term storage.

In addition to the DSC and HSM, the NUHOMS-24P system utilizes handling and transfer equipment to load the DSC with fuel, to seal the DSC, to move the loaded DSC inside a heavily shielded transfer cask from the Fuel Building to the HSM and to insert the DSC into the HSM. The DSC is designed to hold 24 PWR fuel assemblies. The margins of safety in the structural design of the HSM, DSC, and transfer cask are more fully described in Section 8. Tables 8.1 and 8.2 of Reference 1 on page 4-37. Additional information for the handling and transfer equipment is presented in Section 4.8.5, "Transfer Components" on page 4-32.

The fuel assemblies to be stored are described in Section 3.1.1, "Material to be Stored" on page 3-3. The dose to the general public from the operation of the ISFSI is far below the allowable dose limits as set by regulation. Estimates of the annual dose rates are provided in Section 7.7, "Estimated Off-Site Collective Doses" on page 7-33.

It should be noted that the Oconee ISFSI is licensed for the storage of as many as 2112 assemblies; this storage capacity will be added incrementally as needed to support the actual -fueling schedules. HSMs and foundation have been designed to be built in any array size no smaller than 2x3 (three modules side by side and back to back with three additional modules) and no larger than 2x10 (Ten modules side by side and back to back with ten additional modules).

The ISFSI system is designed to interface with existing plant equipment and systems. Roadways, buried pipes, trenches, and positioning aprons were verified to be acceptable for the wheel loadings of the transfer vehicle. Oconee Nuclear Station asphalt roadways were verified as meeting the design minimum thickness requirements of the American Association of State Highway and Transportation Officials as specified for loading comparable to the ISFSI transfer vehicle. Approximately 64 buried pipes and over 26 drain lines were analyzed and verified acceptable according to the applicable codes for each piping materis! All interfacing trench systems were analyzed for transfer vehicle loadings. These include the 115 KV, 230 KV, 525 KV, Radwaste and Interim Radwaste, and the Standby Shutdown Facility trenches. Necessary modifications to these trenches will be completed prior to the transit of the ISFSI transfer vehicle.

The size and weight of the transfer cask, DSC, and crane hook lift adapter are acceptable within the current design limits of the crane, cask handling area, and transfer cask positioning aprons of the spent fuel pools. Design features employed to with tand environmental and accident forces are detailed in Chapter 3. "Principal Design Criteria" on pay this SAR. The DSC and transfer cask are immodeled and transfe

The HSM is designed in accordance with the requirements of ACI 349-85 as discussed in Section 3.2.5.1 of Reference 1 on page 4-37. The HSM is constructed following the guidelines of ACI 318-83 as discussed in Section 4.2.1 of Reference 1 on page 4-37. The DSC and transfer cask are designed and built in

Chapter 4. Storage System



4.1 LOCATION AND LAYOUT

The location and layout of the storage site with respect to other site structures is shown in Figure 4-1 on page 4-5. This figure also denotes the transportation route for movement of the DSCs from the spent fuel pool to the HSMs.

If, during the transfer of a DSC from the fuel building to a HSM, an event requiring return to the fuel handling building occurred inside the Oconee Nuclear Station protected area fence, the tractor-trailer could either continue on around the east side of the Turbine Building and return to the fuel building or, if it is close to the fuel building, it could reverse to return.

From the point where it leaves the Oconee Nuclear Station protected area until the point where it reaches the ramp leading up to the ISFSI, the tractor-trailer has sufficient space to turn around as needed.

Once it is on the access ramp leading to the ISFSI, the tractor-trailer would have to continue to the ISFSI site in order to turn around.

The transport route has been reviewed and found to be within the design basis of the cask drop analysis discussed in Section 8.2 of Reference 1 on page 4-37. The potential causes for cask and DSC drop accidents are described in Section 8.2.5.1 of Reference 1 on page 4-37. The enveloping postulated drop events assumed for design are:

- 1. A horizontal side drop or slap down from a height of 80 inches.
- 2. A vertical end drop from a height of 80 inches onto the top or bottom of the transfer cask.
- 3. A corner drop from a height of 80 inches at an angle of 30° to the horizontal, onto the top or bottom corner of the transfer cask.

These drop scenarios were used to define an equivalent static deceleration load of 75g for cases (a) and (b) and 25g for the corner drop (case (c)). As described in Reference 1 on page 4-37, these deceleration values were developed for assumed surface conditions which will envelop all Oconee site conditions which may be encountered. Specifically, these decelerations are based on the work contained in EPRI report NP-4830 and are applicable to impacted surfaces with target hardness numbers up to 400,000. The maximum target hardness along the Oconee transfer route is 2750 for an edge drop scenario.

The transfer cask route from the fuel buildings to the HSM was evaluated to ensure that the maximum transfer cask drop height of 80 inches is not exceeded. The nominal travel height of the transfer trailer deck is 41 inches which corresponds to a cask drop height of 59 inches. During transit from the fuel building to the HSM site, the trailer deck will be automatically leveled by the trailer's hydraulic suspension units. The maximum design travel for these units can raise the trailer deck height to 52 inches, which corresponds to a drop height of 70 inches. Mechanical stops attached to each suspension unit cylinder ensure that the cask cannot be lifted to a height greater than 70 inches above the ground. Therefore, since the Oconee target hardness and maximum potential cask drop height are less than the values presented in Reference 1 on page 4-37, the deceleration values presented in Reference 1 on page 4-37 envelop all Oconee site conditions.

The site area will be sloped appropriately to permit surface drainage to collection ditches for channelling the water away from the site. As noted in Section 2.4, "Hydrologic Engineering" on page 2-27, the site is 11.9 feet above the probable maximum flood elevation. Local intense rainfall is not a problem since the inlet air opening is 24 in. above yard grade. There is a small drainage pipe passing through the HSM front

4.1 Location and Layout

wall into the plenum area. The base slab of the plenum area is sloped towards this drainage pipe. Additionally, the concrete approach apron is sloped away from the HSM front wall. During a local intense rain, it is remotely possible that some rainwater may backup into the HSM plenum area temporarily, but this water will drain out of the HSM soon after the intense rain subsides. Therefore, due to surface drainage, rain water will not collect to a depth of any concern.



4-4

4.1.1 FIGURES



Figure 4-1. Site Layout and Route



4-5



4.2 STORAGE SITE

The design bases covering the analysis and design procedures for the appropriate loadings are specified in Chapter 3, "Principal Design Criteria" on page 3-1 of this report and in Reference 1 on page 4-37 for the HSM, DSC, transfer cask and transfer trailer. The foundation design includes allowances to the control of the ground accelerations are from the site ground motions specified in Chapter 2, "Site Characteristics" on page 2-1. Liquefaction potential for the ISFSI site is discussed in Soction 2.5, "Geology and Seismology" on page 2-43. Based on the soil investigations and using an equivalent static methodology to account for dynamic effects, spring stiffeners are determined representing the force-deflection relationship of the underlying soil. This information is utilized as input to the structural model described in Reference 1 on page 4-37 to determine settlement effects. See Section 2.5.4, "Subsurface Materials" on page 2-50 for details of the foundation analysis. Temporary loadings from the extreme environmental cases (Chapter 3, "Principal Design Criteria" on page 3-1) and accident conditions (Chapter 8, "Accident Analyses" on page 8-1) have been reviewed and are acceptable.

4.2.1 STRUCTURES

The HSM design bases, materials of construction, codes and standards, etc. are fully described in Reference 1 on page 4-37. The HSM foundation requirements are discussed in Section 2.5.5, "ISFSI Foundation" on page 2-51. The concrete approach aprons will not be attached to the HSM but will be separated by an expansion joint. Differential settlement between the slab and the HSM is not anticipated to be a problem.

The approach aprons are sized for bearing pressures using the same allowable and ultimate pressures as used for the HSM as discussed in Section 2.5.5, "IsFSI Foundation" on page 2-51. Settlement of the approach aprons will be minimal since they are normally unloaded. In addition, the transfer trailer has jacks used in vertically positioning the cask for DSC insertion into the HSM. The trailer leveling procedure will compensate for any differential settlement that may occur between the ¹³SM and the concrete approach aprons. Outlying areas are concrete or asphalt to provide the space required for transfer trailer maneuvers.

4.2.2 STORAGE SITE LAYOUT

Figure 4-2 on page 4-9 depicts the site layout and its functional features.

4.2.3 HSM DESCRIPTION

The HSM is constructed of reinforced concrete and structural steel. The HSMs are placed in service on a load bearing foundation which is within a fenced, controlled access area.

The HSM provides the structural support for the DSC as well as protection against tornado missiles plus neutron and gamma shielding. The exterior walls form an array of modules and the front and roof of the modules are sufficiently thick to provide average surface doses that are below 20 mr hr.

The HSM provides fuel cooling by a combination of radiation, conduction and convection. Natural circulation air flow enters at the bottom of the HSM and passes around the DSC and exits through the flow channels in the top shield slab.



4.2 Storage Site

The design of the HSM system includes consideration of both normal and off-normal operating conditions including a range of credible and hypothetical accidents. The HSM design and analysis were performed in accordance with Chapter 3. "Principal Design Criteria" on page 3-1 and Chapter 8. "Accident Analyses" on page 8-1 of this SAR and Reference 1 on page 4-37.

The design criteria for the operational and accident conditions fall into three main categories; structural, nuclear and thermal-hydraulic. Reference 1 on page 4-37 describes in detail the analysis ... these accidents. The bounding structural loading combinations include thermal, earthquake, to natio and cask drop accidents. For the nuclear analyses, shielding of the DSC end shield plugs, the HSM walls and penetrations, and the criticality analyses were primary considerations. The thermal-hydraulic criteria assures adequate air flow inside the module, acceptable air and concrete temperatures as well as DSC and fuel clad temperatures.

4.2.4 INSTRUMENTATION SYSTEM DESCRIPTION

The Oconee ISFS1 is designed to maintain a safe and secure, long-term containment and storage environment for IFAs using only totally passive components. Therefore, no safety related instrumentation is required for operation of the facility.

Instrumentation is necessary to perform the DSC transfer cask draining purging, and drying operations. This instrumentation consilies of commercial grade pressure gauges. The functions served by pressure instrumentation in the DSC loading procedure are discussed in Chapter 5, "Storage System Operations" on page 5-1 of this SAR.

Radiation monitoring is provided by existing station area, and process effluent monitors. The station radiation monitoring system is described in Oconee FSAR (Chapter 11, "Radioactive Waste Management" on page 11-1).



4-8

4.2 Storage Site

4.2.5 FIGL A.S



Figure 4-" Site Plan





n

Ċ

1

.



a

4.3 TRALSFER SYSTEM

4.3.1 FUNCTION

The function of the transfer system is to transfer the DSC containing irradiated fust assemblies to and from the HSM.

4.3.2 COMPONENTS

The transport system consists of the transfer cask, DSC, transfer cask skid, transfer trailer skid positiciting system and hydraulic ram.

4.3.2.1 Transfer Cask

The transfer cask is used to transfer the loaded DSC to and from the HSM. The cask provides shielding along the axial length of the fuel during transfer, loading and retrieval operations. A description of the transfer over is provided in Reference 1 on page 4-37. For Oconee, two (2) hard six fac. does along to the embance cask sliding characteristics and the liquid neutron which described in Reference 1 on page 4-37, has been replaced by a solid neutron shield comprised of Risco NS-3 which is a cementatious material cast in place in the neutron shield jacket. The drain and fill ports, as well as the expansion tank, which are needed for the liquid neutron shield have been deleted. To ensure that offgassing or vapor expansion will not result in overpressurization of the neutron shield jacket, pressure relief valves are to 45 psig have been added. This change results in a more passive routron shield in that operational and maintenance considerations are reduced. Also, the possibility of a comprise loss of neutron shielding as a result of an accident, although it is still assumed that substantial degradat: in may occur in some localized area.

This subsitution satisfies the requirements of 16CFR 72 because:

- 1. The surface dose rates still satisfy the requirements established in Reference 1 on page 4-37.
- 2. The temperature of the fuel cladding does not exceed the crit. in established in Reference 1 on page 4-37.
- 3. The material characteristics are suitable for the service environment.
- The consequences of postulated accidents are enveloped by the criteria establishe in P.eference 1 on page 4-37.
- 5. The structural integrity of the transfer cask and 'DSC is not compromise'.

4.3.2.2 Dry Storage Canister (DSC)

The DSC provides the primary confinement for up to 24 irradiated fuel assemblies. The DSC provides shielding at the ends and also maintains the fuel array subcritical under the v c rst case conditions. The DSC fits inside the transfer cask for safe movement from the spent fuel pool to the ISFSI site.



REV: (30 JUNE 1991)

4.3 Transfer System



4.3.2.3 Transfer Cask Skid

The purpose of the transfer cask skid is to provide a support base for rotating the transfer cask to a horizontal position and to maintain the transfer cask in the properly aligned position during transport, loading and retrieval operations.

The basic dimensions and layout for the transfer cask skid are presented in Section 1.3.1.5 and Figure 1.3-4 of Reference 1 on page 4-37.

The main load carrying longitudinal skid members are 12WF210 wide flange sections with stiffeners added as required for the design loads. The main load carrying cross members and vertical supports for the upper and lower trunnion pillow blocks are built up steel box sections.

As shown on Figure 1.3-4 of Reference 1 on page 4-37, the transfer cask is secured to the transfer cask skid by the use of bolted top and bottom cask trunnion pillow blocks. The skid is bolted to the transfer cask trailer using the locking brackets shown on the figure.

The transfer cask skid is a non-safety related item which is designed in accordance with the requirements of the AISC code, eighth edition using linear elastic analytical methods and normal allowables for the bounding design basis loading. The design loads for the transfer skid and attachmet's are the same as the transfer cask trunnion loads presented in Section C.1-4 and Figure C.1-2 of Appendix C of Reference 1 on page 4-37.

The design basis loads for the transfer cask skid were conservatively established to envelope all applied loads including downending of the cask, rotational loads, and transport loads during transfer to the ISFSI site. The transfer skid design loads envelope the postulated off-normal and accident loads discussed in Section 8.2 of Reference 1 on page 4-37 such as earthquake and tornado wind loads. Along with the basic Code allowable stresses used in the design analysis, this conservative design basis assures that the skid will adequately support the NUHOMS-24P transfer cask for all postulated events.

4.3.2.4 Transfer Trailer

The transfer trailer has a capacity of 120 tons. The transfer trailer is designed to ride as low to the ground as possible to minimize the HSM height. Four hydraulic jacks are incorporated into the trailer design to provide vertical movement for alignment of the transfer cask with the HSM. The trailer is pulled to the ISFSI by a conventional tractor.

The trailer is configured as a 4x2 hydraulically steered dolly. Eight hydraulic suspensions carry four pneumatic tires each and are located two wide, in four axle lines. There are a total of 32 tires.

Hydraulic suspensions enable coupled steering of all axles around a common point, thus minimizing tire scuffing and the resulting damage to pavement and tires. The suspension, also allow other advantages, such as adjustable deck height, lockout or repair of failed suspensions or tires, and compensation for road surface irregularities.

The trailer is pulled by a conventional tractor via a drawbar unit. The drawbar unit includes hydraulic master cylinders that provide motive force for the slave steering cylinders in the trailer.

Additional features and accessories for the trailer include: diesel power pack and hydraulic control valves, hand held remote control unit, all-wheel braking, and provisions for mounting four bearing pads, hydraulic alignment system hardware, and four hydraulic lifting jacks to the trailer frame.



The trailer is a commercial grade item of the type commonly used to move heavy loads, such as the space shuttle. The design parameters for a typical trailer are provided in Table 4-1 on page 4-19. It is constructed in accordance with the manufacturer's standard QA program using the following codes of construction:

- American Institute of Steel Construction (AISC) Specification for the Design. Fabrication, and Erection of Structural Steel for Buildings
- · American Society of Testing and Materials (ASTM)
- * American Welding Society (AWS) AWS D1.1 Structural Welding Code-Steel
- + Steel Structures Painting Council (SSPC)

4.3.2.5 Skid Positioning System

The Skid Positioning System (SPS) includes the following items which either actively or passively position the skid during storage, transport, and alignment operations, low friction bearing plates, skid tie down brackets, hydraulic lifting jacks, hydraulic x-y-theta positioning cylinders, and all associated instrumentation and controls. Controls for the SPS are located on a control skid which is located several feet from the cask.

The loaded cask is supported by a steel skid structure. The skid's weight is supported by a set of four low friction bearing plates. The bearing material offers a coefficient of friction of 5% or less with negligible breakaway friction. The skid is restrained from lateral motion during transport and storage by a set of tie down brackets which are attached to the trailer frame.

Four support plates for hydraulic lifting jacks are located on the trailer frame. Although the trailer's hydraulic suspensions could be used to perform trailer deck height adjustments, the jacks will more firmly support the load than pneumatic tires. The jacks provide elevation adjustment, plus adjustment of pitch and roll of the trailer frame relative to the concrete HSM pad. The jacks are also used in the fuel building during cask loading. There, the front pair of jacks carries most of the load during the cask setdown and downending.

A system of hydraulic actuators are oriented in the transverse and longitudinal directions on the trailer deck. These cylinders are used to align the cask correctly relative to the HSM after the deck is leveled at the appropriate height.

4.3.2.6 Hydraulic Ram System (HRS) Description

Reference 1 on page 4-37 includes a system description of the hydraulic ram in Section 1.3.1.6, a system operation description of loading and retrieval of the DSC in Section 1.3.1.7 and a functional description in Section 5.2.1.1. Figure 1.3-5 shows a typical design for the hydraulic ram system. Figure 1.3-6 shows the primary operations for the NUHOMS system.

The operations system for loading and unloading of the DSC into and out of the HSM is discussed in Sections 5.1.1.6 and 5.1.1.8 of Reference 1 on page 4-3?. Figure 5.1-4 of the same reference shows the NUHOMS System retrieval operations flow chart. Safety features of the ram system are presented in Section 5.2.1.2 of Reference 1 on page 4-37.

The HRS consists of the following main components: one double-acting hydraulic cylinder (ram): one trailer-mounted tripod support assembly for rear support and alignment of the ram hydraulic cylinder: one ram support frame assembly for front support and alignment of the ram hydraulic cylinder: one grapple

2

4.3 Transfer System

assembly; one hydraulic power unit with controls: and hydraul- tubing, hoses, hose reel and accessories as required for system operation.

4.3.3 DESIGN BASES AND SAFETY ASSURANCE

4.3.3.1 Transfer Cask

The design bases of the transfer cask are given in Section 1.2.2 of Reference 1 on page 4-37. These are based primarily on radiological and structural considerations.

As discussed in Section 4.3.2.1, "Transfer Cask" on page 4-11, the solid neutron shield will be permanently filled with Bisco NS-3 - a neutron absorbing cementatious material cast in place in the neutron shield jacket. Pressure relief valves are designed to relieve pressure in the event that any off-gassing were to create excessive internal pressure.

4.3.3.2 Transfer Cask Skid

The transfer cask skid supports the transfer cask i.: a horizontal position on the trailer deck during the on site road transportation to the ISFSI site. The transfer cask skid is designed to support a transfer cask weighing 110 tons and to allow rotation of the transfer cask between the horizontal and vertical positions. The transfer cask skid is secured to the transfer trailer during movement and is restrained by a securing system to resist the peak loads anticipated under normal conditions of transport between the fuel buildings and the ISFSI.

4.3.3.3 Transfer Trailer

The design parameters for the transfer trailer are presensed in Table 4-1 on page 4-19. Also, as shown in Section 8.2.5 of Reference 4.1, "Location and Layout" on page 4-3, the design basis drop height for the NUHOMS-24P Transfer cask is 80 inches. This analysis bound: the Oconec transport conditions. The nominal travel height of the transfer trailer deck is 41 inches which corresponds to a cask drop height of 59 inches. The maximum design travel of these units can raise the trailer deck height to 52 inches, which corresponds to a drop height of 76 inches. Mechanical stops attached to each suspension unit cylinder ensure that the cask cannot be lifted to a height greater than 70 inches above the ground.

If an event requiring return to the fuel handling building occurred inside the Oconee Nuclear Station protected are: ance, the tractor-trailer could either continue on around the east side of the Turbine Building and return to the fuel building or, if it is close to the fuel handling building, it could reverse to return.

From the point where it leaves the Oconce Nuclear Station protected area until the point where it reaches the ramp leading up to the ISFSI, the tractor- trailer has sufficient space to turn around as needed.

Once it is on the access ramp leading to the ISFSI, the tractor-trailer would have to continue to the ISFSI site in order to turn around.

4.3.3.4 Skid Positioning System (SPS)

The SPS is designed to compensate for the following variance in true alignment between the cask and HSM, in any combination.





8	Pure Sideways Translation	3.
×	Pitch	$1/4^{\mu} \ / \ ft$
	Yaw	3 degrees

In addition to the above corrections, the SPS must move the cask and skid from the transport position, in which the payload's center of gravity lies directly over the centroid of the trailer, to a position where the cask slightly overhangs the rear of the trailer. The required actuator strokes to achieve the design basis a suppensations are (in terms of pure directional motion) approximately:

*	Vertical Travel	6°
*	Transverse Travel	10*
	Longitudinal Travel	39*

The SPS components which restrain the cask and skid during cask setdown and transport are designed to withstand the loads described for the cask trunnions in Appendix C.1 of Reference 1 on page 4-37. The design basis weights for use in sizing SPS actuators and hardware are, in U.S. tons:

÷	Empty Cask	56 tons
÷	Loaded DSC	38 tens
÷	Skid	6 tons
*	Trailer	20 tons

The SPS will be designed and built to the following codes and specifications:

- American Institute of Steel Construction (AISC) Specification for the Design, Fabrication, and Erection of Structural Steel for Buildings
- · American Society of Testing and Materials (ASTM)
- * American Welding Society (AWS) AWS D1.1 Structural Welding Code-Steel
- National Fluid Power Association (NFPA)
- Steel Structures Painting Council (SSPC)

The SPS is specified for use in the following environmental conditions:

*	Ambient	Storage Temperature	+30°F to 116°F
	Ambient	Operating Temperature	$0^\circ F$ to $110^\circ F$
*	Ambient	Humidity	10% to 100%
	Ambient	Radiation	Negligible

The SPS is designed with several safety features to avoid unnecessary delays in the transfer process. The trailer lifting jacks have mechanical locking collars which preclude settling of the trailer deck, due to loss of hydraulic pressure to the jack cylinders. A hand pump provides redundant power as a backup in the event of pump or power unit failure. Operation of the trailer jacks, transverse, and longitudinal cylinders are mutually exclusive; it is impossible to operate more than one sub-system at a time. During alignment

4.3 Tr insfer System

and fuel transfer, the skid tie down brackets are unbolted, and the x-y travel of the skid is limited by both the hydraulic cylinder travel and by mechanical restraints on the low friction bearing travel. A tie down between the HSM and cask provides additional restraint during fuel transfer.

4.3.3.5 Hydraulic Ram

HYDRAULIC RAM SYSTEM (HRS) PERFORMANCE REQUIREMENTS:

- Ram Force 20,000 lb., Push and pull (Nominal) 80,000 lb., Push and Pull (Maximum)
- * Ram Piston Speed 36 in/min (max)
- · Stroke · Approximately 21 feet

CODES AND STANDARDS:

The HRS components are not safety related and are designed to conform to standard industrial requirement. The HRS design conforms to the following codes and standards:

- Ram C-finder National Fluid Power Association (NFPA) Standards Recommended Standards and Ame prioral Standards Institute (ANSI) Standards;
- * Ram ... annions Clevis NFPA and American Institute of Steel Construction (AISC) Standards;
- · Hydraulic Power Unit NFPA Recommended Standards;
- * Grapple Assembly AISC:
- Trailer-Mounted Tripod Support Design · AISC;
- Welding of Tripod Support and Rain Support Frame AWS Structural Welding Code and American Society for Nondestructive Testing. Inc. Recommended Practices;
- * Pump Motor National Electrical Manufacturers Association (NEMA) Standards and Publications;
- Instrumentation and Controls Instrument Society of America (ISA) Standards and the International Organization for Standardization (ISO) Standards;
- Component Materials American Society for Testing and Materials (ASTM) Standards and ANS1 Standards;
- Corrosion Protection Steel Structures Painting Council (SSPC) Specifications and National Association of Corrosion Engineers (NACE) Recommended Practices and Material Requirements.

DESIGN LOAD CONDITIONS:

All system load bearing components of the HRS are designed to withstand the stresses associated with a maximum column load of 80,000 pounds at full extension centered 4.5 inches above the longitudinal axis of the ram cylinder. The system load bearing components include the ram hydraulic cylinder, grapple assembly and ram support frame assembly.

The trailer-mounted tripod support for the ram hydraulic cylinder is designed per American Institute of Steel Construction (AISC) Manual of Steel Construction Standards.

ENVIRONMENTAL CONDITIONS:

The HRS is designed to withstand the following environmental conditions:

- Ambient Storage Temperature Range: -30°F to 116°F
- * Ambient Humidity Range: 10 to 100% (coincident with outdoor temperature range)
- * Radiation Dose Rate (Section 10.3.4.1 of Reference 1 on page 4-37): 250 mrem/hr
- Ram force is limited to 20,000 pounds during the extension and retraction strokes for normal
 operation.
- * Ram force 's limited to no more than 80,000 pounds under all conditions.
- · Ram extension and retraction speeds are variable from 0 to 36 inches per minutes.

INSTRUMENTATION AND CONTROLS:

The HRS is designed to allow the operator to extend and retract the rara hydraulic cylinder using hand-operated devices. The control system includes safety features to prevent the inadvertent operation of the HRS and to regulate speeds and forces of the rarm to within the design criteria limitations.

COMPONENT DESCRIPTIONS:

Ram Hydraulie Cylinder - The ram hydraulie cylinder is a single stage, double-acting, horizontal design. The maximum extension or retraction force is 80,000 pounds at the maximum extension and retraction fluid pressures. The maximum piston speed for extension or retraction is 36 inches per minute (ipm). The cylinder is designed with trunnion front mounts and a rear clevis mount. The trunnions are designed to fit into the ram support frame. The rear clevis fits the ram tripod frame.

Ram Hydraulic Pump - The hydraulic pump is a positive displacement type pump.

Grapple Cylinder and Pump - The grapple is operated with a small, double acting hydraulic cylinder. The cylinder is manually operated with a hydraulic hand pump.

Reservoir - The HRS includes a 140 gallon reservoir sized to meet the system capacity requirements.

Instrumentation and Controls - The control system is designed to allow the operator to extend or retract the ram hydraulic cylinder using manually actuated pressure and flow control devices.

Grapple Assembly - The grapple assembly is depicted by Revision 1 of the NUHOMS-24P Topical Report Figure 1.3-5.

The power for the hydraulic system is provided from a retail 24 KV Oconee support line through a 75 KVA, 3 phase transformer.

Although this system does not have a backup power source, the retail power provided is considered very reliable. However, in the event of a power failure - whether momentary or extended - all efforts to transfer the DSC into the HSM will be halted until power is restored. In the interim period, the hydraulic system will be secured in the "off" position and all personnel will leave the immediate area of the cask. At any point in the transfer process, the HSM, the transfer cask, or a combination of both will provide sufficient shielding to maintain dose rates at acceptable investories during such a loss of power.

4.3.3.6 Other Equipment

All equipment other than the HSM DSC, and transfer cask used in the transfer operations is classified as non-safety related. Equipment u'_{\perp} for loading fuel into the DSC and transfer of the DSC/transfer cask within the fuel building is safety clated and is covered by the Oconee 10CFR Part 50 license.

4.3 Transfer System

The failure of any non-safety related piece of transfer equipment described in Section 1.3 of Reference 1 on page 4-37 will not increase human radiation exposures by any significant amount. As described, the transfer trailer has 32 wheels. The route from the fuel building to the ISFSI site is approximately 1/2 mile. The trailer and all its components are carefully inspected prior to use, and the probability of a breakdown is small. In the event of a component failure, the trailer can be configured to overcome failure of a wheel or suspension unit and off-loading can be completed prior to repairs. A failure in the system hydraclics could be repaired or the trailer pulled to the HSM site and the DSC off-loaded. Because of this design simplicity, failure of the hydraulic ram will be limited to the hydraulic control system. Such a failure would be casy to repair and because the hydraulic controls are located at least 25 feet from the transfer cask, and therefore additional human radiation exposure during repairs would be minimal.

4.3.3.7 Qualification of Components

Qualification of the hydraulic ram system (HRS) ⁴ skid positioning system (SPS) was done per standard administrative procedures and check out tesung for operation of non-safety related equipment. The qualification tests consisted of pre-operational system checkout tests. All phases of the HRS and SPS operation were tested to verify the operability of the system. Normal operation and off-normal events and the respective recovery procedures were confirmed. All system performance criteria were verified to be me.

The HRS and SPS have simple, reliable designs which require minimal maintenance on active components and negligible maintenance on passive components. Primary maintenance requirements consist of periodic inspections of the hydraulic power units, ram hydraulic cylinder, grapple assembly, SPS actuator assemblies, and manual controls. In addition, the hydraulic fluid actuates periodic testing to ensure that no water, dirt or other deleterious materials have contaminated the steem.

4.3.3.8 Maintenance of HRS and SPS

Maintenance requirements for the HRS and SPS are minimized by corrosion protection provided by component design. All components are manufactured from corrosion resistant materials, or coated with corrosion resistant paints, and/or stored ap^A operated with a grease or oil surface protectant. All controls and instrumentation which are subject to corrosion are housed in a weatherproof enclosure. The ram hydraulic cylinder is stored in its retracted position, filled with hydraulic fluid.

Operating procedures, maintenance procedures and storage procedures will ensure that all HRS and SPS components are kept in operable condition throughout the system design life.



4.3.4 TABLES

Table 4-1. ONS ISFSI Project Transfer Trailer Design Parameters		
Ambient Storage Temperature Ambient Operating Temperature Ambient Humidity Ambient Radiation Pressure Altitude	-30°F to 116°F 0°F to 110°F 10% to 100% Negligible 0' to 5000' el.	
Payload (Cask + Skid)	120 tons	
Minimum Deck Height Maximum Deck Height	34* 52*	
Maximum Deck + Steering Unit Length Maximum Deck Length Maximum Width	25'-0* 21'-1* 12'-0*	
Inside Turn Radius Outside Turn Radius	9' or less 27' or less	
Maximum Pulling Speed (Laden) Maximum Grade	5 mph 6.5%	
Road Surface: (Fully Laden) (Empty Cask)	Asphalt Packed Gravel or Asphalt	



4-19



4



ŧ.

4.4 OPERATING SYSTEMS

4.4.1 LOADING AND UNLOADING SYSTEM

Loading and unloading of IFAs from the DSC and transfer cask requires use at the 'cllowing equipment'

- 100 ton spent fuel cask handling crane
- · spent fuel pool manipulator crane auxiliary hoist
- transfer cask lift beam
- · DSC lifting rig
- crane hook lift adapter
- · cask pit platform

4.4.1.1 Preparation for Fuel Loading

Following receipt inspection and acceptance, a DSC is placed in the transfer cask. The orientation of the DSC in the cask is controlled by permanent alignment marks on each DSC and the transfer cask. The DSC is filled with borated water with a minimum concentration of 1810 ppm boron. The transfer cask is then positioned in the decontamination pit. The DSC/transfer cask annulus is filled with demineralized water and sealed with an inflatable seal. The transfer cask is then placed on the cask pit platform in the spent fuel pool.

The following components are used for this operation:

- <u>100 Ton Crane</u> the 100 ton spent fuel cask handling crane is used to place the DSC into the transfer cask and to move the DSC transfer cask to the spent fuel pool. The 100 ton crane is described in Section 9.1.4.2.2, "Loading and Removing Fuel" on page 9-15 of the Oconee FSAR (Reference 2 on page 4-37).
- DSC Lift Rig The DSC lift ig is bolted to two of the four lifting lugs attached to the support ring for the top lead shield plug inside the top of the DSC. It is used for upending the DSC prior to loading into the transfer cask.
- 3. <u>Transfer Cask Lift Beam</u> The transfer cask lift beam adapts the transfer cask to the 100 ton crane hook. It is used during upending and transport of the transfer cask within the fuel building. The transfer cask lift beam is designed, built, and maintained in accordance with the criteria of ANSI N14.6. The lift beam is a passive, open hook design with two parallel lifting beams. It is fabricated from high strength carbon steel plate and is protected by a decomtaminable coating. Figure 4-3 on page 4-26 depicts the transfer cask lift beam.
- 4. <u>Crane Hook Lift Adaptor</u> After the DSC/transfer cask is placed on the cask pit platform, the crane hook lift adaptor is attached between the 100 ton crane hook and the transfer/cask lift beam. The crane hook lift adaptor is designed to prevent wetting the 100 ton crane hook and block when the DSC/transfer cask is lowered from the cask pit platform into the cask pit. The crane hook lift adaptor is designed, built, and maintained in accordance with the criteria of ANSI N14.6. Like the lift beam, it is fabricated from high strength carbon steel plate and is protected by a decontaminable coating. The adaptor has an elongated pin hole (48 inches) and a screw actuator at the lift beam end. For lifting the transfer cask, the adaptor is in the elongated configuration with the lift beam pin supported by the

1

4.4 Operating Systems

bottom of the pin hole. When disengaged from the cask, the adaptor may be placed in the retracted configuration by means of the scr w actuator, which provides support for the lift beam while in this configuration. The retracted configuration is required for the combined adaptor and lift beam to clear the spent fuel pool operating deck. Figure 4-4 on page 4-27 depicts depicts the crane hook lift adaptor.

5. <u>Cask Loading Pit Insert</u> - A removable platform approximately 18 inches in height is placed in the spent fuel pool cask pit. Its functions are to allow release of the transfer cask at an elevation that prevents the 100 ton crane block from contacting spent fuel pool water and to position the cask so that spent fuel can be loaded into the DSC.

4.4.1.2 Spent Fuel Selection

A description of the administrative procedures which are followed in spent fuel identification is presented in Section 10.2.5, "Administrative Controls" on page 16-6. Using special nuclear material control and accountability records, the initial enrichment and burnup for each candidate spent fuel assembly are compared against the acceptable region in Figure 10-1 on page 10-10. Fuel assemblies falling in the acceptable regions will have qualifying reactivity and decay heat characteristics for safe storage in the NUHOMS-24P System. If all requirements for spent fuel qualification are met, then documentation of this fact for a given assembly is transmitted to fuel handling personnel prior to assembly retrieval and placement in the DSC. Based upon station maps and special nuclear materials accountability records which indicate the current location of these assemblies, fuel handling personnel visually verify the assembly identification numbers and transfer these assemblies into the DSC. An independent visual verification (using binoculars or CCTV) of the assembly serial number by two different persons is performed prior to assembly retrieval from the spent fuel pool. After all assemblies have been loaded into the DSC, the assembly identification numbers are again checked. In the event that these assemblies must subsequently be retrieved in the future from the HSM DSC and inserted back into the spent fuel pool, similar accountability/verifi/ tion procedures will be used.

No fuel will be loaded into the DSC which is known to have any gross structural damage. Duke's damaged fuel assembly and component database contains a record of confirmed and suspect fuel assembly cladding and other structural failures. Assemblies which are suspected of having cladding failure are further examined visually (using cameras) to determine the extent of the damage. Of these assemblies, only those showing gross cladding or structural damage will be excluded. This inspection is performed after verification of the assembly identification number. Fuel assemblies which have no record of cladding damage are not inspected in detail; they are observed during the routine fuel handling transfer operation to ensure that the structural integrity of the assembly is maintained.

No fuel assembly cleaning or crud removal operations are planned on initial loadings or retrieval. These operations are not necessary for storage in the NUHOMS-24P system and would likely increase personnel exposures during fuel loading. The DSC will provide full containment of all radioactive crud materials which are dislodged during the handling and/or storage operations.

4.4.1.3 Spent Fuel Loading

The layout of the spent fuel pool area is shown in Figure 4-5 on page 4-28 through Figure 4-7 on page 4-30. After the DSC transfer cask is lowered from the cask pit platform onto the cask loading pit insert. IFAs which have been qualified are loaded into the DSC. The components and equipment used for this operation are described below:

1. Spent Fuel Pool Manipulator Crane - The spent fuel pool manipulator crane mast or its monorail hoist is used to extract 1. As from their pool storage cells and to lower them into the DSC. The spent fuel pool manipulator crane is described in Section 9.1.4.2.2, "Loading and Removing Fuel" on

page 9-15 of the Oconec FSAR. If the monorail hoist is utilized, it is in conjunction with a manual spent fuel handling tool.

 Spent Fuel Handling Tool - The spent fuel handling tool consists of a pneumatically actuated gripper and suspension cables. Its purpose is to provide remote underwater engagement and disengagement of IFAs. This tool has been used at Oconee for loading iFAs into spent fuel shipping casks, and it required no alteration for use with the DSC.

4.4.1.4 DSC Drying, Backfilling, and Sealing

After the IFAs are loaded into the DSC, the top end shield plug is replaced on the DSC. The DSC top end shield plug is suspended by cables from the transfer cask lifting yoke. The 100 ton spent fuel cask handling crane allows fine adjustment of bridge and trelley positions, hook height, as well as rotation of the crane hook. The bottom of the top lead shield plug is charnfered to allow a degree of self-centering by the plug. Two separate paths exist for displacement of DSC cavity water as the shield plug is lowered. A gap exists between the shield plug and the DSC shell, and the DSC vent port is open during installation of the top lead shield plug is recognized as a critical step requiring close attention and gradual movements to assure no misalignment or damage to components. The DSC transfer cask is raised to the cask pit platform by use of the 100 ton crane with the crane hook lift adaptor. As the DSC approaches the surface of the spent fuel pool, the correct placement of the top lead shield plug is verified visually and through dose rate monitoring. On the cask pit platform the crane hook lift adaptor is removed and the crane hook is attached directly to the lift beam in order to provide sufficient lift height during transport of the DSC/transfer cask over the pool deck and into the decontamination pit.

In the decontamination pit the top end shield plug is seal welded, and the water is purged from the DSC. The DSC is then vacuum dried and backfilled with helium. Helium leak tests are performed on the top lead shield plug seal weld and the vent and siphon port seal welds. Finally, the top cover plate is seal welded into place. These oper tions are described in detail in Chapter 5 of Reference 1 on page 4-37.

During the above operation. If we are confined within the DSC with the top end shield plug in place, and the DSC remains seated in the transfer cask. F llowing these operations, the transfer cask lid is placed, the annular water is drained, and the transfer cask is placed on the transfer trailer for transport to the ISFSI.

The design basis and safety assurance features of the DSC are described in Sections 3.2 and 3.3 of Reference 1 on page 4-37. The design basis and safety assurance features of the transfer cask are described in Section 1.3.1.3 of Reference 1 on page 4-37. The DSC drying and sealing equipment and operations the ke use of industry-standard equipment and procedures commonly used for such operations.

4.4.1.5 DSC Unloading

The equipment discussed in Sections 4.4.1.1, "Preparation for Fuel Loading" on page 4-21 and 4.4.1.2, "Spent Fuel Selection" on page 4-22 is used for DSC unloading operations. Appropriate DSC cutting equipment and procedures as discussed in Section 5.1.1.9 of Reference 1 on page 4-37 will be used to open the DSC which is contained within the transfer cask.

4.4.2 DECONTAMINATION SYSTEM

No decontemination facilities are no ded at the ISFSI.

4.4 Operating Systems

Decontamination of the transfer cask is performed in the decontamination pit. The transfer cask exterior is decontaminated manually before removal from the fuel building by use of detergents and wiping cloths.

Also, the DSC top end shield plug is decontaminated in this manner prior to seal welding to the DSC body.

It is not anticipated that either the exterior of the DSC or the inside of the transfer cask will become contaminated. The DSC/transfer cask annulus is filled with demineralized water and sealed with an inflatable seal. However, in the event that such contamination occurs, the DSC/transfer cask annulus will be flushed with demineralized water until an acceptable lev ' is achieved.

4.4.3 DSC REPAIR AND MAINTENANC

No maintenance is required for the DSC for its design life.

4.4.4 TRANSFER CASK REPAIR AND MAINTENANCE

The function of the transfer cask is to ensure integrity of the DSC during applicable design basis accidents and to provide radiological shielding for the operators during handling and transfer operation. Confinement of radioactive materials is provided by the DSC. Accordingly, a periodic maintenance program has been established to ensure the proper operation of the cask valves, bolts, washers, o-rings and neutron shield pressure relief valves. The lifting surfaces of the cask trunnions are periodically inspected for surface deterioration.

4.4.5 UTILITY SUPPLIES AND SYSTEMS

The design of the Oconee ISFSI is based on the NUHOMS-24P system for storage of irradiated fuel. Each module is a self-contained, passive system requiring no support systems during storage.

However, the ISFSI is provided with a 480/208/120 VAC power supply for operation of the transfer trailer hydraulic positioners, hydraulic ram site security equipment and lighting.

Other electrical connections required for ISFSI physical security are described in the ISFSI Physical Security Plan (Reference 3 on page 4-37).

4.4.6 OTHER SYSTEMS

4.4.6.1 Communications and Alarm System

Details of the communication and alarm system are provided in the Physical Security Plan (Reference 3 on page 4-37).

4.4.6.2 Fire Protection System

No flammable or combustible materials are stored within ISFSI or ir its immediate vicinity and the ISFSI is constructed of noncombustible heat-resistant materials (concrete and steel). Therefore, no fixed fire extinguishing system is required; however, portable supression equipment will be provided within the fenced boundary. In the unlikely event of a fire at the ISFSI, the fire brigade will be dispatched from the Oconee Station. The Oconee Nuclear Station Pre-Fire Plan (Reference 4 on page 4-37) will be revised to incorporate fire protection requirements for the ISFSI.



4.4.6.3 Maintenance System

The ISFSI requires no maintenance other than periodic inspection of the HSM air inlets and outlets and removal of debris, if needed. Specific inspection periods and their justification are discussed in Section 10.3.3.1.

4.4.7 FIGURES







4-27

4.4 Operating Systems





Figure 4-5. Spent Fuel Pool Area

4.4 Operating Systems



Figure 4-6. Spent Fuel Pool Area



4-29


Figure 4-7. Spent Fuel Pool Area



4.5 CLASSIFICATION OF STRUCTURES, SYSTEMS, AND COMPONENTS

Table 4-2 on page 4-34 provides a list of major ISFSI components and their classification. Classification of major components as "Safety Related" or "Radwaste Related" is based on the specific need for component performance under accident conditions. However, designation of specific components as "Safety Related" or "Radwaste Related" is not the only basis for establishing whether any part is important to the safe operation of the ISFSI facilities.

As listed in Revision 1 of the NUHOMS-24P Topical Report, Section 3.2. Reference 1 on page 4-37, the NUHOMS reinforced concrete HSM including its foundation and DSC support structure, the DSC and its internal basket assembly, and the transfer cask are components considered important to safety. The design criteria for these components are provided in Section 3.2 and summarized in Tables 3.2-1 through 3.2-9 of Reference 1 on page 4-37.

The Oconee fuel building crane and the lifting beams used for movement of the transfer cask within the fuel building are designed and procured as components important to safety. The lifting beams used in that part of the operation are controlled by 10CFR Part 50 and NUREG-0672 and are designed to ANSI 14.6-1986 criteria for nonredundant yokes.

As noted in Section 4.5.5, "Transfer Components" on page 4-32, all other components of the NUHOMS system, including the transfer trailer and cask support skid, skid positioning system, and hydraulic ram system are required to perform their function to successfully transfer a DSC to and from the HSM. These systems are described in Reference 1 on page 4-37 with design requirements further delineated in Section 4.5.5, "Transfer Components" on page 4-32 of this SAR.

In addition, the transfer cask, HSM, and DSC have been designed to meet very conservative design criteria including postulated conditions which envelop those which may result from the mechanistic failure of the transfer system equipment. Design conditions such as cask drop accident and jammed DSC have been included even though there is no plausible way for these worst case events to occur. Conservative bounding analysis for these conditions have been performed using minimum material yield strengths, strength reduction factors, and factors of safety in accordance with the stringent requirements of the ASME and ACI Codes. Even when applying this conservative criteria considerable design margin for these components and structures remains as evidenced by the analysis results comparisons with acceptance criteria contained in Reference 1 on page 4-37. Further, these components and structures are fabricated and construct d to the rigorous standards and methods of the ASME and ACI Codes under a 10CFR 50 Appendix B Quality Assurance Program. These include material qualification, welding and nondestructive examination, and strict surveillance and quality control inspection. The resulting high integrity of the Transfer Cask, DSC, and HSM provide more than adequate assurance that the health and safety of the public and plant personnel are protected.

4.5.1 TRANSFER CASK

The transfer cask is considered Nuclear Safety Related (QA Condition 1) since it performs primary DSC protection functions under certain transport accident conditions. The transfer cask proposed for use in DSC transfer operations is described in Section 1.3.1.3 of Reference 1 on page 4-37.



4-31

4.5 Classification of Structures, Systems, and Components



The DSC is considered Nuclear Safety Related (QA Condition 1) since it performs criticality control and primary IFA support functions as well as serving as the primary storage containment for the IFAs. It is designed to remain intact under all accident conditions identified in Chapter 8, "Accident Analyses" on page 8-1 of this SAR with no loss of function. DSC components are designed, constructed, and tested in accordance with Nuclear Safety Related requirements as defined by 10CFR 50, Appendix B and the DPC QA-1 Quality Assurance Program.

4.5.3 HORIZONTAL STORAGE MODULE

The HSM functions include shielding, heat removal, DSC support, and DSC tornado missile protection. The HSM is not considered Nuclear Safety Related since it performs no primary IFA containment or criticality control functions. However, HSM functions are considered important to the safe operation of the ISFSI and appropriate levels of documentation and control are applied. The concrete HSM is designed is, accordance with ACI 349-85 and the level of testing, inspection and documentation provided during construction is in accordance with the DPC QA-2 Quality Assurance Program.

As shown by Table 8.1-12 of Reference 1 on page 4-37, the maximum HSM concrete temperature for the long term 70°F ambient storage temperature case in 144°F. This is less than the maximum permissible concrete temperature of 150°F specified by ACI 34º-85. The effect of extreme ambient temperatures will be to increase the maximum concrete temperature. However, the methods used to calculate these temperatures assumed that the extreme ambient temperatures would remain constant until steady state conditions can be established within the HSM. In reality, the extreme ambient temperatures will occur infrequently and will last for a short duration insufficient to cause steady state conditions. Therefore, the long term maximum temperatures for the HSM concrete easily meet the ACI 349 temperature limitations.

Coupled with the conservative reductions in concrete material strength used in the HSM design calculations, the design criteria utilized in Reference 1 on page 4-37 are adequate to ensure that the HSM will perform its intended safety function for all design conditions.

Typical reinforcing steel design for the HSM basemat, walls, and roof is shown in Figure 8.1-9 of Reference 1 on page 4-37. The HSM reinforcing designs are in accordance with the ACI 349-85 Code and are comparable to those previously reviewed and approved by the NRC for the NUHOMS-07P design.

4.5.4 FOUNDATION

The ISFSI foundation is designed, constructed and tested to the same design criteria and quality assurance requirements as the HSM.

4.5.5 TRANSFER COMPONENTS

The remaining DSC transfer components (i.e. transfer cask trailer and skid, skid positioning system, hydraulic ram system) are necessary for the successful loading of the DSC into the HSM. As discussed in Section 4.3, "Transfer System" on page 4-11, failure of these components would not endanger the health and safety of the public or plant personnel. Therefore, transfer components are not considered Nuclear Safety Related and are designed, constructed and tested in accordance with good industry practices.





4.5.6 INSTRUMENTATION

The Oconee ISFSI is designed to maintain a safe and secure, long-term containment and storage environment for IFAs using only totally passive components. Therefore, no Nuclear Safety Related instrumentation is required for operation of the facility. Instrumentation necessary to perform DSC/transfer cask draining, purging and drying operations consists of industrial grade pressure gauges.





4.5 Classification of Structures. Systems, and Components

Oconee ISFSI Safety Analysis Report

0

4.5.7 TABLES

Table 4-2. Oconee ISFSI Major Components and Classification	
Transfer Cask	Safety Related (1)
 Dry Storage Canister (DSC) Basket Spacer Disks Support Rods End Shield Plug/Support (top and bottom) DSC Body End Closure Plates 	Safety Related (2)
 Horizontal Storage Module (HSM) Concrete Shielding DSC Support Assembly 	Radwaste Related (3)
• Foundation	Radwaste Related (3)
 Transfer Components Transfer Trailer/Skid Ram Assembly 	Industrial Grade
Instrumentation	Industrial Grade

- Notes:
- To ensure containment and criticality control under all applicable transport accident conditions, transfer cask components are designed, constructed, and tested in accordance with Nuclear Safety Related requirements as defined by 10CFR 50, Appendix B and the DPC QA-1 Quality Assurance Program.
- To ensure safe and secure, long-term containment and criticality control during transfer and storage of IFAs, DSC components are designed, constructed, and tested in accordance with Nuclear Safety Related requirements as defined by 10CFR 50, Appendix B and the DPC QA-1 Quality Assurance Program.
- 3. Components which are not required to perform a safety function or mitigate the consequences of an accidental radiological release comparable to 10CFR 100 site dose criteria guide values are designed, constructed, and tested in accordance with the DPC QA-2 Quality Assurance Program. Additionally, the concrete HSMs and foundation are designed to withstand Safe Shutdown Earthquake seismic forces and tornado missiles so as to preclude any interaction with the DSC pressure boundary or loss of shielding. Therefore, construction and inspection shall be in accordance with the QA-2 Quality Assurance Program.



4.6 DECOMMISSIONING PLAN

Decommissioning of the ISFSI will be performed consistent with decommissioning of the Oconee Nuclear Station. This is predicated on the ability of the federal government to accept spent fuel at the rates a d dates specified in the Nuclear Waste Policy Act of 1982, as amended. It is anticipated that the DSCs will be transported to a federal repository when such a facility is operational. However, should the slorage facility not accept the DSCs intact, the NUHOMS-24P system allows the DSCs to be brought back into the spent fuel pool and the fuel repositioned into the racks for loading into transport casks to be provided by the Department of Energy.

All components of the NUHOMS-24P system are manufactured of similar materials found in the existing Oconee Station (i.e., reinforced concrete, stainless steel, lead). These components will be decommissioned by the same methods in place to handle similar me erials within the plant. Any of these components that may be contaminated will be cleaned and or disposed of consistent with the decommissioning technology available at the time of decommissioning.

Although operation of the ISFS1 will likely need to continue well beyond decommissioning of the Oconee Nuclear Statical, the costs of decommissioning the ISFSI are expected to represent a stable indinegligible fractio. of the cost of the decommissioning the Oconee Nuclear Station. Reference on page 4:37 submitted a schedule and justification for a decommission plan which will encompass decommissioning of both Oconee Nuclear Station and the Oconee ISFSI in accordance with 10CFR 50.75 and 10CFR 72:30. The financial options for this plan were submitted on July 24, 1990 for the NRC review and approval, Reference 6 on page 4-37.

The radiological impacts due to postulated accidents or operation of the ISFSI are bounding for the conditions when the ISFSI is fully operational. The collective dose to residents within one to two miles c^{*} the ISFSI is based on capacity loading of 2112 spent fuel assemblies in 88 storage modules. The occupational dose to sate workers assumes radiation from an array of 2 x 10 modules loaded with drv shielded casks each containing 24 spent fuel assemblies. The consequences from accidents are based ou failure of 24 spent fuel assemblies contained in a dry shielded cask. The expected radiological impact due to operation of the ISFSI is much less than the regulatory limits specified in 10CFR 72.104 and 10CFR 106(b) and the EPA Protection Action Guides. Furthermore, Duke intends to operate the Oconee ISFSI for the life of the licensed plant. It can be projected that once the spent fuel is removed from the ISFSI, the radiological consequences of decommissioning will actually be lower than that of operating the facility. Thus, the health and safety of the public are not affected by this exemption request.



4.35

4.6 Decommissioning Plan

Oconee ISFSI Safety Analysis Report

4.7 REFERENCES

- 1. Topical Report for the Nutech Horizontal Modular Storage System, for Irradiated Nuclear Fuel NUH-002, Rev. 1A, dated July 1989
- 2. Oconee Nuclear Station Final Safety Analysis Report
- 3. Oconee Nuclear Station ISFSI Physical Security Plan January 12, 1988
- 4. Oconee Nuclear Station Pre-Fire Plan
- 5. Letter from H. B. Tucker to U.S. NRC, Document Control Desk dated May 9, 1989
- 6. Letter from D. L. Hauser to U.S. NRC dated July 24, 1990



Chapter 4. Storage System

1

è

D

Oconee ISFSI Safety Aralysis Report

ġ,

.

Oconee PASI Safety Acalysis Report

Chapter 5. Storage System Operations

CHAPTER 5. STORAGE SYSTEM OPERATIONS

Chapter 5. Storage System Operations

Oconee ISFSI Safety Analysis Report

5.1 OPERATION DESCRIPTION

As a supplement to Sections 4.3, "Transfer System" on page 4-11 and 4.4, "Operating Systems" on page 4-21 of this report and Section 5.1 of Reference 1 on page 5-15 which describe the transport and fuel loading systems and their operation, this chapter describes the actual operations which occur at the ISFSI site after transfer of the DSC from the fuel building.

5.1.1 NARRATIVE DESCRIPTION

The following steps describe the operating procedures which occur after the DSC has been loaded with irradiated fuel assemblies and transferred to the ISFSI site. A more detailed description of HSM loading steps is provided in Section 5.1.1.6 of Reference 1 on page 5-15.

5.1.1.1 Loading of the DSC into the HSM

- Inspect all air inlets and outlets on the HSM to ensure that they are clear of debris. Inspect all screens on the air inlets and outlets for damage. Replace screens if necessary. Using an available yard crane, completely remove the front access door of the HSM. Inspect the interior of the HSM and the DSC support rail surfaces for obstructions or debris.
- 2. Using an appropriate towing vehicle position the transfer cask/trailer assembly inside the gross alignment marks on the HSM pad and move it slowly, toward the HSM until the docking collar is at the minimum distance from the HSM opening to allow for cask hid removal.
- 3. Using the optical alignment system, the targets on the transfer cask and HSM, and the skid positioning system, adjust the position of the cask until the cask is properly aligned with the HSM.
- 4. Unbolt and remove the cask lid and the cask bottom access plate.
- 5. Move the cask against the HSM so that the docking coilar is completely seated in the HSM recess.
- Secure the cask to the HSM using the cask restraint system and the anchors on the front wall of the HSM.
- 7. Assemble and align hydraulic ram assembly with transfer cask/trailer and secure in place. Recheck alignment of the HSM, transfer cask, and ram assembly.
- 8. Extend the hydraulic ram toward the cask and activate the grapple to engage the DSC.
- 9. Continue extension of the hydraulic ram to move the DSC into the HSM. If the ram fails to extend when the load on the hydraulic system is increased beyond 20,000 lbs., or, if a sudden, large increase in hydraulic pressure is observed, the DSC may be jammed or bound. If jamming or binding is suspected, corrective actions as described in Section 8.1.1.4, "Corrective Actions" on page 8-3 will be applied.
- 10. When the DSC is in the HSM, release the grapple from the DSC and retract the hydraulic ram arm from the transfer cask.
- 11. Disassemble and return the hydraulic ram to its transfer trailer and pull the transfer cask trailer a few inches away from HSM.
- 12. Lower the HSM front access door back into the door frame to within a few feet of the closed position.
- 13. Install seismic restraint.



5.1 Operation Description

- 14. Lower to the closed position and tack weld the steel HSM front access docr.
- 15. Measure the change in temperature between the HSM inlet and outlet air vents.
- 16. Return all equipment to storage locations peuding delivery/loading of next DSC.

5.1.1.2 Monitoring Operations

On a 24 hour frequency site personnel will visually inspect all air inlets of each loaded HSM for both obstructions and screen dama_s. HSM air outlets will be monitored daily using site surveillance cameras. Obstructions and/or damage will be removed/repaired immediately. The ISFSI site will also be included in the routine site patrols performed by Oconee's Security personnel.

5.1.1.3 Fuel Identification and Accountability

In compliance with NRC regulations, accountability records for all fuel assemblies transferred to, stored in or removed from the ISFSI will be maintained.

The asymmetrical design features of the DSC allow for easy identification of specific assembly storage locations within the DSC. No visible physical labels are necessary for the individual storage locations. Unique storage location symbols will be administratively assigned to each of the 24 DSC storage cells. This is similar to the method which is currently used to track assembly locations within the spent fuel pools. Unique identifications will be assigned to the HSMs, and will be labeled on the HSM exterior. This visible physical identification in combination with the administrative assignment of cell storage locations within the DSC, and a unique serial number stamped on each DSC, will allow for the positive identification of the locations of all ISFSI spent fuel assemblies.

Once a DSC has been inserted into a HSM, the door will be lowered and tack welded into place. These tack welds will sufficiently indicate any attempts at tampering as required in ANSI 57.9-84.

Unique identification of the transfer cask will not be required since only one transfer cask is to be used. This eliminates the possible mixup of transfer casks which might occur with multiple casks being used for concurrent transport operations. Accountability and control of special nuclear materials will be maintained at all times during the loading, transport, and storage of spent fuel assemblies.

5.1.1.4 Unloading the DSC from the HSM

- 1. In pect the front access components of the HSM and cut tack welds on HSM access door. Remove cask lid.
- Positiv 2 the cask trailer assembly so that the docking collar is at the minimum distance from the HSM to allow for opening of the HSM front access door.
- 3. Using the optical alignment system, the targets on the transfer cask and HSM, and the skid positioning system, adjust the position of the transfer cask until the transfer cask is properly positioned with respect to the HSM.
- Using an available yard crane, raise the front access door of the HSM just high enough to access the seismic restraint.
- 5. Remove seismic restraint.
- 6. Remove the HSM access door from the support rails.
- 7. Move the transfer cask against the HSM so that the docking collar is completely seated in the HSM recess.





- Secure the transfer cask to the HSM, using the cask restraint system and the anchors on the front wall of the HSM.
- 9. Align hydraul., ram assembly with transfer cask/trailer and secure in place. Recheck alignment of H5M, cask and ram assembly.
- 10. Recheck the cask and ram alignment to ensure it is properly positioned with respect to the HSM.
- Extend the hydraulic ram through the transfer cask into the HSM and activate the grapple to engage the DSC.
- 12. Write it is increased by the hydraulic ram to move the DSC out of the HSM and into the transier cask. If the ram fails to retract when the load on the hydraulic system is increased beyond 20,000 lbs., or if a sudden large increase in hydraulic pressure is observed, the DSC may be jammed or bound. If jamping of binding is suspected, corrective actions as described in Section 8.1.1.4, "Corrective Actions" on page 8-3 will be applied.
- 13. When the DSC is in the transfer cask, release the grapple from the DSC and retract the hydraulic ram arm from the transfer cask.
- Replace the top lid and bottom access plate on the transfer cask and move all hydraulic equipment away to allow for transfer cask trailer movement to appropriate location for DSC removal or offsite shipment.

5.1.2 FLOW SHEET

Loading and unloading operations are illustrated in Figure 5-1 on page 5-7.

5.1.3 IDENTIFICATION OF SUBJECTS FOR SAFETY ANALYSIS

5.1.3.1 Criticality Prevention

Criticality in the NUHOMS-24P DSC is prevented through a combination of geometrical separation of the fuel cells, neutron absorption in the cell walls and administrative controls on fuel pool soluble boron concentration and the selection of fuel to be stored in the DSC. The DSC basket malles use of two material thicknesses in the cell walls as well as some over-sleeves at the top and bottom of interior cells to accommodate sufficient neutron absorption with qualified fuel assemblies. While the DSC design features will be essentially fixed, the selection of fuel for storage will be a variable. Administrative controls and procedures are provided in Sections 4.4, "Operating Systems" on page 4-21 and 10.2, "Development of Operating Controls and Limits" on page 10-5. The criticality analysis for the NUHOMS-24P System can be found in Section 3.3.4 of Reference 1 on page 5-15.

5.1.3.2 Instrumentation

The proposed ISFS1 is a system requiring no instrumentation for radiation, temperature or criticality considerations.

5.1.3.3 Maintenance Techniques

Due to the passive nature of the proposed ISFSI, the only maintenance on the HSM will be periodic surveillance of the air inlet and outlet vents to insure continued air flow. Routine maintenance on the transfer cask will also be performed to maintain integrity of top lid, bottom access plate and trunnions.





5.1 Operation Description

5.1.3.1 Administrative Controls to Limit DBT Effects

Administrative controls for limiting transfer operations due to potential tornado weather conditions will not be required. The transfer cask in transit has been evaluated for tornado wind speeds and DBT effects in accordance with 10CFR Part 72 and was found to be enveloped by the evaluation for a design basis cask drop accident.

5-6

5.1 Operation Description

5.1.4 FIGURES



Figure 5-1. NUHOMS System Loading Operations Flowchart



5-7

5.1 Operation Description





5.1 Operation Description



NUHOMS System Loading Operations Flowchart

5.9

5.1 Operation Description

Oconee ISFSI Safety Analysis Report



Figure 5-4. NUHOMS System Loading Operations Flowchart

.

5.1 Operation Description



Figure 5-5. NUHOMS System Loading Operations Flowchart



5-11

5.1 Operation Description

Oconee ISFSI Safety Analysis Report



NOTE: NUHOMS SYSTEM RETRIEVAL OPERATIONS FLOW CHART IS SHOWN IN FIGURE 5.1-4 OF REFERENCE 5.1

Figure 5-6. NUHOMS System Loading Operations Flowchart



5.2 CONTROL ROOM AND CONTROL AREAS

This is a passive system and there is no need for annunicators or other systems to indicate off-normal conditions.

Surveillance for such conditions will utilize visual inspection techniques. Security surveillance will be tied into the main central alarm station and /or recondary alarm station at the Oconee Nuclear Station.

5.2 Control Room and Control Areas



5.3 REFERENCES

1. Topical Report for the Nutech Horizontal Modular Storage System for Irradiated Nuclear Fuel, NUH-002, Revision 1A, dated July 1989



Chapter 5. Storage System Operations

Oconee ISFSI Safety Analysis Report

CHAPTER 6. WASTE MANAGEMENT

No radioactive wastes are generated during the storage life of DSCs. Radioactive wastes generated during loading operations are treated using existing station facilities and procedures.

Contaminated pool water removed from loaded DSCs will normally be drained back into the spent fuel pool with no additional processing. A small amount (<15 CF/DSC) of liquid waste results from transfer cask decontamination. The decontamination procedure results in a small amount of a detergent/demineralized water mixture being collected in the Cask Decontamination Pit. Liquid wastes collected in the Cask Decontamination Pit are directed to the Station Liquid Waste Management System (LWM) for processing.

Potentially contaminated air and helium purged from the DSC following DSC loading and seal welding operations are directed to the Auxiliary Building Ventilation Air System (VA) at a point upstream of the Fuel Building HVAC filter units and radioactive effluent monitor. Purged gases processed with the Fuel Building HVAC filter units are released from the unit vent and will meet station release requirements. This is the same procedure currently utilized for shipping cask operations.

A small quantity (< 5 CF/DSC) of low level solid waste is generated as result of DSC loading operations and transfer cask decontamination. The solid waste generated is processed by compaction or incineration using appropriate facilities. This low level waste consists of disposable Anti-C garments, tape, blotter paper, rags, etc.

Descriptions of the LWM, VA and VR Systems are provided in Chapter 11 of the Oconee FSAR.

1-

Chapter 6. Waste Management

George ISFSI Safety Analysis Report

Chapter 7. Radiation Protection

CHAPTER 7. RADIATION PROTECTION

Chapter 7. Radiation Protection

Oconee ISFSI Safety Analysis Report



7.1 ENSURING THAT CCCUPATIONAL RADIATION EXPOSURES ARE ALARA

7.1.1 POLICY & ORGANIZATIONAL CONSIDERATIONS

Duke Power Co. Radiation Protection and ALARA policies are described in Chapter 12, "Radiation Protection" on page 12-1 of the Oconee FSAR (Reference 1 on page 7.35). These policies will be applied to the Independent Spent Fuel Storage Facility. Duke Power is committed to a strong ALARA program in design and operation of nuclear facilities. The ALARA program follows the general guidelines of Regulatory Guides 1.8, 8.8, 8.10 and 10CFR 20. Plant and design personnel are trained and updated on ALARA practices and dose reduction techniques. Design and implementation of systems and equipment are reviewed to insure ALARA worl, on all new and modification projects. The basic ALARA Program consists of

- 1. The Duke Power Company ALARA Manual:
- continued surveillance and evaluation of in-plant radiation and contamination conditions, as well as the monitoring and control of the exposure of personne', by the station and general office radiation protection staff;
- 3. an ALARA Committee at each station consisting of management and representatives from all groups, including liaison from the general office radiation protection staff, whose purpose is to conduct and appraise the effectiveness of the ALARA program at the nuclear facility.

The committee members has extensive background in nuclear plant radiation and exposure control, including such areas as layout, shielding, personnel access, ventilation, waste management, monitoring systems, operations, and maintenance.

Although upper level management is vested with the primary responsibility and authority for administering the Duke ALARA program, the responsibility for ALARA is extended through lower management to the individual employee.

Specific responsibilities of the general office and station radiation protection staffs are contained in the Duke Power Company Radiation Protection Manual.

This manual ensures that:

- 1. An effective ALARA program is administered at the Oconee nuclear station that , _____, riately integrates Duke management philosophy and NRC regulatory requirements and guidance.
- 2. Facility design features, operating procedures and maintenance practices are in accordance with ALARA program guidelines; and that written reviews of the on-site radiation control program assure that objectives of the ALARA program are attained.
- 3. Pertinent information concerning radiation exposure of personnel from other utilities and research work are reflected in the design and operation of the Duke Facility.
- 4. Appropriate experience gained during the operation of nuclear power stations relative to in-plant radiation control is factored into revisions of procedures to assure that the procedures continually meet the objectives of the ALARA program.
- 5. Necessary assistance is provided to insure that operations, maintenance, and decommissioning activities are planned and accomplished in accordance with ALARA objectives.

7.1 Occupational Radiation Exposures

- 0
- 6. Trends in station personnel and job exposures are analyzed in order to permit corrective actions to be taken with respect to adverse trends.

Reports of the findings of the general office and station radiation protection staffs are also effectively conveyed to management.

Specific responsibilities of station personnel are to ensure that:

- 1. Activities are planned and accomplished in accordance with the objectives of the ALARA program.
- Procedures and their revisions are implemented in accordance with the objectives of the ALARA program.
- 3. The general office radiation protection staff is consulted as necessary for assistance in meeting ALARA program objectives.

Other group and individual responsibilities to the ALARA program are outlined in Section II of the DPC ALARA Manual.

The primary goal of the radiation protection and ALARA programs is to minimize exposure to radiation such that the total exposure to personnel in all phases of design, construction, operation and maintenance are kept As Low As Reasonably Achievable. This is achieved by integrating ALARA concepts into design, construction, and operation of facilities. The radiation protection program identifies the positions and responsibilities of participating organizations in conducting these programs.

Trained personnel adequate to develop and conduct all recessary radiation protection and ALARA programs are provided. These personnel are trained to assure that all procedures are followed to meet company and regulatory requirements. Training programs in the basics of radiation protection and exposure control are provided to all facility personnel whose duties require working in radiation areas. Design personnel responsible for design of systems and equipment in the radiation area are trained in ALARA design techniques and the fundamentals of dose reduction. Radiation Protection personnel are provided training to improve their performance in implementing the radiation protection programs. All personnel are retrained as needed to update practices to current state of the art methods.

The administrative organization is responsible for and has appropriate authority for assuring that the three basic objectives of the Radiation Protection program at Oconee Nuclear Station are achieved. These objectives are to:

- 1. Protect personnel
- 2. Protect the public
- 3. Protect the station

Protection of Personnel, includes surveillance and control over internal and external radiation exposure and maintaining the exposure of all personnel within permissible limits and as low as reasonably achievable (ALARA).

<u>Protection of the public</u>, includes surveillance and control over all conditions and operations that may affect the health and safety of the public. Included are such activities as redioactive gas, liquid and solid waste disposal, shipment of radioactive materials, an environmental radioactivity monitoring plan and maintaining portions o, the station emergency plan.

<u>Protection of the Facility</u>, includes monitoring to warn of possible detrimental changes and exposure hazards, to determine changes or improvement needed, and to note trends for planning future work.

This administrative organization is also responsible for and has appropriate authority for maintaining occupational exposures as far below the specified limits as reasonable achievable by assuring that:

- 1. Station personnel are made aware of management's commitment to keep occupational exposures as low as reasonably achievable;
- 2. Formal reviews are performed periodically to determine how exposures might be lowered;
- 3. There is a well-supervised radiation protection capability with specific defined responsibilities:
- 4. Station workers receive sufficient training;
- 5. Sufficient authority to enforce safe station operation is provided:
- Modification to operating and maintenance procedures and to station equipment and facilities are made where they should substantially reduce exposures at a reasonable cost;
- The radiation protection staff understand the origins of radiation exposures in the station and seek ways to reduce upposures;
- 8. Adequate equipment and supplies for radiation protection work are provided.

The Station Manager is responsible for the protection of all persons against radiation and for compliance with NRC regulations and license conditions. This responsibility is in turn shared by all supervisors. Furthermore, all personnel are required to work safely and to follow the regulations, rules, and procedures that have been established for their protection.

The Duke Power Company Technical System Manager - Radiation Protection establishes the Radiation Protection Program including the program for handling and monitoring radioactive material for Oconee that is designed to assure compliance with applicable regulations, technical specifications, and regulatory uides. This person also provides technical guidance and support for conducting this program, reviews the excetiveness and the results of the program and modifies it as required based on experience and regulatory changes, to assure that occupational radiation exposure and exposure to the general public are maintained as low as reasonably achievable.

Th "echnical System Manager also provides technical assistance to the Vice President. Nuclear Production, who has nanagement authority to implement the "as low as reasonably achievable" (ALARA) occupational exposure policy, to which Duke Power Company is committed.

The Radiation Protection Manager at Oconee is responsible for conducting the Radiation Protection Program that has been established for the station, including the ISFSI. The Radiation Protection Anager has the duty and the authority to measure and control the radiation exposure of personnel; to continuously evaluate and review the radiological status of the station; to make recommendations for control of elumination of radiation hazards; to assure that all personnel are trained in radiation protection; to assist all personnel in carrying out their radiation protection responsibilities; and to protect the health and safety of the public both on-site and in the surrounding area.

In order to achieve the goals of the Radiation Protection Program and fulfill these responsibilities for radiation protection; radiological monitoring, survey and personnel exposure control work are performed on a continuing basis for station operations and maintenauce including the ISFSI.

7.1.2 DESIGN CONSIDERATIONS

The design of the DSC and HSM comply with 10CFR 72 concerning ALARA considerations. Specific considerations that are directed toward ensuring ALARA are

7.1 Occupational Eadiation Exposures

Oconee ISFSI Safety Analysis Report

- Thick concrete walls on the HSM to reduce the surface dose to below an average of 20 mr/hr. The 20 inillirem per hour dose rate was the approved maximum for HSM wall dose rates in the NUHOMS-07P Topical Report. Actual calculated HSM wall surface dose rates are below 10 millirem per hour except a ent and door openings. The HSM shielding design was deemed ALARA considering construction costs, heat dissipation, and access requirements. Also, refer to Section 7.1.2 of Reference 2 on page 7-35 for the basis of the average 20 millirem per hour HSM contact dose rate. Additional shielding analysis is included in Section 7.3.2.2, "Shielding Analysis" on page 7-12 and Table 7.3-2 of Reference 2 on page 7-35.
- Lead shield plug on the ends of the DSC to reduce the dose to workers performing drying, scaling, and loading operations.
- · Use of a shielded transfer cask for handling and transportation operations of loaded DSCs.
- · Fuel loading procedures which follow accepted practice and build on existing experience.
- Recess in the HSM front for the transfer cask to fit into so as to reduce scattered radiation during transfer.
- · Double seal welds on each end of DSC to provide redundant radioactive material containment.
- Placing clean water in the transfer cask and DSC and sealing the DSC transfer cask annulus to prevent contamination of DSC exterior during loading.
- · Placing external shielding blocks over HSM air outlets to reduce direct and streaming doses.
- · Passive system design that requires minimum maintenance.
- · Insertion of internal shielding blocks around air inlets to reduce direct and streaming doses.
- Use of portable shielding during DSC drying welding operations to limit streaming from top end shield plug DSC annulus. The portable shield used during DSC closure operation to limit streaming from top end shield plug DSC annulus consists of 2.0 inches of Bisco NS-3, or equivalent, as shown in Figure A.2. Appendix A of Reference 2 on page 7-35. The portable shield will be put in place to minimize doses during direct-access operations such as top shield plug automatic welding setup, draining and drying operations, and setup of automatic welding equipment for the top cover plate. The portable shield will incorporate provisions to facilitate access to the drain and fill ports but may not be necessary during automatic welding operations.
- To minimize scatter at the HSM door during DSC loading, the top of the transfer cask docks into a recess in the HSM access door opening.
- Use of existing shipping procedures and experience to control contamination during handling and transfer of fuel.
- Leaving water in the DSC cavity and DSC/transfer cask annulus during welding operations as long as possible to reduce streaming through the gap. The water level in the DSC transfer cask annulus is lowered to approximately 5 to 10 inches below the top of the DSC shell. The water level in the DSC cavity is lowered to approximately 4 inches below the bottom surface of the top end shield plug. These levels are maintained during shield plug weiding operations. The remaining water in the annulus is not drained until after the cask cover plate is bolted into place.
- Providing a large control area around the ISFSI and locating the facility well away from normally occupied areas.
- Operation of the ISFSI will be performed under the Radiation Protection program of the station as described in 7.1.1, "Policy & Organizational Considerations" on page 7-3.

 Lead blanket screens may be employed to further reduce dose during decontamination and transfer operations. These and other ALARA measures precautions may be employed as needed based on experience gained from preoperational testing and early fuel loading efforts.

7.1.3 ALARA OPERATIONAL CONSIDERATIONS

Consistent with Duke Power Company's overall commitment to keep occupational radiation exposures as low as reasonably achievable, (ALARA), specific plans and procedures are followed by station personnel to assure that ALARA goals are achieved. Operational ALARA policy statements are formulated at the corporate staff level in the Nuclear Production Department through the issuance of the System Radiation Protection Manual and the ALARA Manual and are implemented at each nuclear plant by means of procedures. These statements and procedures are consistent with the intent of Section C.1 of Regulatory Guides 8.8 and 8.10.

Since the ISFSI is a passive system, no maintenance is expected on a normal basis in the facility. Maintenance operations on the transfer cask, transfer trailer and other ancillary equipment is performed in a very low dose environment when fuel movement is not occurring.

Maintenance activities that could involve significant radiation exposure of personnel are carefully planned. They uclize any previous operating experience, and are carried out using well rained personnel and proper equipment. Radiation Work is (RWPs) for non-routine operations, or Standing Radiation Work Permits (SRWPs) for routine series are issued for each job, listing Radiation Protection requirements that shall be followed by all personnel working in the Radiation Control Area (RCA). Where applicable, specific radiation exposure reduction techniques, such as those set out in Regulatory Guide 8.8, are evaluated and used.

The station ALARA Committee carefully reviews operations and maintenance activities involving the major plant systems to further assure that occupational exposures are kept ALARA.

7.1 Occupational Radiation Exposures

Oconee ISFSI Safety Analysis Report

7.2 RADIATION SOURCES

7.2.1 CHARACTERIZATION OF SOURCES

This section describes the design basis radiation sources and source geometries used for the ISFSI shielding calculations.

Neutron and gamma sources are developed based on the reference irradiated fuel assembly described in Trible 1-1 on page 1-7. The reference fuel assembly is assumed to be irradiated to a burnup of 4 000 mwd/mtu and cooled to a decay heat rate of less than or equal to 0.66K whefore being stored in the LSC. The initial ϵ inchment considered is 4.0 weight percent U-235. The source terms include the irradiated fuel, activated portions of the fuel assemblies and deposited activity from corrosion products in the reactor coolant. All primary sources are considered to be originating in the fuel with secondary gammas generated in the shielding codes used.

The detailed calculation of gamma ray group fractions provided in Table 7.2-2 of Reference 2 on page 7-35 is summarized in Table 7-1 on page 7-10.

The fuel region is modeled as a homogeneous cylinder for shielding calculations as shown in the model geometry descriptions. The homogeneous source over the active fuel region includes fission product, actinide and ligh ment activation product sources. The burnup distribution is assumed flat along the arial and radial extent of source. This modeling technique is used in all shielding calculations except supplementary calculations performed subsequent to ISFSI operation. The supplementary calculations use a heterogeneous source distribution to demonstrate the effect IFA end fitting and plenum region light element activation and reduced self-shielding have on localized TC surface dose rates.

Additional details of the radiation source terms and dose conversion factors used in ISFSI shielding analysis are provided in Section 7.2.1 of Reference 2 on page 7-35.

7.2.2 AIRBORNE RADIOACTIVE MATERIAL SOURCES

The DSC is double seal welded to prevent any gaseous relevise of material during storage. The possibility of release during fuel handling in the spent fuel pool is covered in the accident analysis. The other possible source of airborne radioactive material is the outside surface of the DSC. This surface is protected from contamination while the DSC is in the fuel pool by filling the annulus between the DSC and the transfer cask with domineralized water and sealing the annulus to prevent pool water from coming in contact with the outside surface of the DSC. This prevents any significant accumulation of potential airborne sources on the canister. The outside surface of the transfer cask is considered to be contaminated upon rem. from the fuel pool and will be cleaned and swiped to be sure no unacceptable contamination remains before leaving the fuel building.

Cask venting releases are directed to the fuel pool HVAC units upstream of the HEPA and carbon filter units. The filtered gas is ultimately released through the unit vent after it is monitored by both the spent fuel pool storage area HVAC monitor and unit vent monitor.



Ŧ


7.2.3 TABLES

Cask Energy Group No.	Eupper (MeV)	E _{mean} (MeV)	Gamma Source Strength (Photons/sec/\striftM
23	10.0		0
24	8.0		0
25	6.5	5.50	3.84 + 6
		4.75	
26	5.0	4.25	1.16+7
***		3.75	
97	4.0	3.25	1.53+9
28	3.0	2.80	8.93+9
#0		2.40	
50	2.5	2.00	3.96 + 11
20	3.0		0
30	1.66	1.47	1.88 ± 13
31	1.00	1.11	266+14
32	1.00	4.4.2	0
33	1.0	6.44	434+15
34	0.8	0.63	4.34 7 13
35	0,0	her states in the second	
36	0.4	0.30	1.92 + 14
	0.3	김 아내는 아이들은 것이 같아요.	0
		0.17	
38	0.2	0.12	4.91 + 14
		0.085	
39	0.1	0.055	1.11 + 15
		0.030	
40	0.05	0.010	3.38+15
			5 9.80 + 15
		A	Il Group



Oconec ISFSI Safety Analysis Report

7.3 RADIATION PROTECTION DESIGN FEATURES

7.3.1 INSTALLATION DESIGN FEATURES

The design considerations listed in Section 7.1.2, "Design Considerations" on page 7-5 ensure that occupational exposures to radiation are AI ARA and that a high degree of integrity is obtained for the confinement of radioactive materials. The ISFS1 will be hand monitored as needed for construction. loading and unloading operations. Since the storage facility contains no active systems, no continuous monitoring systems other than fence-mounted dosimetry or alarms are needed. Applicable portions of the guidance given in Regulatory Position 2 of Regulatory Guide 8.8 have been followed: 1) Access control of radiation areas is addressed in Sections 7.1.3, "ALARA Operational Considerations" on page 7-7 and 10.2.5. "Administrative Controls" on page 10-6. 2) Radiation shields substantially reduce exposure of personnel during operations and storage; radiation streaming has been reduced by providing labyrinth-type shield penetrations. 3) NUHOMS-24P is a passive storage system: no process instrumentation or controls are necessary during storage. 4) Airborne contaminants and gaseous radiation sources are controlled by the integrity of the double seal welded DSC assembly. 5) No crud is produced by the NUHOMS-24P system. 6) The necessity for decontamination is reduced by maintaining the cleanliness of the DSC during operations (see Section 5.1. "Operation Description" on page 5-3); the DSC surfaces are smooth, nonporous, and free of crevices, cracks, and sharp corners. 7) No radiation monitoring system is required during storage. 8) No resin or sludge is produced by the NUHOMS-24P system.

Radiation sources are contained within DSCs which are stored in concrete HSMs. The radioactive sources are described in detail in Section 7.2.1 of Reference 2 on page 7-35.

7.3.2 SHIELDING

7.3.2.1 Radiation Shielding Design Features

Radiation shielding is an integral part of both the DSC and HSM designs. The features described in this section assure that doses to personnel and the public are "as low as is reasonably actuevable" (ALARA).

The DSC body is a rolled stainless steel plate. Details of the DSC and HSM and relevant dimensions can be found in the drawings in the proprietary supplement of Reference 2 on page 7-35. Two lead-filled shield plugs provide neutron and gamma shielding at the ends of the DSC. During handling operations, shielding in the radial direction is provided by the NUHOMS-24P transf r cask.

Two penetrations in the top shielded end plug allow water draining, vacuum drying and helium backfilling of the DSC. The penetrations are located away from fuel assemblies and contain sharp, non-coplanar bends to reduce radiation streaming. Figure 7-1 on page 7-15 shows the physical arrangements of the DSC end-shields and location of doses reported in Table 7-2 on page 7-14. These dose rates assume the water levels in the DSC cavity and DSC/Cask ann dus are lowered to the levels specified in Section 1.3.1.7, "System Operation" on page 1-13.

The transfer cask provides radiological shielding during the DSC drying operation and during the transfer to the HSM. Both neutron (solid Bisco NS-3) and gamma (lead) shielding are incorporated into the cask design. The NS-3 neutron shield is 3" thick (nominal) and has a density of 1.76 gm/cc. A 10% hydrogen content loss is assumed in the shielding analysis due to anticipated degassing of the NS-3 induced by elevated temperatures. The as-buil transfer cask lead gamma shield thickness is verified through radiographic examination to be 3.38" thick (nominal), but varies in thickness from approximately 3.15" to



7.3 Radiation Protection Design Features

3.5*. Areas where the lead thickness falls below 3.38° are covered by an additional 1/4° thickness of stainless steel neutron shield jacket to compensate for the reduced gamma shielding effectiveness of the lead.

The HSM provides shielding in both the radial and axial directions during the storage phase. Thirty six inch thick, portland cement, concrete walls and roots provide neutron and gamma shielding. The module's front end opening is covered by a three inch thick carbon steel door with a neutron shield.

Openings to the HSM interior are placed above the end shield regio. and not directly over the active fuel region. Sharp duct bends and concrete shielding caps over the exhaust exits assure that radiation streaming is reduced to a minimum. Figure 7-1 on page 7-15 shows details of the module penetrations and locations of doses reported in Table 7-2 on page 7-14.

Portable shielding during handling operations may be applied during specific handling operations. However, Section 7.4, 'Estimated On-Site Collective Dose Assessment' on page 7-17 provides an assessment of design basis on-site doses without the use of portable shielding.

7.3.2.2 Shielding Analysis

This section describes the radiation shielding analytical methods used in calculating relevant NUHOMS-24P system dose rates during the handling and storage phases. The dose rates were calculated at the locations listed in Table 7-2 on page 7-14. Figure 7-1 on page 7-15 shows these locations on the HSM, DSC and transfer cask. The three computer codes used for analysis are described below.

<u>Computer Codes</u> ANISN (Reference 3 on page 7-35), a one-dimensional discrete ordinates transport computer code, was used to obtain neutron and gamma dose rates at the outer HSM wall, centerline of DSC end plug, and outside the loaded transfer cask. The CASK (Reference 6 on page 7-35) cross section library, which contains 22 neutron energy groups and 18 gamma energy groups, was applied in an S_8P_3 or $S_{16}P_3$ approximation. Calculated radiation fluxes were multiplied by flux-to-dose conversion factors to obtain final dose rates. The ANISN calculations used coupled neutron and gamma libraries. Therefore, both primary and secondary gammas are calculated in each run.

QAD-CG (Reference 4 on page 7-35), a three-dimensional point-kernel code, was used for direct gamma shielding analysis of the HSM door, the DSC and wansfer cask end sections, the DSC transfer cask annuals, and the HSM air vent penetrations. Mass attenuation, and buildup were all obtained from QAD-CG's internal library for eight energy groups. The gamma energy spectrum was determined in the same manner as the ANISN analysis.

Shielding analysis results are summarized in Table 7-2 on page 7-14. Additional details regarding methods, models and assumptions used in ISFSI shielding analyses are provided in Section 7.3 of Reference 2 on page 7-35.

A similar version, QAD-CGGP, is used in supplementary calculations performed to determine the level of localized gamma dose rate peaking which may occur over areas of the TC surface corresponding to IFA end fitting and fuel pin plenum axial elevation. Gamma sources and spectra for the various IFA source regions modeled (i.e., active fuel, upper and lower end fittings, and upper and lower fuel pin plenums) are determined in the same manner as in ISFSI shielding calculations described in Section 7.3 of Reference 2 on page 7-35.



1

1

1

1

1

ł



Oconee ISFSI Safety Analysis Report

7.3 Radiation Protection Design Features

7.3.3 TABLES

7.3 Radiation Protection Design Features

Oconee ISFSI Safety Analysis Report

	Neutror Rate (n	a Dosc ar/hr)	Gamma Do Primary an	se Rate (mr/hr) d Secondary*	Total Dose Rate
Location	Direct	Reflected	Direct	Reflected	(mr/hr)
DSC In HSM			82.043		
1. HSM Wall or Roof	0.1	**	1.1.4	**	7
2. HSM Air Outlet Shielding Cap	No	0.2	<1	83	83
3. HSM Air Outlet (No Shielding Cap)	0.7	15	390	4200	4606
4. Center of Door	37	**	8	**	45
5. Center of Opening	430	**	330	**	760
6. Center of Air Inlets	0.1	2	< 7	86	88
7. 4.5 Ft. From HSM Door	20		4		24
DSC In CASK					
 Centerline Top of DSC Plug (with water in annulus and with 2 inches temporary neutron shielding) Top of DSC Cover Plate (with water in annulus and with 2 inches of temporary neutron shielding) 	5.3	*	10		15
h. Centerline	40	**	30	**	70
b. Gap (Peak)***	32		24	100	156
3. Transfer Cask Surface					
a. Radial (Centerline)	54	**	146	÷	200
Radial (Peak***)	54	**	511	**	565
b. Top axial	15	**	1	**	16
c. Bottom axial	32	**	16	**	48

Notes:

20

* T. DSC/Cask annulus is filled with water and additional neutron shielding material is utilized as required. In addition, all but top six inches of the DSC innui cavity is assumed to be filled with water for this operation.

** The reflected dose at these locations is negligible.

*** The same gap dose rate applies for case where only top lead plug is on DSC. The dose rates reported are with water in the DSC/cask annulus (however, no water was assumed to be in the DSC).

**** Estimated maximum radial surface dose rate localized near IFA end fitting and fuel pin plenum axial elevations

7.3.4 FIGURES





7.3 Radiation Protection Design Features

Oconee ISFSI Safety Analysis Report

Oconee ISFSI Safety Analysis Report

7.4 ESTIMATED ON-SITE COLLECTIVE DOSE ASSESSMENT

7.4.1 OPERATIONAL DOSE ASSESSMENT

This section establishes the expected cumulative dose delivered to site personnel during the fuel handling and transfer activities associated with one NUHOMS-24P module. Chapter 5, "Storage System Operations" on page 5-1 describes in detail the ISFS1 operational procedures, a number of which involve radiation exposure to personnel.

The storage facility is surrounded by a large open area for operational and security purposes. Access to the storage modules is restricted such that for normal operation, no access closer than 50 feet is allowed except for security purposes. Except during periods of additional module constantion, there is no adjacent work area close by, so very little dose is received from fuel in storage. Access is primarily needed to load new canisters into storage modules and dose from previously stored fuel will be received during these operations. The occupational exposure received during DSC transfer operations is included in the operational dose assessment summarized in Table 7-3 on page 7-19. The occupational dose estimates provided in Table 3 on page 7-19 are calculated using reference fuel assembly characteristics (see Table 1-1 on page 1-7) and other site-specific parameters. Dose contributions from hidden module scatter effects and self shielding for an 88 loaded module array are included in the Table 7-3 on page 7-19 results for DSC transfer operations. The dose received for other operations performed within the HSM storage facility secured area is negligible.

The phased construction of modules up to the licensed capacity of 88 will be undertaken on an as-needed basis considering required lead time, station operation and construction schedules. Increments of additional module construction are flexible. Duke expects to construct an average of 4 to 5 additional modules per year (considering normal station discharge rates) until the ultimate licensed capacity of 88 H5Ms is reached. Construction work performed subsequent to the loading of any HSM with spent fuel will result in worker exposures from direct and sky shine radiation in the vicinity of the loaded HSMs.

Duke plans to locate additional storage units beside the existing units to eventually form an effective 2 x 44 array of modules upon completion as shown in Figure 4-3 on page 4-26. In this manner, dose to construction workers will be minimized since the dose rates along the side of the HSM are much lower than those at the front of the module where there are vent openings. Construction materials will be staged away from the adjacent loaded HSMs, and very little work will be done in the higher dose rate area in front of the HSMs. Also, it is expected that new HSM construction will be commenced before all the existing HSMs are filled. Therefore, the HSMs adjacent to the new construction may be empty during some or most of the construction phase. The construction area will be surveyed prior to beginning work to ascertain actual dose rates and temporary shielding may be provided if needed to lower any unacceptable dose rates. The most significant dose rate contributors to the construction area are the inlet and exhaust vent openings. These dose rates may be reduced using temporary shielding screens around the vents near the construction area. After the concrete is placed for the additional modules, the additional shielding will further reduce dose rates.

The dose estimate for additional construction is based on labor cost estimates for a 2 x 10 module array. It is assumed that 60 percent of v^2 labor hours are expended in the radiation area and the prefabrication work is done in low or no dose areas. Table 7-4 on page 7-22 surumarizes expected construction doses by task.

7.4 Estimated On-Site Collective Dose Assessment

The maximum dose received from the loading, construction, and maintenance of Horizontal Storage Modules is 15 Rem per year for the expected loading rates. This is approximately 1.5% of normal station dose. The total includes fuel handling and canister loading operations, additional module construction and general maintenance of the facility. The dose estimate conservatively assumes design basis source terms for all fuel, construction of additional modules at a rate of 5 modules per year and general area doses from a full 88 module array for the entire period of HSM construction. Actual doses will be far below these estimates.

7.4.2 STORAGE TERM DOSE ASSESSMENT

No firm construction schedule for module addition has been developed at this time and thus the array sizes mentioned in Section 7.4.1, "Operational Dose Assessment" on page 7-17 are representative of possible additional increments. Additional increments of HSMs will be constructed as required to balance the off-loading of Oconec's fuel from the storage pools and transhipment to the federal repository.

Figure 7-2 on page 7-24 is a graph of the dose rate (.nr hr) versus distance from the face of a 2 x 3 array of NUHOMS-24P HSMs. Figure 7-3 on page 7-25 and Figure 7-4 on page 7-26 show the dose rate versus distance from the front or side of the array for various other HSM array sizes. These curves were constructed from the shielding analysis described in the previous sections and are for the dose rate in the worst case direction from the modules (perpendicular to the doors). The bounding conditions may be obtained by simply scaling the results from these curves. Direct neutron and gamma flux, as well as the air-scattered radiation from the module surfaces are considered. The surface radiation sources used for the direct and air scattered dose calculations are shown in Figure 7-5 on page 7-27. Neutron and gamma flux spectra for the surface of the HSM are provided in Table 7-5 on page 7-23. The HSM surface spectra are obtained from normalized ANISN model flux data. The ANISN HSM model as well as the CASK cross section library are described in Section 7.3.2.2 of Reference 7.2, "Radiation Sources" on page 7-9. The CASK cross section library is made up of 40 energy groups (group . 1-22 are neutron and groups 23-40 are gamma). Air-scattered dose rates are determined with the computer code SKYSHINE-II (Reference 7.5, "Radiation Protection Program" on page 7-29); direct dose rat is and calculated using the computer code MICROSHIELD (Reference 7.7, "Estimated Off-Site Collective Doses" on page 7-33). The direct flux from the "hidden" row of modules is considered completely shielded by the front row. All HSMs are assumed loaded with sufficiently cooled (≤ 0.66 Kw per assembly) spent fuel.

Oconee ISFSI Safety Analysis Report

7.4 Estimated On-Site Collective Dose Assessment

7.4.3 TABLES

Operation	Number of Personnel	Time ⁽²⁾ (Hours)	Ave. Dist. From Cask/ DSC/Cask Surface (Feet)	Dose Rate (mR 'Hr)	Total Personnel Dose (P-mR)
Location: Fuel Pool					
Load Fuel into DSC	2	8	GA(3)	2	32
Place Shielded End Plug on DSC	2	0.5	GA	2	2
Location: Cask Handling Area					
Decontaminate and Survey Surface of Cask	3	2	8 Side	35	210
Lower Water Level in DSC Cavity and DSC/Transfer Cask Annulus	2 2	0.25	1.5 F/D Port GA	48 2	24 8
Tack Weld Top End Shield Plug to DSC		0.25	1.5 Top Edge	48	12
Set up Automatic Welder and Seal Weld Top End Shield Pug to DSC	2 2	1.5 3	1.5 Top Edge GA	48 2	144 12
Perform Dye Penetrant Test on Welds	1	0.5	1.5 Top Edge	48	24
Remove Remaining Water/ Vacuum Dry DSC Cavity	2 2	0.25 3.75	1.5 F/D Port GA	55 2	27 15
Backfill DSC Cavity With Heliur	n 2	0.5	GA	2	2
Helium Leak Test	1.25	0.5	1.5 Top Edge	61	31
Seal Weld Vent/Siphon Ports	2	1.1	1.5 F/D Port	61	122
Perform Dye Penetrant Test on Welds	4	0.25	1.5 Top Edge	61	15
Install Top Cover Plate	2	0.25	1.5 Top Edge	61	31
Weld Top Cover Plate to DSC	2 2	0.35 2.65	1.5 Top Edge GA	61 2	43 11



7-19

7.4 Estimated On-Site Collective Dose Assessment

(Per	DSC Transfer	to HSM)			
Operation	Number of Personne:	Time ⁽²⁾ (Hours)	Ave. Dist. From Cask/ DSC/Cask Surface (Feet)	Dose Rate (mR/Hr)	Total Personnel Dose (P-mR)
Perform Dyc Penetrant Test on Weld	1	0.5	1.5 Top Edge	61	31
Remove Seal, Drain Cask/DSC Annulus and Swipe	2 2	0.75 3.25	1.5 Top Edge GA	151 2	227 13
Install Cask Head and Bolt Into Place	2	0.5	1.5 Top Edge	85	85
Lower Transport Cask to Skid and Trailer	2 4	1 2	4 Side 8 Side	120 67	240 536
Location: Trailer/HSM					
Attach Skid-Tiedown to Trailer	2	0.25	1.5 Side	210	105
Transport Cask to HSM	1 3		8 Side GA	67 2	67 6
Remove Cask Head, Bottom Cover Plate and Position Ram	2	0.5	1.5 Bot/Top	9()	90
Align Cask with HSM and Install Cask Restraints	4	1.5	4 Side	120	720
Transfer DSC from Cask to HSM	4	0.5	4 Side	120	240
Install Seismic Restraint	2	0.08	1 DSC Top	760	122
Close and Tack Weld HSM Door	1	0.25	4.5 HSM	23	6
Radiation Protection Survey of HSM	1	1	3 HSM	5	5

Table 7-3 (Page 2 of 3). Summary of Escimated On-Site Doses Resulting from ISFSI Operations(1).







Oconee ISFSI Safety Analysis Report

Operation	Personnel	(Hours) Total for T	(Feet) ransfer Operation	(mk/Hr)	(P-mR) 3255
	Number	Time(2)	DSC/Cask Surface	Dose Rate	Personnel
			Ave. Dist.		Total
	(Per DSC Transfer	to H8M)			

Table	7-3 (Page	3 of	3).	Summary	of	Estimated	On-Site	Doses	Resulting	from	ISFSI	Operations	(1)
				(Per DSC)	Trat	nsfer to HS!	M)						

Notes:

 Monitoring operation - Personnel will be monitoring the operation so that any problems which may arise can be swiftly corrected. The personnel may leave the area if necessary and the operation could be monitored from a remote location out of the radiation field.

Estimated times are conservative estimates for personnel working in the radiation field around the cash or "ISM.

 GA refers to General Area and is used to indicate workers in the room or area with the DSC/Transfer Cask to maintain visual control over operations in areas where the dose contribution from ISFSI operations is very low.



7.4 Estimated On-Site Collective Dose Assessment

Oconee ISFSI Safety Analysis Report

RADIAL CONTRACTOR AND A 19	TO ATTRY				
Task	Number Workers	Hours in Radiation Area	Average Dose Rate (mRem/hr.)	Maximum Indiv. dose (P-mRem)	Total Task Dose (P-mRem)
Survey	4	50	2	25	100
Excavation	4	192	2	96	383
Concrete Basemat	8	192	2	48	383
Forming Scaffolding Rebar	8	8496	2	2124	16992
Crane Operation	1	96	2	192	192
Steel Installation	8	1920	승규는 것이	240	1920
Welding	1	30	(81.g. 27)) 1973 - 2073	30	30
Surveyors (steel)	4	167	i la de la	42	167
Crane Operation (steel)	1	75	같은 비행이	75	75
Paint	8	96	1.1	12	96
Clean up	2	42	1	21	42
TOTALS		11354			20379

Table 7-4. Dose Estimate for Construction of Additional Horizontal Storage Modules Based on Labor Estimates for 2 X 10 Array

Estimated dose/module constructed = 1.02 Person Rem

Estimated maximum individual dose/module = 106 Person mRem



conce *SESI Ssl. , Analysis Report

7.4 Estimated On-Site Collective Dose Assessment

	Cask Library Group No.	Normalized Flux
	croup : co.	0.000005
		0.000005
	2	0.000030
		0.000138
	4	0.000982
	5	0.002567
	6	0.092456
	7	0.002970
N	8	0.007440
	9	0.006820
J	10	0.011135
1	11	0.017633
7	12	0.018344
e	13	0.026756
n	14	0.04222*
	15	0.019154
	16	0.024170
	17	0.020290
	16	0.014654
	10	0.019003
	19	0.019803
	20	0.017085
6	21	0.016067
	22	0.726040
Sum of Neut	rons = 1.0	
		0.000018
	42 5.	0.000145
	24	0.000145
	ter and the second s	0.000248
	26	91283
	21 - Carlos - 22 - Carlos - Ca	
	28	0.000275
G	29	0.001008
4	30	0.009272
m	31	0.007947
m	32	0.057772
a	33	0.051577
5	34	0.074007
	35	0.123743
	36	0.093142
	37	0.137524
	38	0.316630
	10	0.125240
		D. D.D. mme

** Sum of Gammas = 1.0



7.4 Estimated On-Site Collective Dose Assessment

7.4.4 FIGURES



Figure 7-2. Dose Rate Versus Distance From Surface of HSM





Oconec ISFSI Safety Analysis Report

1)



7.4 Estimated On-Site Collective Dose Assessment







7.4 Estimated On-Site Collective Dose Assessment

Oconee ISFSI Safety Analysis Report

7-26

Oconee ISFSI Safety Analysis Report



MAM FRONT PLAN

LOCATION	AREA	NELITRON	GAMMA	TOTAL
	(1 ²)	DOSE RATE	DOSE RATE	DOS PATE
ROCF				
1" 2" AREA WEXCHITED AVG PRONT	168.67 33.33 AVG	0.1 0.2 0.1	6.5 50 14	6.6 50.2 14.1
3**	28	0.1	6.5	6.6
4	28	37	7.8	47.6
5'	6.0	2.1	94	96.1
ARIEA WEIGHTED	AVQ	6.6	10	16.8

DIRECT AND REPLECTED DOSES
 DOSE RATES FOR ALL OF AREA 1 ON BOTH THE ROOF AND FRONT WALL ARE ESTIMATED TO BE THE SAME AS THE DOSE RATE CALCULATED AT THE CENTER OF THE ROOF.

Figure 7-5. Radiation Zone Map of Module Surface Dose Rates



7-27

7.4 Estimated On-Site Collective Dose Assessment

Oconee ISFSI Safety Analysis Report

7.5 RADIATION PROTECTION PROGRAM

The ISFSI is to be located adjacent to the Oconee Nuclear Station within the Owner Controlled Area. The Oconee Nuclear Station Radiation Protection Manager will have responsibility for Radiation Protection activities at the ISFSI.

The Radiation Protection and ALARA programs are discussed briefly in Section 7.1.1, "Policy & Organizational Considerations" on page 7-3 and will be as discussed in the Oconee Nuclear Station FSAR, Chapter 12. Detailed discussions of Radiation Protection and ALARA are contained in Duke Power Company's Health Physics and ALARA program manuals.

Radiation protection requirements for all radiological work at the Oconee Nuclear Station is governed by existing station directives, the Oconee Radiation Protection Manual, and station rediation protection (K, P_{\cdot}) procedures. R.P. practices for DSC loading, transfer, storage, monitoring, and retrieval will also be based on existing procedures, as well as on current and anticipated conditions when the task is \rightarrow be performed. These procedures include, but are not limited to, the following.

- 1. Procedure for personnel dosimetry issue.
- 2. Issuance, revision, and ermination of radiation work permits and showing radiation work permits.
- 3. Procedure for roping off, barricading, and posting of radiation control zones
- 4. Decontamination procedure for equipment and areas.
- 5. Smear swab sampling, counting, and calculation.
- 6. Procedure for quantifying airborne radioactivity.
- 7. Radiation Protection ALARA preplanning work.

In addition, the Radiation Work Perm's for the maintenance and fuel han⁴ling tasks associated with DSC operations will incorporate radiological hold points and precautions, where necessary, to ensure these activities are performed in a radiological safe manner and are ALARA.

Procedures and equipment for personnel in decon' amination operations are in place at Oconee and will be utilized as needed for ISFSI operations.



7.5 Radiation Protection Program

Oconec ISFSI Safety Analysis Report

Oconee ISFS1 Safety Analysis Report

7.6 ENVIRONMENTAL MONITORING PROGRAM

The current radiological environmental monitoring program for Oconee Nuclear Station will also serve as the operational program for the ISFSI.

No liquid or airborne effluents are anticipated from the HSM. Therefore, the dose to any offsite point will only be from direct and scattered gamma radiation. Several environmental sampling locations for direct radiation are presently located at the Oconee site boundary surrounding the ISFSI. The closest of these is less than 0.3 miles from the ISFSI, well within the 1-mile exclusion area boundary. In addition, the dose rates at the ISFSI will be monitored periodically with fence-mounted dosimetry as part of the Oconee routine radiological monitoring program. This will be used in part to control occupational exposures and will also augment the environmental program.

As a result, no changes to the environmental program are anticipated.



7.6 Environmental Monitoring Program

Oconee ISFSI Safety Analysis Report

Oconee ISFSI Safety Analysis Report

7.7 Estimated Off-Site Collective Doses

7-33

7.7 ESTIMATED OFF-SITE COLLECTIVE DOSES

Doses to any offsite point are only from direct and scatter gamma radiation from the storage module. The estimated dose from the modules to any dose point beyond the site boundary is well below regulatory limits even when combined with station doses for both airborne and direct gamma dose.

The ISFSI is situated approximately 1 mile from the exclusion area boundary. The estimated maximum dose rate in any direction at 5000 feet for up to an 88 module array of HSM's as provided by Figure 7-2 on page 7-24 through Figure 7-4 on page 7-26 is less than 1.0 x 10⁻⁶ mr/hr. The estimated annual dose to the public is conservatively calculated as 7 person-millrem per year. The maximum dose to the nearest potential future resident from the ISFSI is 7.5E-2 millirem per year.

7.7 Estimated Off-Site Collective Doses

Oconee ISFSI Safety Analysis Report



Oconec ISFSI Safety Anal; sis Report

7.8 REFERENCES

- 1. Oconee Nuclear Station Final Safety Analysis Report
- Topical Report for the Nutech Horizontal Modular Storage (NUHOMS-24P) System for Irradiated Nuclear Fuel, NUH-002, Revision 1A, July 1982
- Oak Ridge National Laboratory, "ANISN Multigroup One-Dimensional Discrete Ordinates Transport Code with Anisotropic Scattering" CCC-254, Oak Ridge National Laboratory (1977)
- Oak R dge National Laboratory, "QAD-CGGP, Point-Kernal Gamma Ray Shielding Code," CCC-396, Oak Ridge National Laboratory (1979)
- C. M. Lampley, "The SKYSHINE-II Procedure: Calculation of the Effects of Structure Design on Neutron, Primary Gamma-Ray and Secondary Gamma-Ray Dose Rates in Air" NUREG/CR-0781, RRA-T7901, USNRC (1979)
- Radiation Shielding Information Center, "CASK: 40 Group Neutron and Gamma Ray Cross Section Data," DLC-23, September 1978
- 7. Grove Engineering, Inc., "Microshield User's Manual, A Program for Analyzing Gamma Radiation Shielding," Version 2.0, 1985

Chapter 7. Radiation Protection

Oconee ISI'SI Safety Analysis Report

8-1

CHAPTER 8. ACCIDENT ANALYSES

In previous chapters, features important to safety have been identified and discussed. The purpose of this chapter is to identify and analyze a range of credible accident occurrences (from minor accidents to the design basis accidents) and their causes and consequences. For each situation, reference is made to the appropriate chapter and section describing the considerations to prevent or mitigate the accident.

ANSI ANS-57.9-1984, "Design Criteria for an Independent Spent Fuel Storage Installation (Dry Storage Type)." defines four categories or design events that provide a means of establishing design requirements to satisfy operational and safety criteria. The first design event is associated with normal operation. The sec ~ d and third design events apply to events that are expected to occur during the life of the installation. The fourth design event is concerned with severe natural phenomena or low probability events. The second design event is addressed in Section 8.1, "Off-Normal Operations" on page 8-3 and the third and fourth design events are discussed in Sectio. 8.2, "Accidents" on page 8-5. (The first design event is addressed in Chapter 4, "Storage System" on page 4-1 and need not be addressed here.)

Chapter 8. Accident Analyses

Oconec ISFSI Safety Analysis Report

8.1 OFF-NORMAL OPERATIONS

In this section, design events of the second type as defined by ANSI/ANS-57.9-1984 are addressed. Design events of the second type consist of events that might occur with moderate frequency on the order of once during any calendar year of operation.

The limiting off-normal event is defined as \pm_1 mmed DSC during loading or unloading at the ambient temperature extremes of -40°F and +125°F as described in Reference 1 on page 8-17 (Section 8.1, "Off-Normal Operations"). This postulated event results in the limiting structural loads on the DSC and thermal loads on the DSC and HSM for all islentified off-normal events. The ambient extremes for the Oconee site are bounded by the assumed values.

8.1.1 JAMMED DSC DURING LOADING OR UNLOADING

8.1.1.1 Postulated Cause of Jammed DSC

If the transfer risk is not accurately aligned with the HSM, the DSC might become bound or jammed during the sfer operation. The maximum tolerable misalignment for the Oconee ISFSI transfer operation is discussed in Section 5.1 of Reference 1 on page 8-17.

8.1.1.2 Detection of Jammed DSC

When DSC jamming occurs, the hydraulic pressure in the ram will increase above normal insertion pressures. When this occurs, the DSC will be presumed to be jammed. The pushing and pulling forces are limited to 20,000 lbs., with override control available to the operator.

8.1.1.3 Analysis of Effects and Consequences

The analysis of the DSC under assumed jamming and binding conditions is covered in Section 8.1.2.1 of Reference 1 on page 8-17. In both jammed DSC scenarios considered, the ciress on the DSC body is shown to be much less than the ASME code allowable stress. Therefore, plastic deformation of the DSC body will not occur and there is no potential for rupture. The analysis presented in Reference 1 on page 8-17 is applicable to Oconee ISFSI operation.

The ram force is limited to 80,000 pounds by a factory set and sealed crossport relief valve. This relief valve is installed in the hydraulic control system and is factory set to relieve at pressures which limit both the extension and retraction forces of the hydraulic cylinder to 80,000 pounds force.

8.1.1.4 Corrective Actions

In cases of DSC jamming or binding, the required corrective action is to reverse the direction of applied force on the DSC, and return the DSC to its previous position. Since no plastic deformation has occurred, the return of the DSC to its previous position will be unimpeded. The transfer cask alignment is then rechecked and the transfer cask repositioned as necessary before reinsertion is renewed.

8.1 Off-Normal Operations



8.1.2 RADIOLOGICAL IMPACT OF OFF-NORMAL OPERATIONS

Based on the off-normal operation analysis results presented, there is no additional radiological impact due to off-normal operations beyond what is presented in Chapter 7, "Radiation Protection" on page 7-1 of this SAR.





8.2 ACCIDENTS

This section addresses design events of the third and fourth types as defined by ANSI/ANS-57.9-1984, and other credible accidents which could impact the safe operation of the Oconee ISFSI. The postulated events addressed are:

- * Loss of Air Outlet Shielding
- Tornado Tornado Missile
- · Earthquake
- Transfer Cask Drop
- · Transfer Cask Loss of Neutron Shield
- · Lightning
- · Blockage of Air Inlets and Outlets
- · DSC Leakage
- · Accidental Pressurization of DSC
- Load Combinations
- Floods
- · Explosions

The postulated accidents listed above include all events identified as potentially resulting in offsite doses in excess of 25 mrem.

8.2.1 LOSS OF AIR OUTLET SHIELDING

This postulated accident involves the loss of both air outlet shielding blocks from the top of the HSM. All other components of the Oconee ISFSI are assumed to be in their normal conditions.

8.2.1.1 Cause of Accident

The air outlet shielding blocks are designed to remain in place and completely functional for all events except tornado missiles. To demonstrate the safety of the ISFSI design, this product assumes that both shielding blocks are completely lost.

The air of thet shield blocks are attached to the HSM by welding to an embedded plate in the HSM root. In the highly unlikely event of a recovery situation, the damaged shield block would be removed from the HSM and temporary shielding would be placed around the outlet opening in such a way that a worker could perform the necessary recovery techniques with a minimal radiation exposure. All Duke ALARA procedures, such as pre-staging construction activities in a no-dose area, would be followed throughout the entire recovery process.

8.2.1.2 Accident Analysis

There are no structural or thermal consequences to the ISFSI facility resulting from the loss of the air outlet shielding blocks. The air flow resistance is less without the shield blocks and, hence the air flow



8.2 Accidents

will increase (slightly) and provide more couling of the DSC. Radiological consequences of this recident are described in the next section

8.2.1.3 Accident Dose Consequences

Offsite radiological consequences result from an increase in air scattered (skyshine) dose due to the loss of the shuld blocks. Onsite radiological consequences result from an increase in direct (during recovery operations on the HSM roof) and skyshine radiation. The calculation of these doses during normal conditions is described in Section 7.4, "Estimated On-Site Collective Dose Assessment" on page 7-17. Removal of the shield blocks results in local surface dose increase of 3600 mr/hr at the vent opening. This increased surface dose was used in the models described in Section 7.4, "Estimated On-Site Collective Dose Assessment" on page 7-17 to calculate the direct and scattered doses as a function of distance from the HSM. Table 8-1 on page 8-13 shows comparisons of the increased dose rate as a function of distance due to loss of the shielding blocks. The dose increase to a person located 100 feet away from the ISFSI installation for eight hours a day for seven days (recovery time) would be 30 mr. The increased dose to an offsite person for 24 hours c day for seven days located 5000 feet away would be minimal.

To recover from the loss of shielding blocks, a new block is transferred to the HSM. After the shield block is transferred to the HSM, a yard crane is used to lift the block into position. The block is then bolted in place. The entire remounting of ration should take less than 30 minutes, of which a mechanic will be on the HSM roof for approximately 15 minutes. During this time he will receive less than 50 mr. An additional dose to the mechanic and to the crane operator on the ground while putting the shield block in place will be 10 mr each (assuming and average distance of 10 ft. from the center of the HSM front wall)

8.2.2 YORNADO/TORNADO MISSILE

8.2 Ause of Accident

The most severe tornado wind loadings specified by NUREG-0800, NRC Regulatory Guide 1.76 and the Oconee FSAR are used as the design basis for this accident condition.

8.2.2.2 Accident Analysis

The applicable design parameters of the design basis tornado (DBT) are specified in Section 3.2.1, "Tornado and Wind Loadings" on page 3-7 of this SAR. The DBT design parameters specified in Section 3.2.1, "Tornado and Wind Loadings" on page 3-7 are identical to those used in the reference Topical Report in the determination of forces on structures for this accident. The analysis of the HSM and Transfer Cask response to DBT loadings is covered by the analysis presented in Section 8.2.2 of Reference 1 on page 8-17.

8.2.2.3 Accident Dose Consequences

The only component of the ISFSI facility which is not capable of withstanding tornado generated missiles are the precast air outlet shielding blocks. The consequences of losing the shielding blocks during this accident is presented in Section 8.2.1.3, "Accident Dose Consequences" of this SAR.

8.2.3 EARTHQUAKE

Oconee ISFSI Safety Analysis Report

8.2.3.1 Cause of Accident

As specified in Section 3.2.3, "Seismic Design" on page 3-10, the ISFSI MHE acceleration value is 0.15g for both vertical and horizontal ground acceleration.

8.2.3.2 Accident Analysis

The reference Topical Report analysis of earthquake loads assumes a value of 0.25g and 0.17g for maximum horizontal and vertical acceleration, respectively. Reference 1 on page 8-17 seismic stress analysis also used a multiplier of 1.5.

Since the value of the seismic accelerations for the Oconee ISFSI site are lower than that assumed in Reference 1 on page 8-17, the stress analysis envelopes the site specific criteria.

In summary, the Oconee ISFSI seismic analysis using site specific criteria is enveloped by the analysis in Reference I on page 8-17.

8.2.3.3 Accident Dose Consequences

Major components of the Oconee ISFSI are designed and evaluated to withstand the forces generated by the MHE. Hence, there are no dose consequences.

8.2.4 CASK DROP

8.2.4.1 Cause of Accident

This section addresses the structural integrity of the DSC and its internals under a postulated transport cask accident condition. It is postulated that the transfer cask described in Section 4.3, "Transfer System" on page 4-11 with the DSC inside is dropped 80 inches onto a thick concrete slab. Due to the use of transfer cask trailer tie-downs, an actual drop event is not considered credible. Cask drop target parameters are given in Table 8-2 on page 8-13.

8.2.4.2 Accident Analysis

The Oconee ISFSI transfer cask is analyzed for an 80 inch drop accident using the method of analysis presented in Section 8.2.5. cf Reference 1 on page 8-17.

The analysis presented in Reference 1 on page 8-17 assumes an 80 inch cask drop using Oconee ISFS1 transfer cask parameters. Hence, the Reference 1 on page 8-17 analysis covers the Oconee accident analysis. Therefore, the stress on the various structural components of the DSC and its internals are the same as those reported in Table 8.2-7 of Reference 1 on page 8-17.

8.2.4.3 Accident Dose Consequences

Since the stress analysis has shown that all components important to safety of the DSC and its internal basket will perform their intended function under this accident condition, there are no dose consequences.

8.2.5 TRANSFER CASK LOSS OF NEUTRON SHIELD

8.2 Accidents



The neutron shield jacket is designed, fabricated, tested, and inspected as ASME Section III. Division 1 Class 2 vessels. The associated ASME quality assurance program will assure that there are no poor joints, or other substandard components in the transfer cask. The Bisco NS-3 neutron shield material is a rigid solid when cured and will not flow freely through openings in the jacket. Therefore, a loss of shield material will only occur in cases of external damage to the shield jacket and concurrent displacement of NS-3 material.

8.2.5.2 Detection of Shield Material Loss

Damage to the neutron shield jacket and material would be visually obvious. Anticipated loss of hydrogen from the NS-3 material resulting from degassing at evaluated temperatures is accounted for in the shielding analysis (see Section 7.3.2, "Shielding" on page 7-11).

8.2.5.3 Analysis of Effects and Consequences

For the purpose of this analysis, it is assumed that the transfer cask neutron shield will be breached as a result of postulated drop accident, and the shielding effect of the NS-3 will be lost. The effect of this will increase the cask surface contact dose from 180 mrem/hour to 837 mrem/hour. The only potential off-site dose consequences would be additional direct and air scattered radiation if the accident were to occur sufficiently close to the site boundary. It is assumed that eight hours would be required to either recover the neutron shield or to add temporary shielding while arranging recovery operations. As result, it is estimated that on-site workers at an average distance of fifteen feet would receive an additional dose rate of 80 mrem/hr.

Off-site individuals at a distance of 2000 feet would receive an additional dose of 5.7E-4 mrem for the assumed eight hour exposure. This increase is well within the limits of 10CFR 72 for an accident condition. Also, this does not preclude handling operations for recovery of the cask and its content. Water bags or other neutron absorbing material could be wrapped around the cask to reduce the surface dose to an acceptable limit for recovery operations thus minimizing exposure of personnel in the vicinity. The actual local and off-site dose rates, recovery time and operations needed to retrieve the cask, and the required actions to be performed following the event will depend upon the severity of the event and the resultant cask and trailer skid damage.

8.2.6 LIGHTNING

8.2.6.1 Cause of Accident

The likelihood of lightning striking the ISFSI and causing an off-normal operating condition is not considered a credible accident given the ISFSI lightning protection provided. The lightning protection system for the ISFSI is designed in accordance with NFPA NO. 78-1979 Lightning. Protection Code. This system precludes any damage to the HSM or its internal, due to lightning.

8.2.6.2 Accident Analyris

8.2.6.2.1 HSM

Should lightning strike the ISFSI, the normal operation of the HSM will not be affected. The current discharged by the lightning will follow the low impedance path offered by the lightning protection system. Therefore, the HSM is not damaged by the heat or mechanical forces generated by current passing

Oconce ISF. I Safety Analysis Report

through the higher impedance concrete. Since the HSM requires no equipment for its continued operation, the resulting current surge from the lightning will not affect the normal operation of the HSM.

8.2.6.2.2 Power Supplies

The hydraulic power supplies for the transfer trailer hydraulic positioners and the hydraulic ram are independent systems. Each of these systems have manually operated pumps which could be used in case of a power failure. Electrical power supplies to the ISFS1 site serve no safety related functions, since their loss would not adversely affect the NUHOMS-24P safety related components or the health and safety of plant personnel or the public.

The electrical power distribution system and associated equipment are electrically bonded to the lightning projection and grounding system for the ISFSI. The retail power transformer is installed with lightning protection features in accordance with National Electric Safety code requirements.

The lightning protection design meets the requirements of NEPA-78, Lightning Protection Code: 1986 Edition and IEEE Standard 665.

8.2.6.2.3 Welding of DSC to Support Structure

Movers int of the DSC from the transfer cask to the fully inserted position in the HSM takes less than 10 minutes. Transfer operations will not be attempted during a major thunderstorm when there is potential danger to plant personnel or costly damage to equipment. Therefore, the possibility of the DSC becoming welded to the support structure by a lightning strike is extremely unlikely. In addition, there is contact between the transfer cask and HSM mating collar, such that the anchorage of the transfer cask to the HSM shown in Topical Report Figure 4.2-6 provides a grounding path to the HSM. To complete this path, the attachment plates are grounded to the HSM reinforcing which will provide additional assurance that this event will not occur. Lightning would likely strike the highest nearby structure, which is a light pole.

The HSM rails are bonded to the HSM grounding system by means of exothermically welding a bare copper conductor to the embedded steel support plates and the HSM grounding system. Additionally, the trailer mounted ram assembly tripod is bonded to the HSM grounding system during cask positioning operations.

8.2.6.3 Accident Dose Consequences

Since no off-normal operating condition will develop as a result of lightning striking the ISFSI, there are no radiological consequences.

8.2.7 BLOCKAGE OF AIR INLETS AND OUTLETS

This accident involves the complete and total blockage of all HSM air inlets and outlets.

8.2.7.1 Cause of Accident

Since the HSMs are located outdoors, the air inlets and outlets could potentially be blocked by debris from such unlikely events as tornados. ISFSI design features such as a perimeter fence and separation of air inlets and outlets reduce the potential for this accident.


8.2.7.2 Accident Analysis

The structural consequences due to the weight of debris blocking the air openings are bounded by the structural consequences of other accidents described in this section (i.e., tornado and earthquake analyses). The thermal consequences of this accident result from heating of the DSC and HSM due to the lcss of natural convection cooling. An analysis of this condition is provided by Section 8.2.7 of Reference 1 on page 8-17.

8.2.7.3 Accident Dose Consequences

There are no offsite dose consequences as a result of this accident. The only dose increase is related to the recovery operation where the onsite worker will receive an additional 700 mr during an estimated 8 hour debris removal period.

8.2.8 DRY STORAGE CANISTER LEAKAGE

The DSC is designed for no leakage and analysis of normal and accident conditions have shown that no credible conditions could breach the canister body or fail the double seal welds at each end of the DSC. However, to show the ultimate safety of the ISFSI system, a total and complete instantaneous leak is postulated.

This postulated accident is the instantaneous release directly to the environment of 30% of all fission gasses mainly Kr_{85} and I_{129} contained in all the fuel rods in all 24 PWR fuel assemblies. This accident assumes that all fuel rods are ruptured and that concurrent DSC leakage occurs. All other components of the ISFSI system remain intact.

8.2.8.1 Cause of Accident

Due to the passive nature of the Oconee ISFSI system and the various design features, there is no credible event that could result in the rupture of all fuel rods concurrent with DSC leakage. However, to demonstrate the safety of the ISFSI design, this accident assumes that the fuel rods and the canister are ruptured due to an event of unspecified origin.

8.2.8.2 Accident Analysis

In the postulated Dry Storage Canister Loakage Accident, it is assumed that one DSC is breeched and fuel fails simultaneously releasing 30% of all fission gasses contained in 24 fuel assemblies. Following long-term wet storage (>7.5 years) the gaseous fission products which can be released are Kr₈₅ and I₁₂₉. The total DSC inventories assumed for Kr₈₅ and I₁₂₉ are 2.75E + 03 and 1.87E-02 Curies, respectively; these inventories are based on ORIGEN-S computer code (Reference 2 on page 8-17) analysis for 24 B&W 15x15 fuel assemblies irradiated for 40,000 MWD/MTU and decayed for 7.5 years.

Whole body and maximum organ doses are calculated for a hypothetical maximum individual assumed to be present at the nearest site boundary location (a distance of approximately 1 mile) for the duration of the event. A meteorological dispersion parameter (X/Q) of 4.5E-04 sec/m³ is used in calculating the maximum potential offsite doses; this X/Q value is consistent with the value referenced in the Oconee SER, Section 3.2.4, Units 2 and 3. Dose conversion factors used are obtained from NRC Legulatory Guide 1.109 and a breathing rate of 3.47E-04 m³ sec is used in calculating inhalation dose.

There are no structural or thermal consequences resulting from the DSC leakage accident described above. The radiological consequences of this accident are presented in Section 8.2.7.3, "Accident Dose Consequences."

8.2.8.3 Accident Dose Consequences

This post-lated accident involves the rupture of one DSC. All fuel rods contained in the ruptured DSC are assumed to fail simultance-sly such that 30% of all the fission gasses in the irradiated fuel assemblies are instantaneously released to the atmosphere. Whole body and maximum organ doses are calculated for a hypothetical individual assumed to be present at the Oconee Nuclear Station exclusion zone for the duration of the event. A meteorological dispersion parameter of 4.5E-4s/m³ is used in calculating the maximum potential offsite doses. The resulting calculated doses are 7 and 200 mr for the maximum offsite whole body and thyroid doses, respectively. These accident doses are well within the 10CFR 72 limit of 5000 mr whole body dose equivalent.

3.2.9 ACCIDENTAL PRESSURIZATION OF DSC

... accident addresses the consequences of accidental pressurization of the DSC.

8.2.9.1 Cause of Accident

Internal pressurization of the DSC could result from fuel cladding failure which would release fuel rod fill gas and free fission gas.

8.2.9.2 Accident Analysis

The maximum DSC accident pressurization is calculated assuming that the fuel rod fission gas release fraction is 30%, and that the original fuel rod fill pressure is 480 psig (Oconee fuel actually has a maximum initial fill pressure of 465 psig). The resulting internal DSC pressures at Oconee's maximum ambient temperature of 116°F and at the minimum ambient temperature of -30°F are below the accident pressures reported in Section 8.2.9 of Reference 1 on page 8-17 (for temperature extremes of 125°F and -40°F). The limiting accident for DSC pressurization is the loss of transfer cask neutron shield. Under these conditions, the gas temperatures in the DSC will rise to 600°T producing a DSC internal pressure of 49.1 psig. The DSC shell stresses due to accident pressurization are enveloped by those reported in Reference 1 on page 8-17.

During DSC opening, appropriate health physics techniques will be employed for respiratory protection of the workers and for preventing any uncontrolled releases to the environment. During cutting operations these techniques may i clude installation of exhaust hoods which discharge to the fuel building ventilation system upstream of the HEPA and carbon filter units and supplied air to the workers. During filling and venting, the vented gases will, also, be routed to the fuel building ventilation system. This is a routine precaution taken for opening of spent fuel shipping casks, and it would provide protection from respirable radioactive particles and, also, from the unlikely presence of a significant amount of escaped fission gases.

8.2.9.3 Accident Dose Calculations

Since the accidental pressurization is within the design basis limits of the DSC, there are no dose consequences.

8.2.10 LOAD COMBINATIONS

The load categories associated with normal operating conditions and accident conditions have been described and analyzed in previous chapters of this report. The load combination evaluation of various ISFSI safety related components is addressed in this section.



8-11

8.2 Accidents



8.2.10.1 Cause of Accident

The simultaneous loading of major ISFSI components by combined accident and normal loads would result in the load combinations analyzed.

3.2.10.2 Accident Analysis

The methodology used in combining normal operating and accident loads and their associated overload factors for various ISFSI components is presented in Section 8.2.10 of Reference 1 on page 8-17. The Reference 1 on page 8-17 analysis envelopes the Oconee ISFSI. The load combination and fatigue analysis in Reference 1 on page 8-17 indicates major ISFSI components can withstand severe load combination and thermal cycling without failure.

8.2.10.3 Accident Dose Consequences

There are no dose consequences for postulated load combination events.

8.2.11 FLOODING

The elevation of the ISFSI vard at Elevation 825.0 is more than eleven feet higher than the maximum flood level postulated for Lake Keowee, and therefore, flooding of the ISFSI site will not occur.

\$2.12 EXPLOSIONS

r 2 12.1 Cause of Accident

The explosion on S.C. Highways 130 or 183 of a tanker containing 8500 gallons of gasoline would subject the ISFSI to a surface overpressure.

8.2.12.2 Accident Analysis

According to the NRC Regulatory Guide 1.91 "Evaluations of Explosives Postulated on Transportation Routes Near Nuclear Power Plants," the explosion of 8,500 gallons of gasoline 1.100 feet from th. ISFS1 on S. C. Highway 130 or 183, would result in a peak overpressure of 1 psi about 1,900 feet from the point of explosion and therefore an overpressure of 2.3 psi at the ISFSI. The HSM has been designed to withstand a maximum tornado wind pressure of 3.0 psi. Therefore, the HSM overpressure from the explosion of a gasoline tanker on either S. C. Highway 130 or 183 is enveloped by the wind pressure analysis and design for a DBT.

8.2.12.3 Accident Dose Consequences

There are no dose consequences for postulated explosions.





8.2.13 TABLES

Table 8-1. Comparison of Total Dose Kates for HSM With and Without Air Outlet Shielding Blocks		
Dist#ce (meters) f am Nearest dSM Wall, 2x10 Array	No. mal Case Dose Rete* (mrem/hr.) (with Shield Blocks)	Accident Case Dose Rate* (mrem/hr.) (Without Shield Blocks)
10	2.85	21.9
100	0.4 587	0.533
500	8.97E-4	2.14F-3
2000	3.77E-8	9.62E-7

* Air scattered plus direct radiation.

Table 8-2. Cask Drop Target Parameters

1. Slab reinforcement:

Bottom mat -#5 s $@ 6^{*}$ c-c each way Top mat -#4's $@ 6^{*}$ c-c each way Yield strength = 60 kai per ASTM 615

- 2. Slab thickness = 1'-6" of concrete
- 3. Concrete strength (28 days) = 4000 pai (minimum)
- 4. Soil ultimate strength 12.0 ksf (Based on laboratory testing)
- 5. Soil elastic modulus = 174 ksf (Based on laboratory testing)
- Poisson's ratio of soil = 0.3 (Based on soil test data and "Foundation Analysis and Design" 3rd Ed., Joseph F. Bowles.)



8.2 Accidents

Oconee ISFSI Safety Analysis Report

8-15

8.3 SITE CHARACTERISTICS AFFECTING SAFETY ANALYSIS

All site characteristics affecting safety analyses presented in this SAR are noted where they apply.

8.3 Site Characteristics Affecting Safety Analysis

Oconee ISFSI Safety Analysis Report

8-16

Oconee ISFSI Safety Analysis Report

8.4 REFERENCES

- 1. Topical Report for the Nutech Horizontal Modular Storage (NUHOMS-24P) System for Irradiated Nuclear Fuel, NUH-002, Revision 1A, dated July 1989
- "SCALE-3: A Modular Code System for Performing Standardized Computer Analysis for Licensing Evaluation," NUREG/CR-0200, ORNL, Revision 3, December 1984

Chapter 8. Accident Analyses

Oconee ISFSI Safety Analysis Report



Oconec ISFSI Safety Analysis Report

Chapter 9. Conduct of Operations

CHAPTER 9. CONDUCT OF OPERATIONS

Chapter 9. Conduct of Operations

Oconee ISFSI Safety Analysis Report





9-2

Oconec ISFSI Safety Analysis Report

9.1 ORGANIZATIONAL STRUCTURE

9.1.1 CORPORATE ORGANIZATION

Duke Power Company is responsible for development of the ISFSI including design, construction, quality assurance, testing and operation of the facility. The corporate organization of Duke Power Company is fully described in Chapter 13, "Conduct of Operations" on page 13-1 of Reference 1 on page 9-17.

9.1.1.1 Corporate Functions, Responsibilities and Authorities

The corporate organization provides line responsibility for operation of the Company. Various departments within the Company have responsibility for design, "enstruction, quality assurance, testing and operation of the Oconee Nuclear Station as well as the iSFSI. Duke's corporate functions, responsibilities and authorities for quality assurance addressed in Topical Report DUKE-1-A, as described in Chapter 11. "Quality Assurance" on page 11-1 of this report, are applicable for appropriate portions of the ISFSI.

9.1.1.2 Applicant's In-House Organization

Duke's Nuclear Generation Department, headed by the Senior Vice President, Nuclear Generation, has corporate responsibility for overall nuclear safety, as established by Technical Specifications. Reporting to the Senior Vice President is a Vice President for each nuclear site, and the General Manager, Nuclear Services.

The Nuclear Generation Department Organization is described in Section 13.1.2, "Operating Organization" on page 13-4 of Reference I on page 9-17.

1.1

1

1

9.1.1.3 Interrelationship with Contractors and Suppliers

The development of the ISFSI including design, construction, testing and operation are managed and conducted by Duke Power Company. Technical support and other services for the program relating to the Nutech Engineers, Inc. supplied NUHOMS-24P are provided by Nutech Engineers. Inc. (now Pacific Nuclear Fuel System.), Inc.).

9.1.1.4 Applicant's Technical Staff

The Corporate technical staff supporting the ISFSI is described in Section 13.1.1, "Corporate Organization" on page 13-3 of Reference 1 on page 9-17.

9.1.2 OPERATING ORGANIZATION, MANAGEMENT, AND ADMINISTRATIVE CONTROL SYSTEM

9.1.2.1 Onsite Organization

The onsite organization of the Oconee Nuclear Station is responsible for operation of the ISFSI facility. The organization for Oconee Nuclear Station is fully described in Section 13.1.2, "Operating Organization" on page 13-4 of Reference 1 on page 9-17.

9.1.2.2 Personnel Functions, Responsibilities and Authorities

The functions, responsibilities and authorities of major personnel positions, including discussions of specific succession of responsibility for overall operation of the Oconee Nuclear Station including the ISFSI facility are described in Section 13.1.2.2, "Personnel Functions, Responsibilities and Authorities" on page 13-5 of Reference 1 on page 9-17.

9.1.3 PERSONNEL QUALIFICATION REQUIREMENTS

The qualifications of personnel in the operating staff are in accordance with Section 4 of ANSI 3.1-1978. "Selection and Training of Nuclear Power Plant Personnel," and are in accordance with Regulatory Guide 1.8 (Rev. 1). Section 13.1.3, "Qualifications of Station Personnel" on page 13-7 of Reference 1 on page 9-17 provides more details on personnel qualification requirements.

9.1.3.1 Minimum Qualification Requirements

The minimum qualification requirements for major operating, technical, and maintmance supervisory personnel are described in Section 13.1.3.1, "Minimum Qualification Requirements" on page 13-8 of Reference 1 on page 9-17.

9.1.3.2 Qualifications of Personnel

The qualification of personnel assigned to the managerial and technical positions are provided in Table 13-2 on page 13-19 of Reference 1 on page 9-17.

9.1.4 LIAISON WITH OTHER ORGANIZATIONS

All aspects of the ISESI development including design, procurement, construction, and operation are managed and conducted by Duke Power Company. Nutech Engineers, Inc. (now Pacific Nuclear Fuel Systems, Inc.), Duke Power (caspany's subcontractor, provides certain engineering, technical support, and other services for the ISE.) project relating primarily to the NUHOMS-24P dry storage cask system design.



Oconee ISFSI Safety Analysis Report

9.1 Organizational Structure

9.1.5 FIGURES

1

Figure 9-1. Deleted



9.1 Organizational Structure

Oconce ISFSI Safety Analysis Report

9.2 PREOPERATIONAL TESTING AND OPERATION

Prior to operation of the ISFSI, complete functional tests of the in-plant operations, transfer operations, and HSM loading and retrieval were performed. These tests verified that the storage system components (e.g. DSC, transfer cask, transfer trailer, etc.) could be operated safely and effectively.

9.2.1 ADMINISTRATIVE PROCEDURES FOR CONDUCTING TEST PROGRAM

Pre-operational testing procedures were written in accordance with existing Oconec procedure controls as governed by Duke Power Company's QA Program.

9.2.2 TEST PROGRAM DESCRIPTION

The testing program required the use of a DSC mock-up, transfer cask and associated handling equipment, transfer trailer, hyd, aulic ram and an HSM. The tests simulated, as nearly as possible, the actual operations involved in preparing a DSC for storage and ensured that they could be performed safely during actual emplacement of IFAs in the ISUSI. Shielding verification - which was not completely achievable during dry runs took place during the initial IFA loadings.

9.2.2.1 Operations

9.2.2.1.1 DSC and Associated Equipment

An actual DSC and a part-length mock-up of a DSC were obtained for pre-operational testing. The DSC was loaded into the transfer cask to verify fit and suitability of the DSC lift rig. Additionally, the DSC was used in operational testing of the transfer equipment and HSM.

The part-length mock-up was similar to the top end of the DSC with lead shield plug facsimile. The mock-up was welded by the automated welding equipment. Emphasis was placed on acceptability of the weld, as well as compliance with approved ALARA practices. The mockup was also used for verification of vacuum drying, helium backfilling, and cutting open operations.

9.2.2.1.2 Transfer Cask and Handling Equipment

Functional testing was performed with the transfer cask, lift beam, crane hook lift adaptor, and remote actuation equipment associated with the lift beam. These tests ensured that the transfer cask could be safely transported from the ONS truck bay to the decontamination pit. From there, the DSC/transfer cask was placed into the spent fuel pool cask pit to verify clearances and travel path and proper operation of the annulus seal.

9.2.2.1.3 Off-Normal Testing of the DSC and Transfer Cask

In the unlikely event that a problem arises during loading of IFAs into the DSC, seal welding/evacuation drying, transport of the DSC, or emplacement of a DSC into an HSM, no immediate action would be required. Operations in the spent fuel pool could be suspended indefinitely with IFA cladding temperatures well below the average long-term storage temperature limit of 340°C. During the other operations the IFA cladding temperature remains well below 570°C - an acceptable temperature for short-term operational and accelent conditions. The DSC transfer cask could be returned to the spent fuel pool if these other operations could not be completed in a timely manner. As stated in Section



9.2 Preoperational Testing and Operation



9.2.2.1.4 Transfer Trailer and HSM

The DSC/transfer cask was loaded with test weights to simulate loaded fuel and placed on the transfer trailer. It was then transported to the ISFSI and aligned with an HSM. Compatibility of the transfer trailer with the transfer cask, negotiation of the travel path to the ISFSI, and maneuverability within the confines of the ISFSI were verified. Additionally, it was verified that the 80 inch design basis height for a postulated cask drop could not be exceeded.

The transfer trailer was aligned and docked to the HSM. The hydraulic ram was used to emplace a DSC loaded with test weights in the HSM and remove it. Loading of the DSC into the HSM verified that the transfer skid alignment system, hydraulic positioners, and ram grapple assembly could operate safely for both emplacement of a DSC into an HSM, and removal of a DSC from an HSM.

9.2.2.1.5 Off-Normal Testing of the Transfer Trailer and HSM

In the unlikely event that a problem should occur that prevents losding the DSC into the HSM, no immediate remedial action will be required. IFAs may be stored in the transfer cask whale corrective action is taken.

The most severe condition would occur if a failure of the hydraulic ram, after partial insertion of a DSC into an HSM, were to prevent complete emplacement of the DSC. (Radiological shielding and decay heat removal are not compromised by this condition, but the transfer trailer may not be moved away until the DSC is completely within the confines of either the transfer cask or the HSM.) Pre-operational testing verified that reversal of DSC movement could be completed by the operator of the hydraulic ram.

9.2.3 TEST DISCUSSION

- The purpose of the pre-operational tests was to ensure that a DSC could be properly and safely
 placed in the spent fuel pool, loaded with IFAs, transported to the ISFSI, emplaced in the HSM, and
 removed from the HSM. Proper operation of the DSC, transfer cask, and transfer trailer, as well as
 the associated handling equipment (e.g. lift beam, welding equipment, vacuum equipment) provided
 this assurance.
- 2. Pre-operational test procedures were developed as stated in Section 9.2.1, "Administrative Procedures for Conducting Test Program" on page 9-7. Specific detailed procedures were developed and implemented by ONS personnel who were responsible for ensuring that the test requirements were satisfied. Changes made to the pre-operational procedures were incorporated into the appropriate loading procedure.
- 3. The result of the pre-operational tests was the successful completion of the following without damage to any component associated piece of equipment: loading of a DSC into the transfer cask, seal welding, drying, backfilling, and cutting open of the mockup DSC, placement of the transfer cask into and out of the ONS spent fuel pool, transporting the transfer cask loaded with a DSC to the ISFSI, and emplacement in an HSM and removal from an HSM.





Oconee ISFSI Safety Analysis Report

9.3 TRAINING PROGRAM

The existing training program for ONS was modified to incorporate the training needed for operation of the ISFSI, in accordance with the Duke Power Employee Training and Qualification System (ETQS) Standards Manual ETQS provides a systemic approach to training as described in the ONS FSAR. Section 13.1, "Organizational Structure" on page 13-3 of Reference 1 on page 9-17.

9.3.1 TRAINING FOR OPERATIONS PERSONNEL

Since the ISFSI is a passive storage system, generalized training is provided in the areas of cooling, radiological shielding, and structural characteristics of the DSC/HSM

Detailed operator training is provided for DSC preparation and handling, fuel loading, transfer cask preparation and handling, and transfer trailer loading. Although operations personnel may not be directly involved in transport or HSM loading, detailed training is provided to permit oversight of these operations by fuel handling personnel.

Additionally, Fire Brigade training has been expanded to include the ISFSI in the Oconee Nuclear Station Pre-Fire Plati

9.3.2 TRAINING FOR MAINTENANCE PERSONNEL

Maintenance personnel, involved with the ISFSI operations, receive generalized training in the NUHOMS-24P storage system. Specific training is provided for use of the automated seal welding equipment for the top end shield plug: operation of the transfer trailer, alignment of the cask skid with the HSM; alignment of the hydraulic ram assembly; and normal and off-normal operation of the hydraulic ram. Specific training is also being provided for cleaning of the HSM air inlets and outlets.

9.3.3 TRAINING FOR RADIATION PROTECTION PERSONNEL

Radiation Protection personnel have received generalized training in the NUHOMS-24P system. Specific training has been provided in radiological shielding design of the system, particularly the top end shield plug. DSC transfer cask, the shielding issue associated with transfer of the DSC into the HSM, and the HSM itself.

9.3.4 TRAINING FOR SECURITY PERSONNEL

Details of the training program for security personnel are provided in the Guard Training Plan contained in a separate enclosure which is withheld from public disclosure in accordance with 10CFR 2.790(d) and 10CFR 73.21.



1

T

REV: (30 JUNE 1991)

9.3 Training Program

Oconee ISFSI Safary Analysis Report



Oconee ISFSI Safety Analysis Report

9.4 NORMAL OPERATIONS

Under normal operations, the ISFSI provides independent storage of Oconee spent fuel away from the Oconee plant facilities. With the exception of some limited physical and continuous electronic security surveillance, the facility functions as a passive system. Loading of fuel assemblies into the facility, which occurs periodically, require specific procedures that are separate from those of normal plant operations.

9.4.1 PROCEDURES

Operating, testing, and maintenance procedures are prepared, revised, reviewed, and approved in accordance with the Duke Power Company Nuclear Production Department "Administrative Policy Manual" (APM). (The APM sets forth the specific requirements of the Duke Power Company QA Topical Report, DUKE-1-A, which has been approved by the NRC as meeting the requirements of 10CFR 50 Appendix B.)

9.4.2 RECORDS

The ISFSI records are maintained in accordance with existing Oconec Nuclear Station procedures.



Oconee ISFSI Safety Analysis Report

9.5 EMERGENCY PLANNING

The Emergency Program for Oconee Nuclear Station has been determined to be adequate to manage the consequences of events which might occur involving the ISFSI. Appropriate reviews were made of the existing emergency plan initiating conditions and it was determined that no changes were necessary. The Emergency Program consists of the Oconee Nuclear Station Emergency Plan and the Duke Power Company Crisis Management Plan for Nuclear Stations and their related implementing procedures. Also included are related radiological emergency plans and procedures of state and local governments. The purpose of these plans is to provide protection of plant personnel and the general public and to prevent or mitigate property damage that could result from an emergency at the Oconee Nuclear Station. The combined emergency preparedness programs have the following objectives:

- 1. Effective coordination of emergency activities among all organizations having a response role.
- 2 Early warning and clear instructions to the population-at-risk in the event of a serious radiological emergency.
- 3. Continued assessment of actual or potential consequences both on-site and off-site.
- 4. Effective and timely implementation of emergency measures.
- 5. Continued maintenance of an adequate state of emergency preparedness.

The emergency plans have been prepared in accordance with Section 50.47 and Appendix E of 10CFR Part 50. The plans shall be implemented whenever an emergency situation is indicated. Radiological emergencies can vary in severity from the occurrence of an abnormal event, such as a minor fire with no radiological health consequences, to nuclear accidents having substantial onsite and/or offsite consequences. In addition to emergencies involving a release of radioactive materials, events such as security threats or breaches, fires, electrical system disturbances, and natural phenomena that have the potential for involving radioactive materials are included in the plans. The plans contain adequate flexibility for dealing with any type of emergency that might occur

The activities and responsibilities of outside agencies providing an emergency response role are detailed in the State Emergency Plans and the emergency plans for Oconec and Pickens Counties.

The emergency response resources available to respond to an emergency consist of the personnel at Duke Power Corporate Headquarters, at other Duke Power nuclear stations, and, in the longer term, at federal emergency response organizations (e.g. NRC, DOE, FEMA). The first line of defense in responding to an emergency lies with the normal operating shift on duty when the emergency begins. Therefore, members of the Oconee staff are assigned defined emergency response roles that are to be assumed whenever an emergency is declared. The overall management of the emergency is initially performed by the shift supervisor until he she is relieved by the Station Manager. In the event of an emergency, he serves as the Emergency Coordinator. Because of his overall knowledge, he is best able to bring the full resources of the plant to bear on controlling the emergency. Onsite personnel have preassigned roles to support the Emergency Coordinator and to implement his directives.

Special provisions have been made to assure that ample space and proper equipment are available to effectively respond to the full range of possible emergencies.

The emergency facilities available include the Oconee Control Room. Operational Support Center, Technical Support Center, Crisis News Center, and the Crisis Management Center (Emergency

9.5 Emergency Planning

Operations Facility). These facilities are described in the station emergency plan and the Crisis Management Plan.

Emergency plan implementing procedures define the specific actions to be followed in order to recognize, assess, and correct an emergency condition and to mitigate its consequences. Procedures to implement the Plan provide the following information

- 1. Specific instructions to the plant operating staff for the implementation of the Plan.
- 2. Specific authorities and responsibilities of plant operating personnel.
- A source of pertinent information, forms, and data to ensure prompt actions are taken and that proper notifications and communications are carried out.
- 4. A record of the completed actions.
- 5. The mechanism by which emergency preparedness will be maintained at all times.

Oconec ISFSI Safety Analysis Report

44.010

9.6 PHYSICAL SECURITY PLAN

The purpose of the security program for the Oconee Nuclear Station ISFSI is to establish and maintain a physical security program that has the capabilities for the protection of spent fuel stored in the NUHOMS-24P system.

Additional information regarding the security program for the ISFSI is contained in a separate enclosure, that is withheld from public disclosure in accordance with 10CFR 2.790(d) and 10CFR 73.21. This enclosure addresses the Physical Security Plan, Safeguards Contingency Plan, Design for Physical Security and Guard Training Plan.



9.6 Physical Security Plan

Oconee ISFSI Safety Analysis Report



Oconec ISFSI Safety Analysis Report

9.7 References

9.7 REFERENCES

1. Oconee Nuclear Station Final Safety Analysis Report (FSAR)



9-17

Conduct of Operations

Oconee ISFSI Safety Analysis Report

Oconec ISFSI Safety Analysis Report

CHAPTER 10. OPERATING CONTROLS AND LIMITS

The ISFSI will basically operate as a passive system requiring minimal surveillance. However, there are some operating controls and limits which will apply. These controls and limits which are listed below are discussed in detail in the following corresponding sections of this chapter. Othe, items which must be controlled such as those related to fuel movement and loading are based on normal operation and postulated accidents as discussed in Chapter 4, "Storage System" on page 4-1 and Chapter 8, "Accident Analyses" on page 8-1, respectively, of this report.



10-1

Chapter 10. Operating Controls and Limits

Oconee IS^cSI Safety Analysis Report





Oconce ISFSI Safety Analysis Report

10-3

10.1 PROPOSED OPERATING CONTROLS AND LIMITS

Operating limits and controls may be found in Reference 2 on page 10-11.

10.1 Proposed Operating Controls and Limits

Occuree ISFSI Safety Analysis Report

Oconec IEFSI Safety Analysis Report

10.2 DEVELOPMENT OF OPERATING CONTROLS AND LIMITS

This section provides an overview and general bases for the operating controls and limits specified in this report. Reference 2 on page 10-11 provides the specifications associated with the operation of the Oconee ISESI to ensure the protection of the public's health and safety.

10.2.1 FUNCTIONAL AND OPERATING LIMITS, MONITOPING INSTRUMENTS AND LIMITING CONTROL SETTINGS

The Oconee ISFS1 utilizes the NUHOM5-24P system which is a parsive design. Therefore, with the exception of the limit placed on the translational force evented on the DSC by the hydraulic ram, no monitoring instruments or limiting control settings are utilized at the 19081 facility. Long term operating variables such as HSM storage temperatures and confinement in the setting be controlled through observance of the operational control and limit specifications describe. In theference 2 on page 10-11.

Another control which falls under the Chonee Station's IOCFR 50 operating license is a restriction on minimum cooling time for fuel stored in certain locations of the spent fuel pools during cask handling operations. These restrictions ensure that any radioactivity releases remain below regulatory guidelines in the event of an in-pool cask drop accident.

10.2.2 LIMITING CONDITIONS FOR OPERATION

10.2.2.1 Equipment

Limiting conditions for the Ocorrec ISFSI equipment are specified in Reference 2 on page 10-11. In addition, the ram hydraulic system will be pre-set to insure that translational loads on a DSC during movement into the HSM are automatically limited to a maximum of 20,000 lbs. (Override control will be available to hydraulic ram operator for use during off-normal remedial action if needed.)

10.2.2.2 Technical Conditions And Characteristics

The following technical conditions and characteristics are required for the NUHOMS-24P system:

- i. Boron Concentration in DSC Moderator
- 2. DSC Vacuum Pressure During Drying
- 3. DSC Helium Backfill Pressure
- 4. DSC Helium Leak Rate
- 5. DSC Dye Penetrant Test of Closure Welds
- 6. Fuel Assembly Retrieval and Inspection
- 7. DSC Surface Contamination
- 8. **DSC Draining Requirements**

A description of the bases for selecting the above conditions and characteristics is detailed in Reference 2 on page 10-11. The overall technical and operational considerations are further described in Section 10.2.2.2 of Reference 1 on page 10-11.





10.2.3 SURVEILLANCE REQUIREMENTS

Surveillance Requirements for the Oconee ISFSI are specified in Reference 2 on page 10-11.

10.2.4 DESIGN FEATURES

Changes to site specific design features important to safety are not anticipated for the Oconee ISFSI Design features of the NUHOMS-24P system important to safe operation are outlined in Section 10.2.4 of Reference 1 on page 10-11 and in Reference 2 on page 10-11. Changes to any of these design features will be implemented only after appropriate regulatory review and approval.

10.2.5 ADMINISTRATIVE CONTROLS

Use of existing and proposed Duke Power Company organizational and administrativ systems and procedures, record keeping, review, augit and reporting requirements (i.e. Duke Power Company Administrative Policy Manual, Oconee Nuclear Station Directives, Operating Procedures, etc.) will be used to ensure that the operations involved in the storage of spent fuel at the Oconee ISFS1 are performed in a safe manner. This includes both the relection of assemblies qualified for ISFS1 storage, and the verification of assembly identification numbers prior to and after placement into individual storage canisters.

10.2.5.1 Qualification of Spont Fuel

Figure 10.1 on page 10-10 represents the fuel assembly acceptance criteria for spent fuel placement and storage in the DSC. Fuel as umbly qualification is based on the requirements for criticality control, decay heat removal, structural integrity, and radiological protection. Criticality control and decay heat removal capabilities are defined by three variables shown in Figure 10-1 on page 10-10. (1) initial assembly enrichment, (2) final assembly burnup, and (3) spent fuel cooling period. Control of these three acministrative procedures, as described below.

For the NUHOMS-24P subcriticality is assumed for fuel assemblies meeting the 4.0 wt% errichment limit of Reference 2 on page 10-11 when the DSC is filled with water borated to at least 1810 ppm (arrequired by Reference 2 on page 10-11) or then the DSC is drained. To ensure subcriticality in the postulated event that the DSC is filled with domineralized, unborated water, the burnup requirements of Figure 10-1 on page 10-10 are specified for any permissible initial enrichment.

Procedures currently in place for special nuclear materials accountability and record keeping will be used to verify initial fuel assembly enrichment and burnup levels at discharge. New fuel enrichments and initial uranium isotopics are recorded from the DOE/NKC Form 741's and stored in both a database file and on duplicate paper copies of the Form 741's. Individual fuel assembly burnups are also stored in the special nuclear materials database. These values are generated by the Oconee Operator Aided Computer utilizing thermal energy production date determined by in-core flux mapping. Burnup and initial enrichment values from special nuclear mate d accountability records will be compared to Figure 10-1 on page 10-10 to verify that the reactivity level is acceptable for DSC loading and storage of each irradiated fuel assembly. Actual qualification procedures may utilize a tabular version of the enrichment-burnup curves which will allow for each linear interpolation between a number of data points. While this enrichment vs. burnup method for reactivity verification will routinely be used - and required by procedures. Duke Power reserves the right to rely on other NRC - accepted analytical methods to qualify fuel assemblies in special cases.





Oconee ISFSI Safety Analysis Report

For decay heat control, only those irradiated assemblies which do not exceed a decay heat level of 0.66 kw will qualify for loading into the DSC. Decay heat loadings at or below this level ensure that peak pin clad temperatures are maintained within acceptable levels. Since individual fuel assembly decay heat levels are a function of both the discharge burnup and the decay time, procedural controls will be used to verify these parameters prior to fuel assembly loading.

For the Oconee fuel design and routine operating histories, the decay time necessary so schieve a .66 kw decay heat level is generally 7.5 years. The variation in required cooling time is a very strong function of discharge burnup and a very weak function of initial enrichment. It is acceptable to store fuel assemblies cooled less than 10 years provided that decay heat production is no more than 0.66 Kw for each fuel assembly and that neutron and gamma source terms for the DSC are verified not to exceed certain values specified in Reference 2 on page 10-11.

As mentioned previously, special nuclear materials accountability records will be used to verify fuel assembly burnup. These records will also be used to verify spent full decay time. The individual assembly burnup ar 4 decay time will then be compared to Figure 10-1 on page 10-10 for DSC loading qualification surposes.

To ensure , a structural integrity of the spent fuel to be loaded into the DSC, station records of all damaged assemblies will be reviewed. A damaged fuel assembly and component database has been compiled which incorporates previous sipping, ultrasonic (UT) testing, and visual observation. This database will be examined as a part of the dry storage qualification process to verify that assemble with gross structural or gross cladding damage are not included.

If the reactivity, decay heat, cooling time structural integrity, and dose limits criteria are all met, then approval for dry storage for a given assembly will be documented. This documentation will subsequently be referenced through procedures at the station prior to loading fuel into the DSC.

10.2.5.2 Spent Fuel Identification

Administrative controls will be utilized to avoid fuel misplacement. Information on fuel assembly qualification for dry storage will be documented and transmitted to fuel handling personnel. Prior to any transfer of a fuel assembly in the DSC, specific DSC loading procedures will require a review of assembly documentation. This will be followed by an independent visual verification of the assembly identification number by two individuals. These procedures ensure that the correct (approved) fuel assembly is being accessed and loaded into the DSC. As a final check, all assembly identification numbers will be checked after the DSC has been fully loaded with 24 assemblies.



10.2 Development of Operating Controls and Limits

Oconec ISFSI Safety Analysis Report



(Series 15FSI Safety Analysis Report

10.3 OPERATIONAL CONTROL AND LIMIT SPECIFICATION

Functional and Operating Limits, Monitoring Instruments and Limiting Control Settings: Limiting Conditions for Operations; and Surveillance Requirements are specified in Reference 2 on page 10-11


10.3.1 FIGUPES



Figure 10-1. Fuel Assembly Acceptance Criteria Cooling Period > 10 Years



Oconec ISFSI Safety Analysis Report

10.4 REFERENCES

- Topical Report for the Nutech Horizontal Modular Storage (NUHOMS-249) System for Irradiated Nuclear Fiel, NUH-002, Revision 1A, July 1989
- Diske Power Company Special Nuclear Materials License SNM-2503, Docket No. 72-4 for the Oconee Independent Spent Fuel Storage Installation, as amended January 29, 1990

Chapter 10. Operating Controls and Limits





Oconee ISFSI Safety Analysis Report

Chapter 11. Quality Assurance

CHAPTER 11. QUALITY ASSURANCE

Duke Power Company maintains full responsibility for assuring that its nuclear power plants are designed, constructed, tested and operated in conformance with good engineering practices, applicable regulatory requirements and specified design bases and in a manner to protect the public health and safety. To this end Duke has established and implemented a quality assurance program which conforms to the criteria established in Appendix B to 10CFR Part 50, "Quality Assurance Criteria for Nuclear Power Plants and Fuel Reprints" and to approved industry standards such as ANSI N45.2-1977 and ANSI N18.7-193. d corresponding daughter standards, or to equivalent alternatives.

The activities associated with the Independent Spent Fuel Storage Installation (ISFSI) will be governed by the applicable portions of the Duke Power Company Quality Assurance Program. This Quality Assurance Program is described in the Duke Power Company Topical Report, DUKE-1-A. The Topical Report provides the current quality assurance program description for Oconec. McGuire, and Catawba Nuclear Stations, Docket Nos. 50-269, 50-270, 50-287, 50-369, 50-370, 50-413, and 50-414.

Addivional, the Topical Report describes the Quality Assurance Program for those systems, components, items, and services which have been determined to be safety related. In addition, Duke's Quality Assurance Program provides a method of applying a graded Quality Assurance Program to certain non-safety related systems, components, items, and services. This method involves defining a Quality Assurance "Condition" for each level of quality assurance required. These will be designated as "QA Condition ______." The following conditions have been defined.

QA Condition 1 covers those systems and their attendant components, items, and services which have been determined to be safety related. These systems are detailed in the <u>Safety Analysis Report</u> applicable to each nuclear station. The Topical Report applies in its entirety to systems, components, items, and services identified as QA Condition 1.

QA Condition 2 covers those systems and their attendant components, items, and structures important to the management and containment of liquid, gaseous, and solid radioactive waste-

QA Condition 3 covers those systems, components, items, and services which are important to fire protection a: defined in the Hazards Analysis for each station. The Hazards Analysis is in response to Appendix A of NRC Branch '(echnical Position APCSB 9.5-1.

QA Condition 4 covers those seismically designed restrained systems, components, and structures whose continued functions are not required during and after the seismic event. The general scope of these systems, components, and structures, identified as Seismic Cetegory 11 (SC11) are defined in Regulatory Guide 1.29, Sciencic Design Classification.

